ISTANBUL BUSINESS RESEARCH

Cilt Volume 53 Sayı Number 2 E-ISSN 2630-5488



Indexing and Abstracting

Emerging Sources Citation Index (ESCI) - Web of Science TR Index EBSCO Business Source Elite EBSCO Business Source Ultimate EBSCO Central & Eastern European Academic Source ProQuest Central ProQuest Central ProQuest Turkey Database DOAJ Erih Plus RePEc IDEAS RePEc EconPapers ASOS Index SOBIAD

ISTANBUL BUSINESS RESEARCH

Volume: 53 • Number: 2 • August 2024 E-ISSN: 2630-5488 • DOI: 10.26650/ibr

OWNER

Prof. Dr. Kamil Ahmet KÖSE

Istanbul University, School of Business, Department of Business Administration, Istanbul, Turkiye

RESPONSIBLE MANAGER

Prof. Dr. Güven ORDUN

Istanbul University, School Of Business, Department Of Business Administration, Istanbul, Turkiye

CORRESPONDENCE ADDRESS

Istanbul University, School of Business, 34322 Avcılar, İstanbul / Turkiye

Phone: +90 (212) 473 70 00

Fax: +90 (212) 590 40 00

E-mail: ibr@istanbul.edu.tr

http://dergipark.org.tr/ibr

http://ibr.istanbul.edu.tr/

PUBLISHER

Istanbul University Press

Istanbul University Central Campus, 34452 Beyazit, Fatih / Istanbul, Turkiye

Phone: +90 (212) 440 00 00

Authors bear responsibility for the content of their published articles.

The publication language of the journal is English.

This is a scholarly, international, peer-reviewed and open-access journal published triannually in April, August and December.

Publication Type Periodical

ISTANBUL BUSINESS RESEARCH

Volume: 53 • Number: 2 • August 2024 E-ISSN: 2630-5488 • DOI: 10.26650/ibr

EDITORIAL MANAGEMENT

EDITOR-IN-CHIEF

Prof. Dr. Seda Tolun TAYALI, Istanbul University, School of Business, Department of Quantitative Methods, Istanbul, Turkiye - *stolun@istanbul.edu.tr*

CO-EDITOR IN CHIEF

Asst. Prof. Dr. Halit Alper TAYALI, Istanbul University, School of Business, Department of Production, Istanbul, Turkiye - halpertayali@istanbul.edu.tr

EDITORIAL MANAGEMENT BOARD MEMBER

Asst. Prof. Dr. Diren BULUT, Istanbul University, School of Business, Department of Business Administration, Department of Marketing, Istanbul, Turkiye – *dbulut@istanbul.edu.tr*

Asst. Prof. Dr. Cem Güney ÖZVEREN, Istanbul University, Institute for Aviation Psychology Research, Istanbul, Turkiye – cem.ozveren@istanbul.edu.tr

Asst. Prof. Dr. Halit Alper TAYALI, Istanbul University, School of Business, Department of Production, Istanbul, Turkiye – halpertayali@istanbul.edu.tr

EDITORIAL ASSISTANTS

Dr. Ceren ÖZVEREN, Istanbul University, School of Business, Department of Accounting, Istanbul, Turkiye – ceren.dincel@istanbul.edu.tr

LANGUAGE EDITORS

Elizabeth Mary EARL, Istanbul University, School of Foreign Languages (English), Istanbul, Turkiye – elizabeth.earl@istanbul.edu.tr

EDITORIAL BOARD

Prof. Dr. Bülent AYBAR, Southern New Hampshire University, Manchester, USA – *c.aybar@snhu.edu* Prof. Dr. Arnold B. BAKKER, Erasmus University Rotterdam, Rotterdam, The Netherlands – *bakker@essb.eur.nl* Prof. Dr. Aykut BERBER, University of the West of England Bristol, Bristol, Unitd Kingdom – *Aykut.Berber@uwe.ac.uk* Prof. Dr. Hamparsum BOZDOĞAN, University of Tennessee, Knoxville, USA – *bozdogan@utk.edu* Prof. Dr. John L. CASTI, Stevens Institute of Techhnology, Hoboken,USA Prof. Dr. Pablo FERNANDEZ, University of Navarra, Pamplona, Spain – *fernandezpa@iese.edu* Prof. Dr. Arjun K.GUPTA, Bowling Green State University, Bowling Green, USA – *zirbel@bgsu.edu* Prof. Dr. Stefan KOCH, Boğaziçi University, İstanbul, Turkiye – *stefan.koch@boun.edu.tr* Prof. Dr. Ali M. KUTAN, Southern Iliinois University Edwardsville, Edwardsville, USA – *akutan@siue.edu* Assoc. Prof. Dr. Carolina F. MACHADO, E.E.G.-Universidade do Minho, Braga, Portugal – *carolina@eeg.uminho.pt* Prof. Dr. Michael MORLEY, University of Limerick Kemmy Business School, Limerick, Ireland – *michael.morley@ul.ie* Assoc. Prof. Dr. C. WARREN NEEL, University Logan, Australia – *chris.rowley@kellogg.ox.ac.uk* Prof. Dr. Yiğit ATILGAN, Sabancı University, İstanbul, Turkiye – *yigit.atilgan@sabanciuniv.edu* Assoc. Prof. Dr. Yiğit ATILGAN, Sabancı University, İstanbul, Turkiye – *yigit.atilgan@sabanciuniv.edu*

Table of Contents

ARTICLES

Research article Behind the Youtube Addiction and Online Stickiness. Does Fear of Missing Out Count?	37
Research article Model Proposal for Evaluation of Technology Development Process: Conceptualisation and Scale Development	61
Research article Dynamics of Digital Financial Inclusion in Türkiye 18 Behice Canatan, Ayşe Akboz-Caner, Gökhan Özbilge	85
Research article Changes in Capital Structure in Seasoned Equity Offerings: Bist Implementation	01
Research article The Impact of Fintech on Economic Growth: Evidence from Panels of Turkic and Southeast Asian Countries	29
Research article Lean Management's Effect on Industry 4.0 Regarding Innovation	49



Istanbul Business Research, 53(2), 137-159

DOI: 10.26650/ibr.2024.53.1339988 http://ibr.istanbul.edu.tr/ http://dergipark.org.tr/ibr

Istanbul Business Research

Submitted: 09.08.2023 Revision Requested: 07.06.2024 Last Revision Received: 24.06.2024 Accepted: 08.07.2024

RESEARCH ARTICLE

Behind the Youtube Addiction and Online Stickiness. Does Fear of Missing Out Count?

Beril Durmuş¹ , Hüseyin Ekizler², Murat Bolelli³

Abstract

As social media has evolved into a predominant aspect of daily life, with over 5 billion users globally, understanding the motivations behind user engagement is essential. The purpose of this study was to examine the relationships between loneliness, self-esteem, FoMO (Fear of Missing Out), YouTube addiction, and online stickiness concepts. This study hypothesizes that self-esteem and loneliness significantly affect FoMO, which in turn influences YouTube addiction and online stickiness. Data were collected from Turkish adults using an online survey tool with convenience sampling method. Confirmatory Factor Analysis, correlation, and path analyses were conducted to test the research hypotheses. The findings indicate positive and significant correlations, as well as the effect of self-esteem on FoMO. Additionally, loneliness was found to be negatively correlated with FoMO Social Experience and positively correlated with FoMO Rewarding Experience, whereas no significant effect of loneliness on both dimensions was found. FoMO Social Experience and FoMO Rewarding Experience were positively correlated with YouTube addiction, and FoMO Social Experience had a significant effect on YouTube addiction. Finally, YouTube addiction is strongly correlated with online stickiness, and its effect on it is significant. The implications of the results are discussed, and future research areas are suggested.

Keywords: Loneliness, Self-esteem, Fear of missing Out, YouTube addiction, Online stickiness

Introduction

Social media have become an important part of daily life in the last two decades (Sampasa-Kanyinga & Lewis, 2015). Research suggests that there are 5.07 billion social media users worldwide, representing 62.6% of the global population as of April 2024. Additionally, the average daily social media use per internet user has increased to 143 min per day (Statista, 2024). This implies that if a person signed up at age 16 and lived to 70, they would spend 5.36 years on social media (Backlinko, 2021).

Used as a generic term, social media can be considered a descriptive concept for all internet-based applications that allow users to create and share content (i.e., knowledge, expe-



¹ Corresponding Author: Beril Durmuş (Prof. Dr.), Marmara University, Faculty of Business Administration, Istanbul, Türkiye. E-mail: beril@marmara.edu.tr ORCID: 0000-0002-9679-9608

² Hüseyin Ekizler (Assoc. Prof. Dr.), Marmara University, Faculty of Business Administration, Istanbul, Türkiye. E-mail: hekizler@marmara.edu.tr ORCID: 0000-0001-5903-713X

³ Murat Bolelli (Assoc.Prof. Dr.), Altınbaş University, Faculty of Business Administration, Istanbul, Türkiye. E-mail: murat.bolelli@altinbaş.edu.tr ORCID: 0000-0002-9707-1387

To cite this article: Durmus, B., Ekizler, H., & Bolelli, M. (2024) Behind the YouTube addiction and online stickiness. does fear of missing out count? *Istanbul Business Research, 53*(2), 137-159. http://doi.org/10.26650/ibr.2024.53.1339988

riences, emotions, thoughts, information, etc.) generally in the form of videos, pictures, and text (Kaplan & Haenlein, 2010). Facebook, Instagram, Snapchat, Twitter, YouTube, Pinterest, Thumblr, Reddit, TikTok, and Foursquare are the most popular platforms among many others.

Considering the great interest in social media, numerous studies have been conducted to explain why people engage actively (by sharing) or passively (by following) in it (Dhir et al., 2018; Roberts & David, 2020). Various theories relate to the popularity of social media, such as belongingness, social comparison, validation, information foraging, and compensatory internet activity. These theories suggest that satisfaction with various needs is one of the most important drivers for engaging in such platforms, where users seek gratification in different forms through their participation (Kardefelt-Winther, 2014; Quinn, 2016; Quan-Haase & Young, 2010; Toma & Hancock, 2013).

Previous studies have suggested that social media engagement is particularly related to satisfying needs such as well-being, belongingness, connectedness, life satisfaction, self-esteem, social involvement, information sharing and collecting, and self-presentation (Beyens et al., 2016; Christofides et al., 2009; Hunt et al., 2018; Kuss & Griffiths, 2017; Nadkarni & Hofmann, 2012; Seidman, 2013; Turel & Serenko, 2012; Utz et al., 2012; Przybylski et al., 2013; Winter et al., 2014). Billions of users worldwide interact over social media applications in an attempt to satisfy these needs, where their experiences function as an inducer for behavioural reinforcement.

In particular, the more users engage in social media and the more satisfying their experiences, the stronger their drive becomes. This mechanism is asserted to create a loop of "addiction to the activities engaged" and lead to habitual and addictive behaviours towards social media use (Abel et al., 2016; Kim et al., 2010; Kuss & Griffiths, 2017). Needs satisfaction has been asserted to foster social media addiction either directly or indirectly (Barry & Wong, 2020; Przybylski et al., 2013).

Applications can be perceived as efficient means of connecting with others, sharing information, and being part of something; therefore, people may resort to using them intensively, directly enhancing the effect. Additionally, unsatisfied needs like loneliness, low self-esteem, low self-compassion, insecure self-perception, and unhappiness can function as mediators or moderators for excessive use of social media (Barry & Wong, 2020; Przybylski et al., 2013). In that case, the relationship between need satisfaction and social media use is indirect.

Nevertheless, in pursuit of fulfilment, maladaptive use of social media is reported to produce negative outcomes such as anxiety, depression, high stress levels, lower sleep quality, increased suicide incidents, reduced relationship quality, and fear of missing out (FoMO) (Adams & Kisler, 2013; Donnelly & Kuss, 2016; Kalpidou et al., 2011; Kross et al., 2013; Lup et al., 2015; Rosen et al., 2013; Tandoc et al., 2015).

FoMO is conceptualised as a feeling of missing a rewarding experience or being absent from an experience that others enjoy (Pollard, 2012; Przybylski et al., 2013). Closely related to negative or insecure self-perceptions and individuals' assessments of their relationships with others, FoMO is asserted to consist of two main components: "the desire to stay connected with what others are doing" (Przybylski et al., 2013) and "worry about others having a good time without them" (Barry & Wong, 2020). In relation to the rapid advancement of information and communication technologies that have shaped social media applications, the possibility of experiencing FoMO has increased significantly (Barry & Wong, 2020). Nourished by needs such as relatedness, belonging, and connectedness (Beyens et al., 2016; Deci & Ryan, 1985), FoMO is reported to induce problematic, addictive digital technology use (Aydın, 2018; 2022; Elhai et al., 2020; Ercengiz, 2020; Erciş et al., 2021; Kurtoğlu & Aydın, 2023; Özdemir, 2021; Sarıbay & Durgun, 2020; Teke & Yılmaz, 2024). Studies have reported two fundamental drivers that induce maladaptive use of social media and form the basis for FoMO: loneliness and low self-esteem (Abel et al., 2016; Burke et al., 2010; De Cock et al., 2014; Hunt et al., 2018; Kalpidou et al., 2011; Song et al., 2014; Xu & Tan, 2012). Mostly caused by general economic, political, and social conditions, loneliness, alienation, isolation, and related negative affects are asserted to be increasing worldwide (Blai, 1989; Lu et al., 2023; Rook, 1984; Sundberg, 1988; Wang et al., 2021; Zhu et al., 2021). Because loneliness is a strong driver for negative feelings that are undesirable (Burke et al., 2010), social media is found to be an easy and inexpensive (compared to the tangible and intangible costs of real relationships) venue by a large number of users to avoid or compensate for these feelings. Research shows that loneliness is related to not only higher consumption of social media but also FoMO, especially among younger generations (Barry & Wong, 2020).

Self-esteem can be defined as how people feel about themselves based on cognitive or affective self-evaluations (Leary & Baumeister, 2000). Social media is reported to provide safe and gratifying venues for such needs as self-expression, external validation, and self-image, which are related to self-esteem and are used to elevate them (Báachnio et al., 2016; Tian et al., 2019). Social media offers a secure space, especially for individuals with low self-esteem, where they can express themselves and connect with others without the boundaries they feel or experience in real life. On the other hand, social media also functions as a source that enables users to retrieve information about others for use in self-comparisons. Hence, it can be argued that through social media applications, people tend to both send and collect information to overcome loneliness and establish and improve a sense of self-worth, which lays the foundation for building self-esteem.

Considering the relationship between need satisfaction and social media use, it can be theorised that drivers such as self-esteem and loneliness should affect FoMO, thus reinforcing addiction and stickiness towards the applications. To examine this assumption, YouTube, which is the second most visited website in the world after Google (Alexa, 2019), was selec-

ted. On YouTube, users share and view videos on various topics, including politics, social or business events, products, people, games, music, TV, and education. It is reported that YouTube has over 2 billion monthly logged-in users (almost one-third of all internet users and 25% of the world population), with over one billion hours of content viewed every day (YouTube, 2021). Research has revealed that viewers tend to identify with their favourite YouTubers, and this one-way relationship is positively correlated with loneliness and self-esteem (Bérail et al., 2019; Tian et al., 2019). Although it is not reciprocal, these relationships are asserted to satisfy the need to belong, particularly for people who are lonely, socially depressed, or have social skills deficiencies, by replacing the absence of real relationships (Hartmann, 2016). The intensity of need and satisfaction is linked to levels of YouTube addiction (Baek et al., 2014).

Conceptual Framework

FoMO is defined as "a pervasive apprehension that others might be having rewarding experiences from which one is absent and characterised by the desire to stay continually connected with what others are doing" (Przybylski et al., 2013). The concept is suggested to have existed in various communication channels throughout history, such as newspapers, almanacks, letters, bulletins, etc., which are instrumental in satisfying curiosity about the lives of others (Wortham, 2011).

In line with technological advancements that have made it easier to send and receive information than ever before, social media applications have emerged, providing user-friendly and attractive services for sharing activities and experiences. One downside of this convenience is the increased anxiety level of individuals who fear missing out (Ellison et al., 2007). Mainly related to this self-reinforcing loop, three-quarters of young adults and nearly 70% of adults experience FoMO (JWTIntelligence, 2012; Przybylski et al., 2013).

The antecedents of FoMO are low self-esteem, loneliness, low needs satisfaction, lower feelings of acceptance and approval, and social anxiety, which foster high engagement in social media (Alt, 2015; Barry et al., 2017; Bérail et al., 2019; Blackwell et al., 2017; Hunt et al., 2018; Oberst et al., 2017). Studies have shown that individuals with high levels of FoMO are more likely to experience stress, anxiety, depression, sleeplessness, social media addiction, sleep problems, diminished well-being, and lower self-esteem (Adams et al., 2017; Alt, 2015; Baker et al., 2016; Beyens et al., 2016; Oberst et al., 2017). Additionally, high levels of FoMO are associated with eating disorders, poor mental health, and low self-esteem.

Taking previous studies into consideration, it can be asserted that using FoMO antecedents to examine related causal relationships is productive (Barry & Wong, 2020; Beyens et al., 2016; Blackwell et al., 2017; Elhai et al., 2018; Przybylski et al., 2013; Rozgonjuk et al., 2021; Stead & Bibby, 2017). Loneliness and self-esteem constructs are efficient in studying FoMO. One of the theoretically closest concepts to FoMO, loneliness, is shown to be correlated with self-reported FoMO. Together with loneliness, FoMO affects addictive social media engagement (Alt, 2015; Barry et al., 2017; Blackwell et al., 2017; Song et al., 2014). Although people tend to use social media applications to avoid or compensate for loneliness (Burke et al., 2010; De Cock et al., 2014; Xu & Tan, 2012), experimental research on the relationship between concepts indicates that limiting social media use significantly reduces both loneliness and FoMO, suggesting that these concepts work as self-reinforcing constructs that strengthen each other (Hunt et al., 2018).

The other research variable, self-esteem, is a result of individuals' positive or negative self-evaluations (Smith et al., 2015). Social media applications can be instrumental in self-assessment in various ways. The internet provides data to establish a basis for benchmarks that result in a subjective sense of self-worth or value. People gather information about others, develop frames of "norms," "standards," or "expected values" concerning family, wealth, achievements, success, well-being, life in general, and also compare themselves with others using content they find on the internet. Social media can also be used for external validation that affects self-esteem. Components such as wealth, fame, body image, belongings, lifest-yle, affiliations, etc., are subjected to both social comparison and external validation and are instrumental in forming self-esteem. Especially when self-worth is strongly related to others' approval or acceptance, social media applications can also serve as a tool to compensate for the lack of popularity and friendships in real life, helping to increase self-esteem. In line with this argument, research has revealed that social media can bridge social capital among college students who have low self-esteem (Steinfield et al., 2008).

Taking all into consideration, it can be asserted that people with lower self-esteem are prone to engage in social media more intensively (Báachnio et al., 2016; Hunt et al., 2018), and there is a negative correlation between self-esteem and FoMO (Barry & Wong, 2020; Buglass et al., 2017; Kalpidou et al., 2011).

Although listed among the most popular social media applications (Alexa, 2019), You-Tube is asserted to be fundamentally different from other platforms as it does not contain traditional social networking features and is thus seen more as a content provider (Kaplan & Haenlein, 2010; Kuss & Griffiths, 2017). There are various reasons for this differentiation.

First, YouTube focuses on content viewing, whereas others focus more on user interactions (Khan, 2017). Second, YouTube supports and offers longer videos called video blogs or "vlogs." Utilising this feature, almost anything can be shared in detail, from news to lectures, to individual experiences to art, with few length or file size limitations. Third, having the advantage of the ability to provide longer videos, YouTube appeals to a distinct user profile that can be categorised into two main groups: viewers and content creators. Although the first group is almost similar to other social media users, there is a great deal of diversity among the latter. Content creators include individuals, companies, interest groups, non-profit organisations, teachers, professional content developers, and hobbyists (Khan, 2017). Fourth, in relation to the reasons mentioned above, vlogs are more efficient at facilitating deeper and wide-ranging emotional expressions than other platforms (Tian et al., 2019). Fifth, the dynamics of the relationship between YouTubers and followers differ from other social media applications. YouTubers share their personal information, interests, opinions, experiences about life, products, services, etc., in other words, they engage in self-sharing activities to build a base of viewers and sustain their attention (Chen, 2016; Ferchaud et al., 2018). Viewers generally remain anonymous and watch content, making the relationship between YouTubers and viewers non-reciprocal. In line with the nature of this process, YouTube is more addictive for content creators than viewers (Balakrishnan & Griffiths, 2017). Sixth, compared to other social media platforms where people tend to share carefully arranged and pictured "positive," "happy," "healthy," "wealthy," "selected" moments of their lives, which presents an edited or filtered perception for others, YouTubers appear real and natural due to the format that allows them to share a great deal of information and emotions about themselves, intentionally or unintentionally (Tian et al., 2019).

It is theorised that YouTube can be used to create online relationships that function as surrogates for real relationships and help people overcome personal problems (Caplan & High, 2010; Ferris, 2001). If this is the case, then YouTube should be instrumental in strengthening self-esteem and reducing loneliness.

One of the biggest challenges for social media platforms is to allow users to stay longer or stick around on their websites or applications. The more time users spend on them, the better it is for the company in terms of visibility and profitability. One way to achieve this goal is to focus on developing features that can be used for satisfying the needs of users, thereby keeping them on the platform.

Stickiness is defined as a user's willingness to return to and prolong their visits to the platform or application (Teng, 2010). Research has revealed that continuance motivation and sharing behaviours are the main determinants of stickiness (Chiang and Hsiao, 2015). Sharing behaviour is related to individuals' willingness to share, and continuance motivation refers to individuals' continuous inner drive or motivation to share their experiences, ideas, feelings, thoughts, or content with others.

Users' needs, environmental, and personal factors such as social norms, identification, reputation, self-expression, interactivity, altruism, confidence, and self-efficacy have a significant impact on sharing behaviour (Chang and Chuang, 2011; Chiang and Hsiao, 2015; Choi and Chung, 2013; Compeau and Higgins, 1995; Hsu and Lin, 2008; Jiacheng et al., 2010;

Lee et al., 2011; Marakas et al., 1998; Park et al., 2011; Yang et al., 2010). Stickiness can be considered a habit (Wu et al., 2010).

Users gain intangible benefits—although sometimes they receive tangible benefits like ad revenues, sponsorships, or direct payments as well—in the form of appreciation and praise as viewers like, share, or comment on their posts. This mechanism not only builds reputation and self-esteem but also fosters addictive social media use by reinforcing the drive for it.

Considering all this, studying the relationships among self-esteem, loneliness, FoMO, YouTube addiction, and online stickiness concepts can contribute to the current literature. Based on the literature review, the research model presented in Figure 1 was developed, and it has four hypotheses to be tested.



Figure 1. Research Model

- H_1 : Self-esteem is related to FoMo.
- H₂: Loneliness is related to FoMo.
- *H*₃: *FoMo is related to YouTube addiction.*
- *H*₄: Youtube addiction is related to online stickiness.

Methodology

Instruments

Developed by Przybylski et al. (2013), the FoMO scale was designed as a 10-item, singledimension instrument that was evaluated on a 5-point Likert scale. Turkish adaptation was conducted by Gökler et al. (2016). In the initial validation study by Przybylski et al. (2013), the Cronbach's alpha reliability coefficient for the scale was found to be .90, indicating excellent reliability. In Gökler et al. (2016), $\alpha = .81$, reflecting strong internal consistency. Items included statements such as "It bothers me when I miss an opportunity to meet up with friends" and "I fear my friends have more rewarding experiences than me."

Rosenberg's Self-Esteem Scale was adapted to Turkish by Çuhadaroğlu (1986). The scale was originally developed as a 10-item, single-dimension instrument and is evaluated on a 5-point Likert scale. The Cronbach's alpha reliability coefficient in the initial validation study was .89 (Rosenberg, 1965), indicating high reliability. In the Turkish adaptation by Çuhadaroğlu (1986), $\alpha = .89$, thereby maintaining high reliability. Items included statements such as "On the whole, I am satisfied with myself" and "I feel that I am a person of worth, at least on an equal plane with others."

Loneliness Scale The Loneliness Scale used in this study was adapted from Hays and DiMatteo's (1987) short form of Russell et al.'s (1980) UCLA Loneliness Scale. The initial validation study consists of 20 items, but the short form used here contains eight items. The Cronbach's alpha reliability coefficient in the initial validation study by Russell et al. (1980) was.94, indicating excellent reliability. For the short form by Hays and DiMatteo (1987), α = .85, indicating good reliability. Items included statements such as "I feel excluded" and "I feel isolated from others."

The Online Stickiness Scale developed by Tsao (2014) consists of four items measured on a 5-point Likert scale. The initial validation study reported $\alpha = .79$, indicating acceptable reliability. This scale assesses user engagement and propensity to revisit online content. Items include statements such as "I would visit this website as often as I can" and "I intend to prolong my staying on this website."

YouTube Addiction Scale adapted from Moghavvemi et al. (2017). The Cronbach's alpha reliability coefficient in the initial validation study was .83, indicating good reliability. Items included statements such as "When I use YouTube, I lose track of time" and "Time goes by very quickly when I am using YouTube."

All statements in the adapted scales were evaluated on a five-point Likert scale ranging from 1 (Totally Disagree) to 5 (Totally Agree), except for the Loneliness Scale, which was evaluated on a scale ranging from 1 (Never) to 5 (Always).

Sampling and Data Collection

This study aimed to test the effects of self-esteem and loneliness on FoMO, which leads to YouTube addiction in the first phase and online stickiness in the second phase. The target population of this research is Turkish adults who use YouTube. An online Turkish questionnaire was developed using a survey tool, and the link was posted on various social media platforms for sampling. The study was approved by ethics committee of Altınbaş University on 25.04.2023 (register number: E-96136591-050.06.04-44095). And informed consent form was obtained from the participants for the study. The questionnaire was mainly divided into three sections: demographic profile questions (gender, marital status, education level, age and monthly income), YouTube usage behaviour of respondents (daily average time and number of videos watched on YouTube), and questions-statements related to the constructs in the research model. All participants voluntarily responded to the questionnaire, where a screening question was initially administered ("Do you watch YouTube videos?"). Those who answered this question positively were allowed to continue the survey, and a total of 1084 valid responses were collected during the period April 28 and May 11, 2023. Considering a 5% error margin and 95% confidence interval, the sample size exceeded the minimum sample size requirement.

Demographic Profile of the Sample

Table 1 presents the demographic profile of the sample. The mean age of the participants was 30.5 years with a standard deviation of 7.9 years, ranging between 18 and 70 years. A total of 54.7% of all participants (n=593) were in the 19-29 age group and 31.8% (n=345) were in the 30-39 age group; hence, the sample could be considered young. The majority of respondents were female (62.1%) and single (64.3%), and the sample was reported to be highly educated because most of them (83.8%) hold at least a Bachelor's degree. Most respondents (48%) had an income level around the minimum wage, which is parallel to Turkey's wage distribution. The sample's daily average time spent on YouTube varied; nevertheless, a majority of them (54.2%) indicated that they spend less than 1 hour on a daily basis, and 44.7% stated that they are watching 1 to 3 videos on average.

Table 1Demographic Profile of the Sample

Variable	Category	Frequency	Percentage	Variable	Category	Frequency	Percentage
	Female	673	62.1%		<5000 TL	520	48.0%
Gender	Male	409	37.7%	Monthly Income	5001-10,000 TL	375	34.6%
	Rather not to say	2	0.2%		10,001- 15,000 TL	105	9.7%
Marital	Married	387	35.7%		> 15,000 TLs	84	7.7%
Status	Single	697	64.3%				

Variable	Category	Frequency	Percentage	Variable	Category	Frequency	Percentage
	High school and less	90	8.3%		1	63	5.8%
Education	Vocational school	86	7.9%	Average number of	1-3	485	44.7%
	Bachelor	612	56.5%	videos on VauTuba	3-5	318	29.3%
	Master's	266	24.5%	rourube	5-8	114	10.5%
	PhD	30	2.8%		More than 8	104	9.6%
	19-29	19-29 593 54.7% Daily Ava	Daily Avora	Less than 1 h	588	54.2%	
Age Group	30-39	345	31.8%	ge Time on YouTube	1-3 hours	372	34.3%
-g. s P	40-49	110	10.1%		3-5 hours	88	8.1%
	50 and more	36	3.3%		5 h or more	36	3.3%

Findings

Validity and Reliability

Standard confirmatory factor analysis (CFA) procedures were performed using IBM SPSS Amos 26. The CFA results indicated an acceptable model fit ($_X^2(224)=873.810 \text{ p}<.001$; GFI=.936; AGFI=.915; CFI=.962; TLI=.953; RMSEA=.052). The assessment of the measurement model provided significant and sufficient convergent and discriminant validity and construct reliability. We followed Fornell-Larcker criterion for convergent and discriminant validity.

As stated in Table 2, the composite reliability (CR) of all constructs was greater than.75, indicating high reliability and evidence for convergent validity (Chin, 1998). The item factor loadings were all above .596, further confirming convergent validity (Bagozzi and Yi, 1988). Average variance extracted (AVE) scores exceeded the threshold of 0.50. Additionally, the constructs showed high internal consistency among the items with at least .75 Cronbach's alpha (α) score. The square roots of AVEs were greater than the correlation of constructs (see Table 3) and revealed an adequate level of discriminant validity (Fornell & Larcker, 1981).

Table 2					
Validity and Reliability of Measures					
Factors and Items	Factor Loadings				
<i>Self Esteem-Belief</i> (α=.92; CR=.91; AVE=.67)					
At times, I think I am not good at all.	.923				
On the whole, I am satisfied with myself.	.863				
I feel that I am'm a person of worth, at least on an equal plane with others.	.746				
I certainly feel useless at times.	.753				
I am able to do things as well as most other people.	.791				

Factors and Items	Factor Loadings
Online Stickiness (a =.92; CR=.92; AVE=.74)	
I would visit this website as often as I can.	.906
I intend to prolong my staying on this website.	.844
I would stay a longer time on the LINE than other applications.	.836
I intend to link to this website every time I am online.	.845
<i>Self Esteem - Affective</i> (α =.86; CR=.85; AVE=.53)	
I take a positive attitude toward myself.	.777
All in all, I am inclined to feel that I am a failure.	.763
I feel I do not have much to be proud of.	.751
I feel that I have a number of good qualities.	.749
I wish I could have more respect for myself.	.596
<i>Loneliness</i> ($\alpha = .81$; CR=.76; AVE=.51)	
I feel left out.	.705
I feel isolated from others.	.673
I am unhappy being so withdrawn.	.759
<i>FOMO-Social Experience</i> (α =.75; CR=.75; AVE=.50)	
It bothers me when I miss an opportunity to meet up with friends.	.689
It is important that I understand my friends "in jokes".	.675
When I miss out on a planned get together it bothers me.	.752
<i>FOMO-Rewarding Experience</i> (α =.89; CR=.89; AVE=.81)	
I fear others have more rewarding experiences than me.	.888
I fear my friends have more rewarding experiences than me.	.908
<i>YouTube Addiction</i> (α =.94; CR=.94; AVE=.89)	
When I use YouTube, I lose track of time.	.937
Time goes by very quickly when I am using YouTube.	.949
<i>GFI</i> =.936; <i>AGFI</i> =.915; <i>CFI</i> =.962; <i>TLI</i> =.953; <i>RMSEA</i> =.052; χ^2 (224)=873.810 p<.001	

CR: Composite Reliability, AVE: Average Variance Extracted, GFI: Goodness of Fit, AGFI: Adjusted Goodness of Fit Index, CFI: Comparative Fit Index, TLI: Tucker-Lewis Index

In the preliminary analysis, Pearson correlations were also examined (see Table 3). Apart from loneliness, all the constructs were significantly associated with YouTube addiction and online stickiness, supporting previous studies. In particular, the association between YouTube addiction and online stickiness (r=.721) was found to be remarkably high.

1 2 3 4 5 6 7 .818 1. Self Esteem - Belief 2. Online Stickiness .086** .858 3. Self Esteem – Affective -.138** .073* .730 4. Loneliness -.323** .041 .441** .713 5. FoMo - Social Experience .673** .165** .257** -.128** .706 .429** .170** 6. FoMo - Rewarding Experience .072* .169** .372** .898 .154** .721** .109** 7. YouTube Addiction .017 .230** .118** .943

Table 3Correlations and Square Roots of AVEs

Note: Diagonal values are the square roots of the AVEs: **p < .01 and *p < .05

Assessment of the Structural Model

The second procedure is the assessment of the structural model; thus, path analysis was conducted. The structural model was found to fit the data ($x^2(176)=763.800 \text{ p}<.001$; GFI=.936; AGFI=.916; CFI=.962; TLI=.955; RMSEA=.056), as all the fit indices obtained were satisfactory.

Table 4

Hypothesis Testing

Hypothesis and Related Paths	β	t	Std. β	Result			
H_{1a} : Self Esteem – Belief \rightarrow FoMo – Social Experience	.636	20.384**	.733	Supported			
H_{1b} : Self Esteem – Belief \rightarrow FoMo – Rewarding Experience	.134	4.783**	.149	Supported			
H_{1c} : Self Esteem – Affective \rightarrow FoMo – Social Experience	.421	11.135**	.376	Supported			
H_{1d} : Self Esteem – Affective \rightarrow FoMo – Rewarding Experience	.540	12.518**	.468	Supported			
H_{2a} : Loneliness \rightarrow FoMo – Social Experience	ns	-	ns	Not Supported			
H_{2b} : Loneliness \rightarrow FoMo – Rewarding Experience	ns	-	ns	Not Supported			
H_{3a} : FoMo – Social Experience \rightarrow YouTube Addiction	.302	6.912**	.241	Supported			
H_{3b} : FoMo – Rewarding Experience \rightarrow YouTube Addiction	ns	-	ns	Not Supported			
H_4 : YouTube Addiction \rightarrow Online Stickiness	.648	25.069**	.720	Supported			
GFI= .936; AGFI= .916; CFI= .962; TLI= .955; RMSEA= .056; ${}_{X}^{2}$ (176)=763.800 p<.001							

** p<.001; * p<.01; ns: not significant.

Table 4 and Figure 2 demonstrate the relationships of the paths, indicating that both selfesteem dimensions—belief and affective—had significant positive effects on FoMO dimensions—rewarding experience and social experience; thus, H₁ was fully supported. In detail, self-esteem-belief (β =.636, t=20.384, p<.001) and self-esteem-affective (β =.421, t=11.135, p<.001) had a significant effect on FoMO-social experience.

Furthermore, self-esteem-belief (β =.134, t=4.783, p<.001) and self-esteem-affective (β =.540, t=12.258, p<.001) had a significant effect on FoMO-rewarding experience.

Loneliness was found to have an insignificant effect on FoMO-social and FoMO-rewarding experiences; thus, H₂ was not supported. FoMO was found to be partially affecting You-Tube addiction. FoMO-social experience (β =.302, t=6.912, p<.001) had a significant effect on Youtube addiction, whereas FoMO-rewarding experience had an insignificant effect, H₃ was partially supported. Finally, YouTube addiction (β =.648, t=25.069, p<.001) as independent variable, had a significant effect on online stickiness, implying that H₄ was also supported.



Figure 2. Path Model

Conclusion and Discussion

The purpose of this study was to examine the relationships between YouTube addiction, online stickiness, and FoMO concepts through their predictors of self-esteem and loneliness. Therefore, the effects of self-esteem and loneliness on FoMO are tested in the first phase. After studying the antecedents of FoMO, its effects on YouTube addiction is examined. Finally, the link between YouTube addiction and stickiness is reviewed.

The first finding of the study is the positive and significant effect of self-esteem (affective and belief) on FoMO. Previous studies have suggested that self-evaluations, either cognitive or emotional, are related to needs such as approval, acceptance, and validation, which directly affect the level of self-esteem. In this sense, correlations found between self-esteem belief, self-esteem affective, FoMO-rewarding experience, and FoMO-social experience are likely to suggest that those needs tend to be verified, addressed, and satisfied through friendships and social comparisons. In line with the developments on the internet and telecommunication technologies, a growing industry of social media applications is providing efficient and user-friendly tools that can be used for satisfying those motives. Hence, considering that a significant portion of relationships (not only personal but also professional ones as well) are started, lived, or ended online in our times, it can be argued that social media can be taken as a "preferred venue" in which communication and relatedness needs are satisfied in one way or another. On the other hand, social media applications are also claimed to enhance the natural human drive to seek information as well (Gazzaley & Rosen, 2016). To identify opportunities, avoid threads, understand what is going on, what others are doing, protect themselves from risks, decide their place in the social hierarchy, etc. people seek information constantly even though they cannot possibly process all the data out there (Whiting & Williams, 2013; Roberts et al., 2014; Abel et al., 2016). One other point of view that may contribute to the discussion is that, according to social comparison theory, individuals determine their own personal worth based on comparisons with others (Festinger, 1954). In other words, people compare themselves with others and establish a sense of self-value according to the results (Abel et al., 2016). Social media is providing sufficient—even at an annoying level—accessible opportunities to obtain information about others that can be used for this purpose (Roberts & David, 2020). Easy access to technology and information encourages people to compare their lives with others, which generally adversely affects psychological well-being negatively (Tian et al., 2019). In line with the results of previous studies, considering the effects of connectedness and communication needs, the drive for assessing self-worth, and information collection as a basis for subjective self-esteem perception, it can be said that lower self-esteem and poor self-image is positively correlated with excessive use of social media (Hunt et al., 2018), creating concerns about disconnecting with others and not experiencing rewarding activities that others enjoy, hence inducing FoMO. This study's results show that, especially for those whose self-esteem is dependent on what others do or if others' experiences are more pleasant than theirs, the concept functions as a reinforcer for FoMO. In such cases, vulnerable or dependent self-esteem structures can be theorised. Current literature on self-esteem tends to focus mostly on the self-assessment and self-evaluation aspects of the concept. Results of this study offer a different perspective, suggesting a positive relationship between self-esteem and FoMO that may urge adoption of a new approach to explain the concept better. In light of the findings mentioned above, studies on high and low self-esteem and the relationship between FoMO can be suggested for further investigation.

Loneliness did not have a significant effect on both dimensions of FoMO. Although study results indicate correlations between loneliness and FoMO in line with the literature, no significant effect was found (Alt, 2015; Barry et al., 2017; Blackwell et al., 2017; Burke et al., 2010; De Cock et al., 2014; Song et al., 2014; Xu & Tan, 2012). Specifically, social experience is negatively correlated to loneliness, whereas rewarding experience is positively correlated to FoMO. Looking deeper into the questionnaire items, social experience represents staying up to date or being on the same page with friends; hence, the negative correlation found with FoMO is self-explanatory. On the other hand, rewarding experiences represent fear of others having better payoffs in terms of experiences, and loneliness is found to strengthen this feeling. Nevertheless, loneliness was not found to have a significant effect on both FoMO social and rewarding experiences. Considering previous findings suggesting that limiting social media engagement reduces loneliness and FoMO, it can be theorised that the relationship between constructs is reciprocal but not unilateral. Therefore, further studies are suggested to take loneliness as a complementary construct rather than an antecedent for the FoMO concept.

Another finding of this research is the significant effect of FoMO Social Experience on YouTube addiction. YouTube is fundamentally different from other platforms since the relationship between YouTubers and viewers is generally non-reciprocal. An interesting fact about engaging in social media is that to satisfy the need to belong, efforts do not need to be mutual. In other words, even following or sharing without the other party doing the same thing seems to be enough to feel good and involved. This mechanism tends to create an unorthodox form of relationship called a "parasocial relationship" in which followers consume media about someone they do not actually know or meet before (Horton & Wohl, 1956; Dibble, Hartmann, & Rosaen, 2016). In an attempt to compensate for their lack of real relationships, individuals develop bond parasocial connections to satisfy their need to belong (Horton & Wohl, 1956; Hartmann, 2016). Considering that the human brain is flawed in terms of distinguishing between real friends and parasocial relationships, it can be argued that even one-way connections can be substitutes for real relationships to satisfy social needs (Kanazawa, 2002). In line with the literature, this study's findings indicate that FoMO social experiences induce YouTube addiction. In an aspect, it can be asserted that YouTube is providing a substitute for social relationships even if it is claimed as "parasocial" and from the other perspective, it can be said that online YouTube meetings satisfy social needs, and FoMO can be reduced by attending such gatherings. Another interesting finding is that although both social and rewarding experiences are positively correlated with FoMO, the latter does not have a significant effect on YouTube addiction. YouTube is reported to be more addictive for content creators than for viewers (Balakrishnan & Griffiths, 2017). Therefore, it can be asserted that regular users who are "just watching" are indifferent in terms of feeling rewarded simply because of the non-reciprocal nature of the application. On the other hand, since all the passively engaged users have the same experience, it is not possible to increase the individual payoff in terms of rewarding experiences due to the one-way nature of YouTube; thus, everyone has the same rewards, and no one is better than others. An important point is that the dataset of the study does not have a YouTube viewer-content creator distinction; therefore, it is not possible to conduct difference analyses. It can be suggested for future studies to differentiate the two groups to further enlighten the subject.

Finally, YouTube addiction was found to have not only a strong correlation but also a significant effect on online stickiness, as expected. Literature suggests that stickiness is a habitual behaviour (Wu et al., 2010) that is strengthened by various tangible and intangible benefits, such as ego gratification, identification, reputation building, sponsorships, and direct advertisement. This study asserts that self-esteem, FoMO, and YouTube addiction constructs affect online stickiness, which requires further research.

This study examined the relationships among self-esteem, loneliness, FoMO, YouTube addiction, and online stickiness concepts. Findings indicated positive effects of self-esteem on FoMO, but loneliness was not found to affect the construct. On the other hand, FoMO social experience has shown a positive effect on YouTube Addiction and YouTube addiction on online stickiness. The positive effect of self-esteem on FoMO is particularly interesting and requires further research, especially taking needs satisfaction, self-value, and vulnerable-dependent self-esteem concepts into consideration. Although correlations were found between loneliness and FoMO, no direct effect was observed. Therefore, the relationship between loneliness and FoMO might present another fertile area for further research. Finally, studies on interactions between specific YouTube features and parasocial relationships may also contribute to the growing literature.

Limitations

This study is not without limitations. First, self-report measures are used to obtain the data. Use of different, more sensitive scales can be suggested to compare and contrast the results of this study. Second, although the sample size is adequate, using a different dataset with participants from different countries may increase the generalizability of the results as well as enable cross-cultural comparisons. Third, given this is a cross-sectional study, longitudinal research can help explore exploring the causal relationships between the concepts.

References

- Abel, J. P., Buff, C. L., & Burr, S. A. (2016). Social media and the fear of missing out: Scale development and assessment. *Journal of Business & Economics Research (JBER)*, 14(1), 33-44. Retrieved from https:// journals.klalliance.org/index.php/JBER/article/view/192
- Adams, S. K., & Kisler, T. S. (2013). Sleep quality as a mediator between technology-related sleep quality, depression, and anxiety. *Cyberpsychology, Behavior, and Social Networking*, 16(1), 25–31. https://doi. org/10.1089/cyber.2012.0157
- Adams, S. K., Williford, D. N., Vaccaro, A., Kisler, T. S., & Francis, A., & Newman, B. (2017). The young and the restless: socializing trumps sleep, fear of missing out, and technological distractions in first-year college students. *International Journal of Adolescence and Youth*, 22(3), 337–348. https://doi.org/10.108 0/02673843.2016.1181557
- Alexa (2019). The top 500 sites on the web. Retrieved from http://www.alexa.com/topsites on March 24, 2019.
- Alt, D. (2015). College students' academic motivation, media engagement and fear of missing out. Computers in Human Behavior, 49, 111–119. http://dx.doi.org/10.1016/j.chb.2015.02.057.
- Aydın, H. (2018). Sosyal Medyadaki Gelişmeleri Kaçırma Korkusunun (FOMO) İçgüdüsel Alımlara Etkisinin Öz Belirleme Kuramı İle Açıklanması. Uluslararası İktisadi ve İdari İncelemeler Dergisi 415-426. https://doi.org/10.18092/ulikidince.439179
- Aydın, H. (2022). FOMO, YOLO ve JOMO Sosyal Medya Hastalıklarının Tüketim Bağlamında Değerlendi-

Peer-review: Externally peer-reviewed.

Conflict of Interest: The authors have no conflict of interest to declare.

Grant Support: The authors declared that this study has received no financial support.

Author Contributions: Conception/Design of study: B.D., H.E, M.B.; Data Acquisition: B.D., H.E.; Data Analysis/Interpretation: B.D., H.E.; Drafting Manuscript: M.B.; Critical Revision of Manuscript: B.D., H.E, M.B.; Final Approval and Accountability: B.D., H.E, M.B.

Ethical Approval: The study was approved by ethics committee of Altınbaş University on 25.04.2023 (register number: E-96136591-050.06.04-44095).

Informed consent: Informed consent form was obtained from the participants for the study.

rilmesi Üzerine Literatür Taraması. Uluslararası İktisadi ve İdari İncelemeler Dergisi (37), 19-34. https://doi.org/10.18092/ulikidince.1039467

- Baachnio, A., Przepiorka, A., & Rudnicka, P. (2016). Narcissism and self-esteem as predictors of dimensions of Facebook use. *Personality and Individual Differences*, 90, 296-301. https://doi.org/10.1016/j.paid.2015.11.018
- Backlinko, 2021. Retrieved from 29 May 2022 https://backlinko.com/social-media-users
- Baek, Y. M., Cho, Y., & Kim, H. (2014). Attachment style and its influence on the activities, motives, and consequences of SNS use. *Journal of Broadcasting & Electronic Media*, 58(4), 522–541. https://doi.org/ 10.1080/08838151.2014.966362
- Bagozzi, R. P., & Yi, Y. (1988). On the evaluation of structural equation models. *Journal of The Academy of Marketing Science*, 16(1), 74-94.
- Baker, Z. G., Krieger, H., & LeRoy, A. S. (2016). Fear of missing out: Relationships with depression, mindfulness, and physical symptoms. *Translational Issues in Psychological Science*, 2, 275–282. https://doi. org/10.1037/tps0000075
- Balakrishnan, J., & Griffiths, M.D. (2017). Social media addiction: What is the role of content in YouTube? Journal of Behavioral Addictions, 6, 364-377. https://doi.org/10.1556/2006.6.2017.058
- Barry, C. T., Sidoti, C. L., Briggs, S. M., Reiter, S. R., & Lindsey, R. A. (2017). Adolescent social media use and mental health from adolescent and parent perspectives. *Journal of Adolescence*, 61, 1–11. doi:10.1016/j.adolescence.2017.08.005https://doi.org/10.1016/j.adolescence.2017.08.005
- Barry, C. T., & Wong, M. Y. (2020). Fear of missing out (FoMO): A generational phenomenon or an individual difference?. *Journal of Social and Personal Relationships*, 37(12), 2952-2966. https://doi. org/10.1177/0265407520945394
- Bérail, P., Guillon, M., & Bungener, C. (2019). The relations between YouTube addiction, social anxiety and parasocial relationships with YouTubers: A moderated-mediation model based on a cognitive-behavioral framework. *Computers in Human Behavior*, 99, 190-204. https://doi.org/10.1016/j.chb.2019.05.007
- Beyens, I., Frison, E., & Eggermont, S. (2016). I don't want to miss a thing": Adolescents' fear of missing out and its relationship to adolescents' social needs, Facebook use, and facebook related stress. *Computers in Human Behavior*, 64, 1–8. https://doi.org/10.1016/j.chb.2016.05.083
- Blackwell, D., Leaman, C., Tramposch, R., Osborne, C., & Liss, M. (2017). Extraversion, neuroticism, attachment style and fear of missing out as predictors of social media use and addiction. *Personality and Individual Differences*, 116, 69–72. https://doi.org/10.1016/j.paid.2017.04.039
- Blai, B. (1989). Health consequences of loneliness: A review of the literature. Journal of American College Health, 37, 162-167. https://doi.org/10.1080/07448481.1989.9938410
- Buglass, S. L., Binder, J. F., Betts, L. R., & Underwood, J. D. M. (2017). Motivators of online vulnerability: The impact of social network site use and FOMO. *Computers in Human Behavior*, 66, 248–255. http:// dx.doi.org/10.1016/j.chb.2016.09.055
- Burke, M., Marlow, C. & Lento, T. (2010). Social network activity and social well- being. *Postgraduate Medical Journal*, 85, 455–459. https://doi.org/10.1145/1753326.1753613
- Caplan, S. E., & High, A. C. (2010). Online social interaction, psychosocial well-being, and problematic Internet use. In K. Young & C. Nabuco de Abreu (Eds.). *Internet addiction: A handbook for evaluation and treatment* (pp. 35–53). New York, NY: John Wiley & Sons. https://doi.org/10.1002/9781118013991.ch3
- Chang, H.H. and Chuang, S.S. (2011), "Social capital and individual motivations on knowledge sharing:

participant involvement as a moderator", Information & Management, Vol. 48 No. 1, pp. 9-18. https://doi. org/10.1016/j.im.2010.11.001

- Chen, C. P. (2016). Forming digital self and parasocial relationships on YouTube. *Journal of Consumer Culture, 16*(1), 232–254. https://doi.org/10.1177/1469540514521081
- Chiang, H.S., and Hsiao, K.L. (2015). YouTube stickiness: the needs, personal, and environmental perspective. *Internet Research*, 25(1), 85-106. https://doi.org/10.1108/IntR-11-2013-0236
- Chin, W. W. (1998). The partial least squares approach to structural equation modeling. In G. A. Marcoulides (Ed.), *Modern Methods for Business Research* (pp. 295-336). Lawrence Erlbaum Associates Publishers.
- Choi, G., and Chung, H. (2013). Applying the technology acceptance model to social networking sites (SNS): Impact of subjective norm and social capital on the acceptance of SNS. *International Journal of Human-Computer Interaction*, 29(10), 619-628. https://doi.org/10.1080/10447318.2013.780449
- Christofides, E., Muise, A., & Desmarais, S. (2009). Information disclosure and control on Facebook: Are they two sides of the same coin or two different processes? *Cyberpsychology, Behavior and Social Networking*, 12, 341–345. https://doi.org/10.1089/cpb.2008.0226
- Compeau, D. R., and Higgins, C. A. (1995). Computer self-efficacy: Development of a measure and initial test. MIS Quarterly, 19(2), 189-211. https://doi.org/10.2307/249688
- Cuhadaroğlu, F. (1986). Adölesanlarda benlik saygısı [Self-esteem in adolescents]. [Unpublished specialist thesis]. Hacettepe University.
- Deci, E. I., & Ryan, R. M. (1985). Intrinsic motivation and self-determination in human behavior. Plenum. https://doi.org/10.1007/978-1-4899-2271-7
- De Cock, R., Vangeel, J., Klein, A., Minotte, P., Rosas, O., & Meerkerk, G. J. (2014). Compulsive use of social networking sites in Belgium: Prevalence, profile, and the role of attitude toward work and school. *Cyberpsychology, Behavior, and Social Networking, 17*(3), 166–171. https://doi.org/10.1089/ cyber.2013.0029
- Dhir, A., Yossatorn, Y., Kaur, P., & Chen, S. (2018). Online social media fatigue and psychological wellbeing—A study of compulsive use, fear of missing out, fatigue, anxiety and depression. *International Journal of Information Management*, 40, 141-152. https://doi.org/10.1016/j.ijinfomgt.2018.01.012
- Dibble, J. L., Hartmann, T., & Rosaen, S. F. (2016). Parasocial interaction and parasocial relationship: Conceptual clarification and a critical assessment of measures. *Human Communication Research*, 42(1), 21-44. https://doi.org/10.1111/hcre.12063
- Donnelly, E., & Kuss, D. J. (2016). Depression among users of Social Networking Sites (SNSs): The role of SNS addiction and increased usage. *Journal of Addiction and Preventative Medicine*, 1, 1–6. https://doi. org/10.19104/japm.2016.107
- Elhai, J. D., Levine, J. C., Dvorak, R. D., & Hall, B. J. (2016). Fear of missing out, need for touch, anxiety and depression are related to problematic smartphone use. *Computers in Human Behavior*, 63, 509–516. https://doi.org/10.1016/j.chb.2016.05.079
- Elhai, J. D., Levine, J. C., Alghraibeh, A. M., Alafnan, A. A., Aldraiweesh, A. A., & Hall, B. J. (2018). Fear of missing out: Testing relationships with negative affectivity, online social engagement, and problematic smartphone use. *Computers in Human Behavior*, 89, 289–298. https://doi.org/10.1016/j.chb.2018.08.020
- Elhai, J. D., Yang, H., & Montag, C. (2020). Fear of missing out (FOMO): Overview, theoretical underpinnings, and literature review on relations with severity of negative affectivity and problematic technology use. *Brazilian Journal of Psychiatry*. https://doi.org/10.1590/1516-4446-2020-0870

- Ellison, N.B., Steinfield, C., & Lampe, C. (2007). The benefits of Facebook "friends": Social capital and college students' use of online social network sites. *Journal of Computer-Mediated Communication*, 12, 1143–1168. https://doi.org/10.1111/j.1083-6101.2007.00367.x
- Ercengiz, M. (2020). Dürtüsellik ile Sosyal Medya Aşerme Arasındaki İlişkide FOMO'nun Aracılık Etkisi. OPUS International Journal of Society Researches, 15(22), 854-877. https://doi.org/10.26466/ opus.648795
- Erciş, A., Deligöz, K., & Mutlu, M. (2021). Öğrencilerin FOMO ve Plansiz Satın Alma Davranışları Üzerine Bir Uygulama. Atatürk Üniversitesi İktisadi ve İdari Bilimler Dergisi, 35(1), 219-243. https://doi. org/10.16951/atauniiibd.772544
- Ferchaud, A., Grzeslo, J., Orme, S., & LaGroue, J. (2018). Parasocial attributes and YouTube personalities: Exploring content trends across the most subscribed YouTube channels. *Computers in Human Behavior*, 80, 88–96. https://doi.org/10.1016/j.chb.2017.10.041
- Festinger, L. (1954). A theory of social comparison processes. Human Relations, 7(117). https://doi. org/10.1177/001872675400700202
- Fornell, C., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, 18(1), 39-50. https://doi.org/10.2307/3151312
- Gazzaley, A., & Rosen, L. D. (2016). The distracted mind: Ancient brains in a high-tech world. MIT Press. https://doi.org/10.7551/mitpress/10414.001.0001
- Gokler, M. E., Aydin, R., Unal, E., & Metintas, S. (2016). Determining validity and reliability of Turkish version of Fear of Missing out Scale. *Anadolu Psikiyatri Dergisi*, 17(S1), 53-60.
- Hartmann, T. (2016). Parasocial interaction, parasocial relationships, and well-being. In L. Reinecke, & M.
 B. Oliver (Eds.), *The Routledge Handbook of Media Use and Well-being: International Perspectives on Theory and Research on Positive Media Effects* (pp. 131–144). Routledge.
- Hays, R. D., & DiMatteo, M. R. (1987). A short-form measure of loneliness. *Journal of personality assessment*, 51(1), 69-81. https://doi.org/10.1207/s15327752jpa5101_6
- Horton, D., & Richard Wohl, R. (1956). Mass communication and para-social interaction: Observations on intimacy at a distance. *Psychiatry*, 19(3), 215-229. https://doi.org/10.1080/00332747.1956.11023049
- Hsu, C.L., & Lin, J.C.C. (2008). Acceptance of blog usage: The roles of technology acceptance, social influence and knowledge sharing motivation. *Information & Management*, 45(1), 65-74. https://doi. org/10.1016/j.im.2007.11.001
- Hunt, M. G., Marx, R., Lipson, C., & Young, J. (2018). No more FOMO: Limiting social media decreases loneliness and depression. *Journal of Social and Clinical Psychology*, 37(10), 751-768. https://doi.org/10.1521/jscp.2018.37.10.751
- Jiacheng, W., Lu, L., & Francesco, C.A. (2010). A cognitive model of intra-organizational knowledge-sharing motivations in the view of cross-culture. *International Journal of Information Management*, 30(3), 220-230. https://doi.org/10.1016/j.ijinfomgt.2009.08.007
- JWTIntelligence. (2012, March). Fear of missing out (FOMO). Retrieved June 2012, from JWT: http://www. jwtintelligence.com/wp-content/uploads/2012/03/F_JWT_FOMO-update_3.21.12.pdf
- Kalpidou, M., Costin, D., & Morris, J. (2011). The relationship between Facebook and the well-being of undergraduate college students. *Cyberpsychology, Behavior, and Social Networking*, 14, 183–189. https:// doi.org/10.1089/cyber.2010.0061
- Kanazawa, S. (2002). Bowling with our imaginary friends. Evolution and Human Behavior, 23(3), 167-171.

https://doi.org/10.1016/S1090-5138(02)00093-5

- Kaplan, A. M., & Haenlein, M. (2010). Users of the world, unite! The challenges and opportunities of Social Media. Business Horizons, 53(1), 59–68. https://doi.org/10.1016/j.bushor.2009.09.003
- Kardefelt-Winter, D., De Kaminski, M., Dunkels, E., & Hydén, H. (2014). Dataspelande inte orsak till dåliga skolresultaten. Dagens Nyheter (DN debatt).
- Khan, M. L. (2017). Social media engagement: What motivates user participation and consumption on You-Tube? *Computers in Human Behavior, 66*, 236–247. https://doi.org/10.1016/j.chb.2016.09.024
- Kim, W., Jeong, O.-R., & Lee, S.-W. (2010). On social Web sites. *Information Systems*, 35, 215-236. https:// doi.org/10.1016/j.is.2009.08.003
- Kross, E., Verduyn, P., Park, J., Seungjae, D., Lin, N., Shablack, & Ybarra, O. (2013). Facebook use predicts declines in subjective well-being in young adults. *PLoS One*, 8(8), e69841. https://doi.org/10.1371/ journal.pone.0069841
- Kurtoğlu, E., & Aydın, H. (2023). Sosyal Medya Bağımlılığının ve Gelişmeleri Kaçırma Korkusunun Tüketim Davranışlarına Etkileri Üzerine Bir Araştırma. Yaşar Üniversitesi E-Dergisi, 18(72), 356-379. https:// doi.org/10.19168/jyasar.1281329
- Kuss, D. J., & Griffiths, M. D. (2017). Social networking sites and addiction: Ten lessons learned. International Journal of Environmental Research and Public Health, 14(3), 311. https://doi.org/10.3390/ ijerph14030311
- Leary, M., & Baumeister, R. (2000). The nature and function of self-esteem: Sociometer theory. In M. P. Zanna (Ed.), Advances in Experimental Social Psychology (Vol. 32, pp. 1-62). Academic Press. https://doi.org/10.1016/S0065-2601(00)80003-9
- Lee, D., Park, Y., Kim, J., Kim, J., & Moon, J. (2011). Understanding music sharing behaviour on social network services. Online Information Review, 35(5), 716-733. https://doi.org/10.1108/14684521111176452
- Lu, H., Liang, G., Li, D. L., Liu, M. X., Yin, Z. J., Li, Y. Z., ... Pan, C. W. (2023). Sleep quality as a mediator of the relationship between screen time and negative emotions among Chinese college freshmen. *Psychology, Health & Medicine, 28*(10), 2883–2896. https://doi.org/10.1080/13548506.2023.2217380
- Lup, K., Trub, L., & Rosenthal, L. (2015). Instagram #Instasad?: Exploring associations among Instagram use, depressive symptoms, negative social comparison, and strangers followed. *Cyberpsychology, Beha*vior, and Social Networking, 18, 247–252. https://doi.org/10.1089/cyber.2014.0560
- Marakas, G.M., Yi, M.Y., & Johnson, R.D. (1998). The multilevel and multifaceted character of computer self-efficacy: Toward clarification of the construct and an integrative framework for research. *Information Systems Research*, 9(2), 126-163. https://doi.org/10.1287/isre.9.2.126
- Moghavvemi, S., Sulaiman, A. B., Jaafar, N. I. B., & Kasem, N. (2017, July). Facebook and YouTube addiction: The usage pattern of Malaysian students. In 2017 International Conference on Research and Innovation in Information Systems (ICRIIS) (pp. 1-6). IEEE. https://doi.org/10.1109/ICRIIS.2017.8002453
- Nadkarni, A., & Hofmann, S. G. (2012). Why do people use Facebook? Personality and Individual Differences, 52, 243–249. https://doi.org/10.1016/j.paid.2011.11.007
- Oberst, U., Wegmann, E., Stodt, B., Brand, M., & Chamarro, A. (2017). Negative consequences from heavy social networking in adolescents: The mediating role of fear of missing out. *Journal of Adolescence*, 55, 51–60. https://doi.org/10.1016/j.adolescence.2016.12.008
- Özdemir, S. (2021). Sosyal İlişkilerde Gelişmeleri Kaçırma Korkusu Sosyal Dışlanmayı Etkiler Mi? Sosyotelizmin Rolü. *Hacettepe Üniversitesi Sosyal Bilimler Dergisi*, 3(1), 64-77.

- Park, N., Jin, B., & Jin, S.A. (2011). Effects of self-disclosure on relational intimacy in Facebook. Computers in Human Behavior, 27(5), 1974-1983. https://doi.org/10.1016/j.chb.2011.05.004
- Pollard, G. (2012). JWT explores fear of missing out –report SXSW presentation spotlight how brands can leverage FoMOCision. PRWeb. Retrieved from http://www.prweb.com/releases/2012/3/prweb9265660. htm
- Przybylski, A. K., Murayama, K., DeHaan, C. R., & Gladwell, V. (2013). Motivational, emotional, and behavioral correlates of fear of missing out. *Computers in Human Behavior*, 29(4), 1841-1848. https://doi. org/10.1016/j.chb.2013.02.014
- Quan-Haase, A., & Young, A. L. (2010). Uses and gratifications of social media: A comparison of Facebook and instant messaging. *Bulletin of Science, Technology & Society, 30*, 350–361. https://doi. org/10.1177/0270467610380009
- Quinn, K. (2016). Why We Share: A uses and gratifications approach to privacy regulation in social media use. *Journal of Broadcasting & Electronic Media*, 60, 61-68. https://doi.org/10.1080/08838151.2015.11 27245
- Roberts, J. A., Petnji YaYa, L. H., & Manolis, C. (2014). The invisible addiction: Cell-phone activities and addiction among male and female college students. *Journal of Behavioral Addictions*, 3(4), 254–265. https://doi.org/10.1556/JBA.3.2014.015
- Roberts, J. A., & David, M. E. (2020). The social media party: Fear of missing out (FoMO), social media intensity, connection, and well-being. *International Journal of Human–Computer Interaction*, 36(4), 386-392. https://doi.org/10.1080/10447318.2019.1646517
- Rook, K. S. (1984). Promoting social bonding: Strategies for helping the lonely and socially isolated. American Psychologist, 39(12), 1389-1407. https://doi.org/10.1037/0003-066X.39.
- Rosen, L. D., Whaling, K., Rab, S., Carrier, L. M., & Cheever, N. A. (2013). Is Facebook creating "iDisorders"? The link between clinical symptoms of psychiatric disorders and technology use, attitudes and anxiety. *Computers in Human Behavior*, 29, 1243–1254. https://doi.org/10.1016/j.chb.2012.11.012
- Rozgonjuk, D., Sindermann, C., Elhai, J. D., & Montag, C. (2020). Fear of missing out (FoMO) and social media's impact on daily-life and productivity at work: Do WhatsApp, Facebook, Instagram and Snapchat use disorders mediate that association? *Addictive Behaviors*, 110, 106487. https://doi.org/10.1016/j. addbeh.2020.106487
- Rozgonjuk, D., Sindermann, C., Elhai, J. D., & Montag, C. (2021). Individual differences in Fear of Missing Out (FoMO): Age, gender, and the Big Five personality trait domains, facets, and items. *Personality and Individual Differences*, 171, 110546. https://doi.org/10.1016/j.paid.2020.110546
- Russell, D., Peplau, L. A., & Cutrona, C. E. (1980). The revised UCLA Loneliness Scale: Concurrent and discriminant validity evidence. *Journal of Personality and Social Psychology*, 39(3), 472. https://doi. org/10.1037/0022-3514.39.3.472
- Sampasa-Kanyinga, H., & Lewis, R. F. (2015). Frequent use of social networking sites is associated with poor psychological functioning among children and adolescents. *Cyberpsychology, Behavior and Social Networking, 18*, 380–385. https://doi.org/10.1089/cyber.2015.0055
- Sarıbay, B., & Durgun, G. (2020). Dokunmatik Toplumların Fobileri: Netlessfobi, Nomofobi ve FOMO Kavramları Üzerine Bir İçerik Analizi Çalışması. Yaşar Üniversitesi E-Dergisi, 15, 280-294.
- Seidman, G. (2013). Self-presentation and belonging on Facebook: How personality influences social media use and motivations. *Personality and Individual Differences*, 54, 402–407. https://doi.org/10.1016/j.

paid.2012.10.009

Smith, H. E., Mackie, D., & Claypool, H.M. (2015). Social Psychology (3rd ed.). Psychology Press.

- Song, H., Zmyslinski-Seelig, A., Kim, J., Drent, A., Victor, A., Omori, K., & Allen, M. (2014). Does Facebook make you lonely?: A meta-analysis. *Computers in Human Behavior*, 36, 446–452. https://doi. org/10.1016/j.chb.2014.04.011
- Stapleton, P., Luiz, G., & Chatwin, H. (2017). Generation validation: The role of social comparison in use of Instagram among emerging adults. *Cyberpsychology, Behavior, and Social Networking*, 20(3), 142-149. https://doi.org/10.1089/cyber.2016.0444
- Statista, 2024. Retrieved from: https://www.statista.com/statistics/617136/digital-population-worldwide/ on 20/06/2024.
- Stead, H., & Bibby, P. A. (2017). Personality, fear of missing out and problematic internet use and their relationship to subjective well-being. *Computers in Human Behavior*, 76, 534–540. https://doi.org/10.1016/j. chb.2017.08.016
- Steinfield, C., Ellison, N. B., & Lampe, C. (2008). Social capital, self-esteem, and use of online social network sites: A longitudinal analysis. *Journal of Applied Developmental Psychology*, 29(6), 434–445. https://doi.org/10.1016/j.appdev.2008.07.002
- Sundberg, C. P. (1988). Loneliness: Sexual and racial differences in college freshman. *Journal of College Student Development, 29*, 298-305.
- Tandoc, E. C., Jr., Ferrucci, P., & Duffy, M. (2015). Facebook use, envy, and depression among college students: Is Facebooking depressing? *Computers in Human Behavior*, 43, 139–146. https://doi.org/10.1016/j. chb.2014.10.053
- Teke, D., & Yılmaz, T. (2024). Sosyal Ortamlardaki Gelişmeleri Kaçirma Korkusu (Fomo) ve İlişkili Faktörlerin Değerlendirilmesi: Lisans Öğrencileri Üzerine Bir Alan Araştırması. Yönetim Bilimleri Dergisi, 22(51), 1-24. https://doi.org/10.35408/comuybd.1253419
- Teng, C.-I. (2010). Customization, immersion satisfaction, and online gamer loyalty. Computers in Human Behavior, 26(6), 1547-1554. https://doi.org/10.1016/j.chb.2010.06.006
- Tian, K., Zhu, H., & Guang, T. (2019). The Moderating Role of Social Media Platforms on Self-Esteem and Life Satisfaction: A Case Study of YouTube and Instagram. *The Journal of Applied Business and Economics*, 21(6), 208-216. https://doi.org/10.33423/jabe.v21i6.2408
- Turel, O., & Serenko, A. (2012). The benefits and dangers of enjoyment with social networking websites. European Journal of Information Systems, 21(5), 512–528. https://doi.org/10.1057/ejis.2012.1
- Toma, C. L., & Hancock, J. T. (2013). Self-affirmation underlies Facebook use. Personality & Social Psychology Bulletin, 39, 321-331. https://doi.org/10.1177/0146167212474694
- Tsao, W. Y. (2014). Enhancing competitive advantages: The contribution of mediator and moderator on stickiness in the LINE. *Journal of Retailing and Consumer Services*, 21(6), 933-941. https://doi.org/10.1016/j. jretconser.2014.08.016
- Utz, S., Tanis, M., & Vermeulen, I. (2012). It is all about being popular: The effects of need for popularity on social network site use. *Cyberpsychology, Behavior and Social Networking*, 15, 37–42. https://doi. org/10.1089/cyber.2010.0651
- Wang, W., Yang, Y., & Yang, X. (2021). Perceived stress and smartphone addiction in medical college students: the mediating role of negative emotions and the moderating role of psychological capital. *Frontiers* in Psychology, 12, 660234. https://doi.org/10.3389/fpsyg.2021.660234

- Whiting, A., & Williams, D. (2013). Why people use social media: A uses and gratifications approach. *Qualitative Market Research: An International Journal*, 16(4), 362–369. https://doi.org/10.1108/QMR-06-2013-0041
- Winter, S., Neubaum, G., Eimler, S. C., Gordon, V., Theil, J., Herrmann, J., et al. (2014). Another brick in the Facebook wall - how personality traits relate to the content of status updates. *Computers in Human Behavior*, 34, 194–202. https://doi.org/10.1016/j.chb.2014.01.048
- Wortham, J. (2011, April 9). Feel like a wallflower? Maybe it's your Facebook wall. *The New York Times*. Retrieved May 2012, from http://www.nytimes.com/2011/04/10/business/10ping.html
- Woods, H. C., & Scott, H. (2016). Sleepyteens: Social media use in adolescence is associated with poor sleep quality, anxiety, depression and low self-esteem. *Journal of Adolescence*, 51, 51, 41–49. https://doi. org/10.1016/j.adolescence.2016.05.008
- Wu, J.H., Wang, S.C., & Tsai, H.H. (2010). Falling in love with online games: The uses and gratifications perspective. Computers in Human Behavior, 26(6), 1862-1871. https://doi.org/10.1016/j.chb.2010.07.033
- Xu, H., & Tan, B. C. (2012). Why do I keep checking Facebook: Effects of message characteristics on the formation of social network services addiction. *Cyberpsychology, Behavior and Social Networking*, 15(2), 88-92. https://doi.org/10.1089/cyber.2010.0061
- Yang, C., Hsu, Y.-C., & Tan, S. (2010). Predicting the determinants of users' intentions for using YouTube to share video: Moderating gender effects. *CyberPsychology, Behavior & Social Networking*, 13(2), 141-152. https://doi.org/10.1089/cyber.2009.0105
- YouTube. (2021) Retrieved on March 21, 2023, from https://www.youtube.com/intl/tr/about/press/
- Zhu, Y., Zhang, L., Zhou, X., Li, C., & Yang, D. (2021). The impact of social distancing during COVID-19: A conditional process model of negative emotions, alienation, affective disorders, and post-traumatic stress disorder. *Journal of affective disorders*, 281, 131-137. https://doi.org/10.1016/j.jad.2020.12.004



Istanbul Business Research, 53(2), 161-183

DOI: 10.26650/ibr.2024.53.1338316 http://ibr.istanbul.edu.tr/ http://dergipark.org.tr/ibr

Istanbul Business Research

Submitted: 05.08.2023 Revision Requested: 09.06.2024 Last Revision Received: 10.07.2024 Accepted: 19.07.2024

RESEARCH ARTICLE

Model Proposal for Evaluation of Technology Development Process: Conceptualisation and Scale Development*

Fethi Aslan¹ 💿

Abstract

Traditional approaches based on intuitive decisions of executivies, teams, or individuals often prove inadequate in managing complex technology development processes. Theoretical models proposed for successful technology development processes emphasize the necessity of standardized processes. However, empirical evidence regarding the applications and results of these models is limited. In this context, a model has been developed to examine how standardized technology development processes are implemented at the organizational level in companies. This model is defined by a scale that includes dimensions of technical research, business research, development, and performance. The proposed model is based on data obtained from companies with the highest R&D expenditure in Turkey. The findings provide evidence that a structured technology development process at the corporate level is necessary for success. It has been revealed that technical and business research phases are particularly determining factors in the development process. Additionally, it has been determined that the correct implementation of these processes has a positive impact on the performance of the process.

Keywords: Technology development, Standardised process, Exploratory factor analysis, Confirmatory factor analysis, Scale development

Introduction

As a result of globalisation, national businesses increasingly face competition from foreign businesses in distant regions of the world (Griner, Keegan, & Goldin, 2000, p. 19). Businesses are under pressure from the increasing complexity of customer demands and rapid changes in user experience, quality, performance and consumer attitudes. In addition, the intensity of competition and the pace of technological development are increasing businesses' challenges (Iansiti, 1995, p. 259).

New technologies can be applied to existing or new products, services and processes. These new technologies may change the conditions of competition and contribute to the



^{*} This article is derived from the dissertation titled 'A Maturity Model Proposal for Evaluating the Technology Development Process: An Application on the Activities and Approaches of R&D-500 Companies' prepared by Fethi ASLAN.

¹ Corresponding Author: Fethi Aslan (Dr.), Elazığ Special Provincial Administration, Elazığ, Turkiye.. E-mail: fethi.aslan@outlook.com ORCID: 0000-0002-5567-9706

To cite this article: Aslan, F. (2024). Model proposal for evaluation of technology development process: conceptualisation and scale development. *Istanbul Business Research*, 53(2), 161-183. http://doi.org/10.26650/ibr.2024.53.1338316

emergence of new markets (UNFCCC, 2017, s. 6). Businesses have been looking for new solutions for a long time to gain competitive edge and adapt to changing conditions. Developing appropriate technologies to meet current needs and future trends is recognised as a key solution. Related to this solution, it is also necessary to shorten the technology development time and increase the success rate (Griner et al., 2000, p. 29). In this way, businesses can maintain their competitive advantage and increase their growth potential.

The technology development process encompasses all decisions, actions, and capabilities involved in defining activities from idea exploration (opportunities) to transitioning to production. Its goal is to develop new items, components, systems, products, methods, or to enhance existing products and processes. This involves preparing the correct information at the appropriate level of detail and systematically implementing. As stated in the definition, technology development is a complex process involving many activities. It requires long-term effort and is characterised by high uncertainty. This is why careful planning and a flexible management approach are so important. Uncovering the unknown is the source of uncertainty. Uncertainty and complexity increase enterprises' risk levels in the technology development process. A high risk level leads to the failure of technology development activities (UNFCCC, 2017, p. 6). One of the main reasons for these risks is that technology development activities are carried out without a standardised process. Although technology development activities can be successfully completed without a standardised process, achieving successful results depends on individual skills (Richrath, Plano, & Nesbitt, 2016, p. 1). However, to increase the chances of success in technology development activities, organisations need more than just individuals' skills. Therefore, organisations are focusing more on business strategies and processes (Catlin, Scanlan, & Willmott, 2015, p. 1).

A clear definition of the research focus is crucial for addressing the research gap and enhancing the potential to generate new knowledge. Furthermore, clarifying the research scope contributes to a better understanding of the study and enhances its credibility. To this end, a comprehensive review of the literature was undertaken to examine studies that emphasise the importance of the field and to identify topics that have not previously been addressed. The study conducted by (Branscomb & Auerswald, 2002, p. 16) revealed that there is a substantial body of research on strategy, management, organisational motivation and financial issues related to technology development activities. Nevertheless, it was observed that research on the technology development process is scarce and that the number of documented studies is relatively limited. According to Carbonell, Rodriguez Escudero, & Munuera Aleman (2004, p. 83), studies on measuring the success of new products using the stage-gate evaluation approach are insufficient. According to Stern (2008, p. 132), the technology development process is long term and its results cannot be fully predicted in advance. A long period is required between the initiation of technology development activities and the achievement of clear results and impact. Therefore, there is a need for clear and precise information on how

the technology development process takes place. A study conducted by PwC revealed that organic growth of organisations is more than twice as high as other growth models (mergers, acquisitions, etc.). Agreements such as mergers, acquisitions and partnerships is frequently to procure assets pertaining to research and development capabilities, intellectual property rights and products. Consequently, the utilisation of internal R&D activities is considered a pivotal factor in the growth models adopted by firms (Wobig, 2015, p. 1). Furthermore Richardson (2017, p. 3), stated that the lack of setting common priorities for the implementation of activities at the organisational level may make comprehensive decision-making and implementation mechanism hinders agility. It becomes difficult to make the right decisions to stop low-performing technology development activities and focus on high-performing activities. Finally, according to Kolossovski (2019), in the product development process, 4 out of every 7 product ideas are found valuable, 1.5 of them are attempted and only 1 of them is successful. This situation is important in demonstrating difficulties in development and commercialisation phases of product development process.

Furthermore, studies on research and development, product and innovation management, which are either directly or indirectly related to technology development, have been analysed within a broad conceptual framework. Carbonell et al. (2004), examined the relationship between the various dimensions of fit/unfit criteria and new product success throughout the product development process in 957 firms in the mechanical equipment, computer equipment, electrical machinery, electronic equipment, measuring instruments and motor vehicles and other transport equipment industries in Spain. The study involved 77 firms and found that the relative impact of the five dimensions of fit/unfit criteria on new product success variedvaries depending on the stage of the development process and technological innovation. Song & Montoya-Weiss (2001), developed a theoretical model in their study to examine the regulatory effect of perceived technological uncertainty on the new product development process. Günday (2007), examined the influence of firm structure, firm characteristics, firm strategy, sectoral conditions, and relationship factors on innovativeness. In the second stage of the study, the effect of innovation on firm performance was analysed.

The study presents evidence based on survey data obtained from 169 firms. The evidence indicates that innovation determinants exert a significant positive effect on the firm's innovative ability. Furthermore, innovativeness is an effective factor in both production and innovation performance. Ernst, Hoyer, & Rübsaamen (2010), investigated the effects of cross-functional collaboration between sales, marketing and R&D on the new product development process and performance in a new product development project in the highest-income enterprises in Germany. The findings of this study indicate that collaboration between sales and R&D is crucial for new product success, particularly during the concept and product development stages. Tekin (2016), investigated the impact of innovation knowledge and in-

novation management on innovation capabilities and new product success in 10 firms within the Turkish white goods sector. The findings indicated that innovation capability significantly influences product success. Guimaraes, Paranjape, & Walton (2019), examined the impact of organisational culture, competitive intelligence, technology management, innovation project management, and absorptive capacity on new product development success. According to their findings, factors other than strategic leadership significantly influence the success of new product development. Meier & Kock (2022), examined the impact of agility in the organisation of research and development (R&D) units within an industrial firm operating globally in the field of mechanical engineering across more than 80 countries. In this context, they developed a six-dimensional scale, including agility culture, customer integration, autonomy, an iterative work method, cross-functional capabilities, and a horizontal hierarchy. The developed scale demonstrated a direct correlation between agility in R&D organisation and front-end success.

A comprehensive review of the existing literature provided valuable insights into gaps in the literature. This analysis has also helped us identify specific areas where our current research can make meaningful contributions. The reviewed existing studies provide a summary of the various factors and methodologies related to technology development, innovation product development processes in different industries and contexts. These studies deepen existing knowledge about technology development, innovation and product development processes, while also highlighting areas for further unexplored research. Previous studies have typically examined the technology development process as part of innovation and product development processes. However, as Kolossovski (2019) also highlighted, the probability of success decreases rapidly as the process progresses. The evaluation of innovation or product development process alone does not provide sufficient information about the technology development process. Therefore, it is necessary to place greater emphasis on the deepening of the technology and innovation literature, with a focus on the entire process from idea generation to the transition from technology to product/production. In previous studies, industries with different levels of technology (e.g., mechanical, computer equipment, electrical machinery, biotechnology, chemicals, and pharmaceutical industries) were examined separately. There is a need for evidence on the applicability of a comprehensive model that encompasses the practices related to technology development processes in industries with high, medium-high, medium-low, and low levels of technology. Finally, previous studies have generally focused on performance factors related to market and financial success. There is a need for evidence on process performance factors, including technical success, timing, and budgeting to determine whether the process has been successfully implemented.

In this context, a three-stage approach was used to determine the scope of the research. The first of these, as shown in Figure 1, is the internal R&D activities among the technology acquisition and development methods.



Figure 1. The Technology Acquisition Model

(Adapted from the studies of Chiesa, Frattini, Lazzarotti, & Manzini, 2009; Gallardo, 2013)

The second stage in determining the scope of the research is related to defining the evaluation process. Technology development projects/programmes are a process that involves inputs, activities, outputs, outcomes and impacts. The evaluation was conducted within the framework of a methodology that considered the factors shown in Figure 2. Therefore, the scope of the research was conducted within the framework of a methodology that considers the evaluation factors shown in Figure 2. The inputs provide information about the firm's characteristics. Activities refer to actions intended to obtain desired outputs by utilising resources. The immediate and tangible outcomes generated by activities must be measured for long-term results and impacts to achieve lasting benefits (Griffin, 1993, p. 115). In the evaluation approach, the relationship between inputs and outputs is expressed as efficiency the relationship between objectives and outcomes is expressed as effectiveness, and the relationship between needs and impact is expressed as benefit and sustainability. Different purposes can be associated with evaluation (Arnold, Åström, Glass, & Scalzi, 2018, p. 15). The purpose of this study was to determine the level of goal achievement depending on the implementation level of the activities.

The final stage in determining the scope of the research is related to the level at which the evaluation will be conducted. "*Organisational level focuses on organisational structures, processes, resources, and management issues*" (Bolger, 2000, p. 4). At the organisational level Figure 3, evaluation addresses the overall performance factors and capabilities of the organisation in fulfilling its tasks. These factors refer to internal regulations, processes, and

plans that determine how individuals' abilities should be utilised and directed (United Nations Development Programme, 2008, pp. 6–7). Assessing capacity at the organisational level add-resses issues at the organisational level. However, it also influences and is directly influenced by individual-level as well as sectoral and environmental interactions (Bolger, 2000, p. 4).



Figure 2. Theory of Evaluation (Change) (Arnold et al., 2018)



Time/Resources

Figure 3. Capacity Development: Conceptual Framework (Bolger, 2000)
Considering the scope of the study and the methodology employed, it is inevitable that there will be some limitations. The first is that this study focuses solely on internal technology development activities within the firm. Therefore, the proposed model does not provide evidence of its suitability for technology acquisition from external sources and technology development processes through collaboration. In addition, it presents information regarding outputs and outcomes achieved through inputs and activities. Finally, despite considering privacy principles related to the obtained data, research outcomes may not fully reflect the actual situation because of participant bias.

The aim of this study is to develop an evaluation model for measuring the standard process and performance of technology development activities, which will serve as a reference for conducting such activities. The model is characterised by a structure consisting of four dimensions: technical research, commercial research and performance. If the proposed model is successfully implemented, all personnel involved in the process will perform their activities based on predictable and evidence-based decisions. Once the proposed model is fully implemented, resources are efficiently used, ensuring minimal deviations in cost, time, and technical requirements. As a result, the success rate increased.

In line with this objective, the study relies on evidence from managers and experts working in research and development, technology development and innovation in the top 500 businesses with the highest R&D expenditures in Türkiye.

Method

This study was conducted in accordance with the positivist paradigm. The phenomena examined in this research were approached objectively in the context of causal relationships, in accordance with the conceptual framework. In this context, a quantitative methodology was employed in the study to examine phenomena objectively within the framework of causal relationships. The study has both exploratory and descriptive aspects. The data in this study were collected using the survey method.

Verbal informed consent was obtained from all participants before the study and the ethics committee approval of this study was obtained from Ethics Committee for Social and Human Sciences Research at Firat University. (Date: 14.11.2019; Number: 5).

In this study, the use of quantitative methods aims to:

1. Enable the generalisation of findings to the population by measuring the implementation level and frequency of a formal and structured technology development process in the selected sample. 2. Provide evidence to identify and justify fundamental relationships between variables.

3. Ensure obtaining objective results through statistical tests (Sun, 2012, p. 7; Walwyn & Chan, 2019, p. 14)

Factor and Item Acquisition Processes

Technology development processes consist of a series of activities that occur from the idea stage to the market launch stage (Pleschak, 1997, p. 11). Technology development processes involve defining each stage and organisational structure, standardising activities accordingly (Möller, Menninger, & Rober, 2011, p. 6). The fundamental idea of technology development process is to ensure that businesses allocate their limited resources to the right projects, thereby minimising technology development costs, development time and deviations in technical performance requirements. By following these processes, businesses develop solutions to the challenges they encounter and achieve success. Although each business may have unique processes and activities, these processes and activities often do not take place within a universally accepted framework (Pleschak, 1997, p. 11). Technology development processes are designed in different ways but with the same purpose in mind. The existence of various approaches creates uncertainty for implementers in selecting the appropriate approach (Ajamian & Koen, 2002). Technology can be aimed at meeting an existing need; however, it can also result from entirely new ideas that can create novel requirements (Stern, 2008, p. 11). The purpose of business processes is to ensure that tasks performed at the organisational level are performed in a systematic manner, in a sequence of consecutive steps, and following a specific methodology (Dumas, La Rosa, Mendling, & Reijers, 2018, p. 9).

In businesses with a well-organised structure, an integral part of a process and are executed in a suitable manner according to this process. It should be noted that this process is a comprehensive set of activities that enable businesses to transform inputs through various stages into outputs to achieve their objectives (Çetindamar, Phaal, & Probert, 2016, p. 2).

A standard process ensures clarity and consistency regarding where, how, and when each stage and activity constitutes a business process in a business will be executed. In organisations with structured processes, even if the same task is assumed by different individuals over time, it is expected that these tasks will be performed in the same manner (Robbins, & Barnwell, 2006, p. 111). Activities in a standard technology development process serve as guiding principles to support effective and efficient technology development efforts. This process focuses on best rather than mandatory rules (United States Department Of Energy, 2007, p. 4). The objective is to identify and synchronize what needs to be done at different stages of the process and by different working groups to align with common goals. However, a standard technology development process does not contain information about how it should be

implemented and by whom, as it lacks an implementation plan feature (Doerry, 2010, p. 22).

However, a standard technology development process does not contain information about how it should be implemented or by whom, as it lacks an implementation plan feature.

The scale development process was conducted in four stages. In the first stage, potential scale items and factors were determined by utilizing activities and stages obtained from normative technology development models. In the second stage, the scope validity of the scale was tested, and in the final stage, empirical evaluations were conducted.

Establishing a conceptual foundation is crucial to the scale development process. The conceptual foundation guides the researcher on what should be included in the developed model. Additionally, it defines the boundaries of the model, allowing the researcher to focus and reduce ambiguity in scope (Carpenter, 2018, p. 13). In the process of scale item generation, either inductive or deductive methods or a combination of both, can be applied. In the deductive method, items are created by scanning relevant literature and existing scales in the field. On the other hand, in the inductive method, items are generated based on information and qualitative data obtained through focus groups and individual interviews (Boateng, Neilands, Frongillo, Melgar-Quiñonez, & Young, 2018, p. 1).

In this study, the deductive method was used during the item development phase. For this purpose, 32 normative models widely accepted in the literature were analyzed. The selected normative models comprise both academically based models and models applied in the public and different industries. The activities and stages included in these normative models were classified by thematic analysis method. Normative models consist of different numbers of stages and activities depending on the level of detail. The identified common themes form a potential scale item pool. During the creation of the item pool, all normative models were examined, and a comprehensive approach that considered the activities agreed upon in the current studies was adopted. Additionally, to measure the performance of the process, relevant items (pertaining to performance) were obtained by referring to the literature.

Themaile Thailysis of Termany e Troub Tems			
Factors	Items	References	
Business Research	Market Potential	(Schulz, Clausing, Fricke, & Negele, 2000), (Loutfy & Belkhir, 2001), (Cooper, 2006), (Kausch, 2007), (Canez, Puig, Quintero, & Garfias, 2007), (Gaubinger & Rabl, 2014)	
Technical Research	Technology Development Capability	(Myers & Marquis, 1969), (Schulz et al., 2000), (Loutfy & Belkhir, 2001), (Ajamian & Koen, 2002), (Cooper, 2006), (Lind, 2006), (Whitney, 2007), (Crill & Siegler, 2017), (Kausch, 2007), (Gaubinger & Rabl, 2014),	

Thematic Analysis of Normative Model Items*

Table 1

Factors	Items	References
Development	Protection of Technology	(Canez et al., 2007), (Myers & Marquis, 1969), (Office of Technology Assessment (OTA), 1982), (United States Government Accountability Office (GAO), 2006), (Cooper, 2006), (Kausch, 2007), (Miller, Busta- mante, Roesch, Boshell, & Ayuso, 2015)
Performance	Compliance of Technical Success with Budget and Time Schedules (Earned Value Approach)	(Moser, 1985), (Chiesa et al., 2009), (Association for Project Manage- ment, 2013)

* Partial analysis results have been included

The number of factors was determined considering the number of stages in the normative models, which are partially presented in Table 2. The existing technology development models were created with 2 to 7 stages. In most models, a 4-stage structure was predominantly used.

Table 2

Stages	of Normative	Technology	Developm	ent Process*
Suges	01 Normanive	rechnology	Developin	eni i rocess

Stages	Number of Stages	References
 I-Idea Generation, 2-Project Planning, 3-Technology Concept, 4-Technology Development 	4	(Gaubinger & Rabl, 2014)
1-Invention, 2-Project Scope, 3-Technology Concept Develop- ment, 4-Technology Development, 5-Technology Optimization, 6-Technology Transfer	6	(Caetano, Araujo, Amaral, & Guerri- ni, 2011)
1-Explore, 2-Development, 3-Technology Transition	4	(United States Government Accoun- tability Office, 2006)
1-Basic Research, 2-Applied Research, 3-Development	3	(Bronzino, 1992)
1-Target İdentification, 2-Lead Discovery 3-Preclinical Trial 4- Clinical Trial 5-Approval	5	(Romasanta, Van der Sijde, & Van Muijlwijk-Koezen, 2020)

* Partial analysis results have been included

After identifying the stages and activities, scale items and factors were derived within the context of the model using an interpretive approach. To establish a clear, comprehensible, and consistent conceptual framework for the intended construct to be measured, it is necessary to define each factor of the scale. For this purpose, descriptions of technical and commercial research, development, and performance factors have been provided.

New technology development is a discovery process involving technical and business research activities that require a search for new knowledge. This situation arises from the presence of broad objectives and significant uncertainties at this stage (Sheasley, 1999, p. 52). In these stages, even simple changes in assumptions can lead to significant financial and technical risks. Allocating business resources to a technology initiative may result in neglecting other initiatives (Day, 2008, p. 52). A successful research phase can prevent interruptions and the emergence of conceptual and outcome changes within the process (Harmancioglu,

McNally, Calantone, & Durmusoglu, 2007, p. 422). The technical and business research factor, which examines technology, markets, and financial matters, represents the initial stages of the technology development process. It is designed to define the stages consisting of exploratory activities. These two factors consist of activities from generating technology ideas to determining the characteristics of technology, including conceptual development stage. Activities that involve in-depth examination and analysis related to the market, competition, and customers. The technical research factor consists of activities that analyse the feasibility of technology ideas, such as its suitability, feasibility, implementation possibilities, intellectual property status, and firm capabilities. Effective research activities assist in early termination of unpromising and unfeasible ventures.

The development phase defines activities aimed at practical advancement of technology. Activities during this stage are based on information obtained during the technical and business research stages. The technology requirements, working principles, and physical characteristics are defined. Additionally, the feasibility of the technology is demonstrated through laboratory and real-world application environments, and outputs that can be subject to intellectual property rights are generated. These activities also encompass processes related to the transition of the technology to the final product/production stage.

The final factor in the model is related to measuring the performance of the process. Performance refers to measuring how successful a task is being executed. The process includes decision-making, planning, problem-solving, monitoring interactions and evaluation activities. If the process is carried out successfully, the business shows good performance and becomes successful (Lusthaus, Adrien, Anderson, Carden, & Montalván, 2002, p. 76). The aim of this factor is to assess the business's success and performance during the technology development process. Objective measurements are necessary to ensure the success of the technology development process and the accuracy of decisions. Accurate analyses can be conducted using proper measurements. These accurate analyses by providing a foundation for sound decisions, facilitate the achievement of objectives (Milbergs & Vonortas, 2006, p. 2). At the organisational level, measuring performance can be conducted to encompass one or more criteria, such as effectiveness, efficiency, compliance, and financial feasibility (Lusthaus et al., 2002, p. 76). In this study, the identified performance factor was derived from criteria related to measuring operational outcomes. These criteria include technical performance (product's technical functionality and quality), development cost, and development time. The success of the process is evaluated based on meeting all these criteria at the highest level (Tatikonda & Rosenthal, 2000, pp. 76–77).

Research Population and Sample

The aim of this study is to examine the technology development process at the firm level. Therefore, implementing the sample in businesses where technology is developed will ensure line with the research purpose. For this purpose, the universe of research is constituted by the top 500 businesses in Türkiye that made the highest R&D expenditures in 2020. The R&D 500 firm list is based on the R&D data declared by the top 500 businesses in Türkiye in terms of export ranking, businesses disclosing R&D data on the Public Disclosure Platform, and businesses approved by the Ministry of Industry and Technology of the Republic of Türkiye as R&D centres. However, some businesses' names were not disclosed, yet they were included in the ranking, resulting in a total of 472 businesses comprising the research universe (Turkishtime, 2021, p. 26). The research employed the scale development methodology. Expert opinions were sought for content validity. The survey method was utilized for exploratory and confirmatory factor analyses. Survey data were collected through Google Forms during the period from November 2021 to May 2022, based on the responses of R&D, innovation, technology unit managers, and experts. During this period, data were collected from the same research population in two stages at different time points. The data collected in the first stage were intended for exploratory factor analysis, while the data from the second stage were gathered for confirmatory factor analysis. Analyzing issues such as missing data, outlier values, and multicollinearity was conducted. In total, 625 responses were used in the study. The distribution of responses was examined based on the classification established by the OECD in 2001 according to R&D intensities. Accordingly, 180 respondents (28.80%) were from the high-tech industry, 284 (45.4%) from the medium-high-tech industry, 94 (15.04%) from the medium-low-tech industry and 67 (10.72%) from the low-tech industry.

Findings

Based on the information obtained from Tables 1 and 2, a final structure consisting of 26 items and 4 factors was established. To assess the scope validity of this structure, support was sought from an evaluation group consisting of 8 experts. The structure was evaluated by an expert group comprising 2 experts in scale development, 4 experts working in the industry, and at least 2 academics who have received national or international project support. The number of questions in the initial survey form was excessive and included repetitive structures. Following the final evaluation, questions were removed from the survey that were deemed to have low clarity, terms that did not accurately define the expression, statements creating uncertainty that would prevent participants from understanding correctly, and questions that were deemed not to directly contribute to the main objectives of the survey. The Lawshe technique was applied to measure the validity of expert judgments. As a result of the expert panel evaluation, a content validity value of 0.75 was measured.

A good empirical study should have two main characteristics. These are the design of the researched model and the selection of relevant target groups approved by the participants. Kaiser-Meyer-Olkin (KMO) is a measure of sampling adequacy used in factor analysis. After the analysis, the KMO value was determined to be 0.909. Since this value is above 0.6, it can be stated that the data is sufficient for factor analysis, providing a solid basis for conducting

the analysis. The obtained value indicates that the suitability of the variables with the sample is excellent. The Bartlett's sphericity test was conducted, and the result was found to be significant with a p-value of 0.000 (p < 0.05), indicating that the assumption of sphericity has been met. This finding indicates that there is a significant correlation among the variables. Furthermore, the variables themselves show a certain level of correlation. They also exhibit high levels of correlation with other variables. Thus, the suitability of the obtained data for factor analysis was observed (Hair, Black, Babin, & Anderson, 2019, p. 136).

In the initial stage of scale development, a literature review and content validity assessment led to the decision to select a 4-factor structure. In determining the optimal number of factors, an approach that considered factor determination techniques along with a review of relevant literature was employed. In addition to the mentioned approaches, (Carpenter, 2018, p. 37) has suggested examining 2-factor and 6-factor structures when 3, 4, or 5 factor structures are obtained. This recommendation implies that, even if the initial 3, 4, or 5-factor structures seem reasonable, exploring alternative factor solutions can potentially identify a better factor structure. As a result of the statistical analysis, a 4-factor structure was obtained. The obtained structure was evaluated for 2 to 6 factors, and it was decided that the most suitable factor number is a 4-factor structure. Consequently, a structure consisting of 4 factors and 20 items was obtained.

Table 3		
Multicollinearity Analysis		
Factors	Tolerance Value	Variance Inflation Factor
Business Research	,463	2,159
Technical Research	,394	2,540
Development	,429	2,328

Dependent Variable: Performans

To determine whether the results obtained from exploratory factor analysis are samplespecific or generalizable to the population, confirmatory factor analysis was conducted in the second stage for cross-validation using the data obtained. During the confirmatory factor analysis stage, multiple collinearity analyses, validity-reliability tests, goodness of fit, and findings regarding the final structure of the model were obtained. When the VIF value is less than 5 and the tolerance value is greater than 0.2, multicollinearity problem does not occur (O'brien, 2007, p. 685). The analysis results for the presence of multicollinearity using VIF (variance inflation factor) and tolerance values are presented in Table 3. It was observed that the tolerance and VIF values are in accordance with the threshold values, which indicates that there is no multicollinearity problem.

To ensure discriminant validity, heterotrait-monotrait (HTMT) value should be below the strict threshold of 0.85 and the acceptable threshold of 0.90 or 1.00 (Gaskin, Godfrey, & Vance, 2018, p. 10). As seen in Table 4, values related to the model are below the acceptable threshold, indicating that discriminant validity has been achieved.

Heterotrait-Monotrait (HTMT) analysis							
Factors Business Research Technical Research Development Perform							
Business Research	1						
Technical Research	,887						
Development	,850	,903					
Performance	,702	,778	,798	1			

 Table 4

 Heterotrait-Monotrait (HTMT) analysis

To establish convergent validity, it is sufficient for the average variance extracted (AVE) value to be greater than 0.5. However, even if the AVE value is less than 0.5, convergent validity can still be achieved if the composite reliability (CR) value is greater than 0.6 (Safiih & Azreen, 2016, p. 43). Accordingly, within the scope of convergent validity, the AVE and CR values were calculated for each factor, as presented in Table 5. Comparing the calculated AVE and CR values with the threshold values, it is concluded that the structure shows convergent validity.

Table 5			
Validity and Reliability Resu	lts		
Factors	AVE	CR	Cronbach's Alpha
Business Research	,491	,852	,825
Technical Research	,651	,848	,812
Factors	AVE	CR	Cronbach's Alpha
Development	,517	,882	,841
Performance	,719	,911	,817

To determine the reliability level, both composite reliability and Cronbach's alpha values were examined. In structural equation modelling, composite reliability is preferred over Cronbach's alpha for assessing reliability. In this case, for reliability, the Cronbach's alpha value should be at least 0.7, and the composite reliability should be above 0.6 (Hair et al., 2019, p. 768). As shown in Table 5, both the composite reliability value and Cronbach's alpha value are above the specified threshold, indicating that reliability has been achieved for the entire structure. This suggests that the data used in the analysis are reliable and consistent and provide a strong foundation for the interpretation and conclusions of the study.

The model fit indices are used to assess the level of fit between the model and the data used (DiStefano & Hess, 2005, p. 227). The model fit indices are presented in Table 6. As seen in the table, the fit index values of the model fall within the range of reference values, indicating that the model fits the data appropriately.

Table 6		
Model Fit Indices		
Model Fit indices	Obtained Value	Recommended Value
Chi-square Test $(\chi^{2/sd})$	2,983	$1 < \chi^{2/sd} < 3$
Root Mean Square Error of Approximation (RMSEA)	0.07	RMSEA < 0.08

Model Fit indices	Obtained Value	Recommended Value
Standardised Root Mean Square Residual (SRMR)	0.026	0.00≤SRMR≤0.10
Goodness-of-Fit Index (GFI)	0.903	0.90≤GFI≤1.00
Adjusted Goodness-of-Fit (AGFI)	0.857	0.85≤AGFI≤1.00
Normed Fit Index (NFI)	0.901	0.90≤NFI≤1.00
Comparative Fit Index (CFI)	0.932	0.90≤CFI≤0.97

A model consisting of technical research, business research, development, and performance factors was obtained. The process components of the model are formed by technical research, business research, and development factors, and these factors play a determinant role in the performance factor, which is the outcome component. The structural representation of the model is shown in Figure 4.



Figure 4. Conceptual Model of Scale

The factor loadings obtained for each item through exploratory (EFA) and confirmatory factor analysis (CFA) are provided in Table 7.

Table 7 Item Loadings in EFA and CEFA

Factor	Items	EFA	CFA
	The current state of the market is evaluated.	,77	,82
ness earch	The market potential is evaluated.	,77	,83
Busi Rese	The competitive situation is evaluated.	,80	,77
ц	The level of openness to newness and improvement of the technology idea is evaluated.	,66	,64
arc	The suitability of the technology idea to scientific and technical approaches is evaluated.	,71	,68
al Rese	The restrictions imposed by existing patents on the development of the new technology are evaluated.	,66	,70
nica	The organization's technology development capabilities are evaluated.	,70	,73
èch	The application area and potential beneficiaries of the technology are evaluated.	,63	,72
Ĺ	The intellectual property protection level for the technology is evaluated.	,76	,73

Factor	Items	EFA	CFA
It	The technology is tested experimentally inin a laboratory environment for technical, performance, and functionality aspects.	,76	,68
	The technology is tested for technical performance and functionality in a real-world application environment.	,74	,73
lopmeı	Responsible teams are formed to transition from technology to product development/ production.	,68	,74
Deve	The maturity level of the technology is evaluated for its transition to the product stagep- roduction.	,65	,80
	Tests are conducted with potential beneficiaries (end-users), and feedback is collected.	,63	,65
	Evaluations regarding scaling up are carried out.	.64	.76
	Activities are carried out for the intellectual property protection of the technology.	.66	.66
	The level of completed work in technology development projects is aligned with the timeline.	,86	,84
ormance	The cost of completed work in technology development projects aligns with the planned budget.	,79	,87
Perf	In technology development projects, technical performance aligns with the prediction.	,70	,85
	In technology development projects, technical success aligns with the budget and sche- dule.	,85	,83

Results and Discussion

This study focuses on the characteristics of standard technology development processes. This section defines the standard technology development process, determine the level of its implementation, and assess its impact on process performance. The proposed model evaluates the implementation of necessary activities for standard technology development processes, their level of implementation, and their impact on process performance. This study is built upon the assumption that the top 500 businesses in Türkiye, which spend the most on R&D, have implemented a standard technology development process in some way and that this implemented process has made a positive contribution to process performance. Technology development is a long-term and complex process, consisting of numerous critical decisions and activities. However, previous research has predominantly focused on the ideation phase, organisational attitudes, behaviours, culture and leadership. An approach explaining the relationship between process implementation and performance, and how process performance should be measured, has not been developed. Alongside obtaining results regarding whether the standard technology development process works or not, the study has also provided answers to the questions of why and how. The impact of the technology development process on operational activities and their operational outcomes has been analyzed, and the standard technology development process has been conceptualised and supported using empirical results.

The development of new technology is a discovery process involving technical and commercial research activities that require new knowledge. A successful research phase can prevent interruptions and changes in concepts and outcomes during the process (Harmancioglu et al., 2007, p. 422). There are strong connections between marketing research and both engineering skills and R&D, which affect product outcomes (Sun, 2012, p. 7). Market, business opportunities, and competitive conditions play an important role in facilitating technology management (Sweeney, 1990, p. 9). Market-oriented behaviours facilitate strategy implementation and have a positive impact on performance (Dobni & Luffman, 2003, p. 583). Our findings, which are in line with this perspective, provide evidence supporting the necessity of incorporating factors associated with technical and business research into the technology development process.

The accuracy and reliability of early-stage (research) information shape subsequent technical stages (Frishammar, 2005, p. 16). The process is completed successfully when the required criteria are met at each stage. A process that is successfully completed shows good performance and achieves success (Lusthaus et al., 2002, p. 76). Decisions regarding product content and project scope have strategic importance due to their impacts on project performance. There is a relationship between decisions regarding the scope of the product to be developed, and project scope, and project performance. The transition between fundamental and applied research and product development processes is particularly important. Activities related to the selection, development, and transition of technology have a significant impact on product development performance (Iansiti, 1995, p. 260). The findings of this study support the information presented in previous studies. The evidence obtained from this study shows that the research stage impacts the development stage. Additionally, in the technology development process, the stages of technical research and development play a decisive role in performance. However, it is not possible to mention that business research has an impact on performance. Adequate implementation of activities related to the business and technical research stages leads to a positive impact on the development stage. This situation also contributes to the technology development initiative to achieve the desired level of success from an operational perspective.

From the perspective of the literature, specialisation in technology development processes is important due to the field's ongoing significance and potential to play an even greater role. In this regard, the study will also make a significant contribution to the expansion of the technology development process and its related literature. Moreover, it possesses comprehensive features by examining thirty-two normative models that directly reference technology development and technological innovation, which serve as the basis for the proposed model. Another important finding of this study is the evidence that technology development initiatives can be evaluated in the context of operational activities, and their performance can be measured.

From a practical perspective, the evidence obtained suggests that businesses should adopt a standard technology development process. The developed model will benefit technology development initiatives conducted by businesses at both national and international levels, across different industries, and of various sizes. Successful completion of the technology development process provides information about the challenges encountered throughout the process. The results obtained from the full implementation of the proposed model will help address the gaps related to the reasons for shortcomings in technology development processes by comparing them with previous practises. In this way, the proposed model can contribute to improving processes by learning from past applications.

In conclusion, structured processes are essential for businesses to shorten the technology development cycle, increase success rates, foster collaboration, and promote knowledge sharing while ensuring proper allocation of resources. By identifying and implementing these structured processes, more successful outcomes can be achieved. However, even if businesses have all the necessary capabilities, achieving perfect implementation of the process may not be possible.

Further Work

Challenges in technology development processes persist. The primary focus of process improvement is the decisions made and the extent to which employees can implement them. Researchers should always consider these factors. Each stage from idea generation to process performance measurement should be approached using different innovative approaches. Performance requirements, especially concerning technology development processes, should be examined using different factors and criteria. Even if a good process exists throughout the organisation, the impact of the relationship between team flexibility and empowerment required (Krishnan, 2013, p. 144) by team members, which is inherent in the nature of technology development, and the role of the leadership approach should be examined in the process.

Peer-review: Externally peer-reviewed. Conflict of Interest: The author has no conflict of interest to declare. Grant Support: The author declared that this study has received no financial support.

References

- Ajamian, G. M., & Koen, P. A. (2002). The PDMA toolbook for new product development (P. Belliveau, A. Griffin, & S. Somermeyer, Eds.). New York: John Wiley & Sons, Inc.
- Arnold, E., Åström, T., Glass, C., & Scalzi, M. D. (2018). How should we evaluate complex programmes for innovation and socio-technical transitions? Brighton: Technopolis. Retrieved from Technopolis website: https://www.technopolis-group.com/how-should-we-evaluate-complex-programmes-for-innovation-andsocio-technical-transitions/
- Association for Project Management (Ed.). (2013). Earned value management handbook. Princes Risborough, Buckinghamshire: Association for Project Management.

- Boateng, G. O., Neilands, T. B., Frongillo, E. A., Melgar-Quiñonez, H. R., & Young, S. L. (2018). Best Practices for Developing and Validating Scales for Health, Social, and Behavioral Research: A Primer. *Frontiers in Public Health*, 6, 149. https://doi.org/10.3389/fpubh.2018.00149
- Bolger, J. (2000). Capacity Development: Why, What And How. Canadian International Development Agency, Vol. 1(No. 1), 1–9.
- Branscomb, L. M., & Auerswald, P. E. (2002). An Analysis of Funding for Early-Stage Technology Development (No. NIST GCR 02–841; p. 153). National Institute of Standards and Technology. Retrieved from National Institute of Standards and Technology website: https://www.nist.gov/document/gcr02-841pdf.
- Bronzino J. D. (1992). *Management of medical technology: A primer for clinical engineers*. Boston: Butterworth-Heinemann. Retrieved from https://books.google.com.tr/books?id=NOJbAwAAQBAJ&printsec =frontcover&dq=MEDICAL+TECHNOLOGY:+ASSESSMENT+AND+ACQUISITION&hl=tr&sa=X &redir_esc=y#v=onepage&q=MEDICAL%20TECHNOLOGY%3A%20ASSESSMENT%20AND%20 ACQUISITION&f=false
- Caetano, M., Araujo, C. S., Amaral, D. C., & Guerrini, F. M. (2011). Open innovation and technology development: process: The gap on partnership adoption from a case study perspective. *Product Management & Development*, 9(2), 111–120. https://doi.org/10.4322/pmd.2012.003
- Canez, L., Puig, L., Quintero, R., & Garfias, M. (2007). Linking Technology Acquisition to a Gated NPD Process. *Research-Technology Management*, 50(4), 49–55. https://doi.org/10.1080/08956308.2007.1165 7451
- Carbonell, P., Rodriguez Escudero, A. I., & Munuera Aleman, J. L. (2004). Technology Newness and of Go/No-Go Criteria on New Product Success. *Marketing Letters*, 15(2/3), 81–97. https://doi.org/10.1023/ B:MARK.0000047386.33282.51
- Carpenter, S. (2018). Ten Steps in Scale Development and Reporting: A Guide for Researchers. Communication Methods and Measures, 12(1), 25–44. https://doi.org/10.1080/19312458.2017.1396583
- Catlin, T., Scanlan, J., & Willmott, P. (2015). Raising your Digital Quotient. McKinsey & Company, 13.
- Çetindamar, D., Phaal, R., & Probert, D. (2016). Technology management: Activities and tools (Second edition). New York: Palgrave Macmillan.
- Chiesa, V., Frattin, F., Lazzarotti, V., & Manzini, R. (2009). Performance measurement in R&D: Exploring the interplay between measurement objectives, dimensions of performance and contextual factors. *R&D Management*, 39(5), 487–519. https://doi.org/10.1111/j.1467-9310.2009.00554.x
- Cooper, R. G. (2006). Managing technology development projects. *IEEE Engineering Management Review*, 35(1), 67–67. https://doi.org/10.1109/EMR.2007.329141
- Crill, B. P., & Siegler, N. (2017). Exoplanet Exploration Program, Program Technology Development Plan. National Aeronautics and Space Administration. Retrieved from National Aeronautics and Space Administration website: https://exoplanets.nasa.gov/exep/technology/gap-lists
- Day, G. S. (2008). Is It Real? Can We Win? Is It Worth Doing? Harvard Business Review, 85(12), 110-146.
- DiStefano, C., & Hess, B. (2005). Using Confirmatory Factor Analysis For Construct Validation: An Empirical Review. *Journal of Psychoeducational Assessment*, 23(3), 225–241. https://doi. org/10.1177/073428290502300303
- Dobni, C. B., & Luffman, G. (2003). Determining the scope and impact of market orientation profiles on strategy implementation and performance. *Strategic Management Journal*, 24(6), 577–585. https://doi. org/10.1002/smj.322

- Doerry, N. (2010). Transitioning Technology to Naval Ships (No. Ser 05T/017). Washington: National Academy of Sciences. Retrieved from National Academy of Sciences website: https://apps.dtic.mil/sti/citations/ADA525441
- Dumas, M., La Rosa, M., Mendling, J., & Reijers, H. A. (2018). Fundamentals of Business Process Management (Second). Berlin, Heidelberg: Springer Berlin Heidelberg, 2017. https://doi.org/10.1007/978-3-662-56509-4.
- Ernst, H., Hoyer, W. D., & Rübsaamen, C. (2010). Sales, Marketing, and Research-and-Development Cooperation across New Product Development Stages: Implications for Success. *Journal of Marketing*, 74(5), 80–92. https://doi.org/10.1509/jmkg.74.5.080
- Frishammar, J. (2005). Towards a Theory of Managing Information in New Product Development (Dissertation, Luleå Tekniska Universitet). Luleå Tekniska Universitet. Retrieved from https://www.avhandlingar. se/avhandling/abfa7587e1/
- Gallardo, V. G. O. (2013). *Technology acquisition: Sourcing technology from industry partners* (Dissertation). University of Cambridge, London.
- Gaskin, J., Godfrey, S., & Vance, A. (2018). Successful System: It's Not Just Who You Are, But What You Do. AIS Transactions on Human-Computer Interaction, 10(2), 57–81. https://doi.org/10.17705/1thci.00104
- Gaubinger, K., & Rabl, M. (2014). Structuring the Front End of Innovation. In O. Gassmann & F. Schweitzer (Eds.). *Management of the Fuzzy Front End of Innovation* (pp. 15–30). Cham: Springer International Publishing. https://doi.org/10.1007/978-3-319-01056-4 2
- Griffin, A. (1993). Metrics for Measuring Product Development Cycle Time. Journal of Product Innovation Management, (10), 112–125.
- Griner, C., Keegan, B. W., & Goldin, D., S. (2000). Enhancing Mission Success A Framework for the Future. NASA Integrated Action Team.
- Guimaraes, T., Paranjape, K., & Walton, M. (2019). Model of Success Factors for NPD Performance. International Journal of Innovation and Technology Management, 16(07), 1950054. https://doi.org/10.1142/ S0219877019500548
- Günday, G. (2007). *Innovation models and implementations at firm level in manufacturing industry* (Yüksek Lisans Tezi). Sabancı University, İstanbul.
- Hair, J. F., Black, W. C., Babin, B. J., & Anderson, R. E. (2019). *Multivariate data analysis* (Eighth edition). Andover, Hampshire: Cengage Learning EMEA. Retrieved from www.cengage.co.uk
- Harmancioglu, N., McNally, R. C., Calantone, R. J., & Durmusoglu, S. S. (2007). Your new product development (NPD) is only as good as your process: An exploratory analysis of new NPD process design and implementation. *R&D Management*, 37(5), 399–424. https://doi.org/10.1111/j.1467-9310.2007.00486.x
- Iansiti, M. (1995). Technology development and integration: An empirical study of the interaction between applied science and product development. *IEEE Transactions on Engineering Management*, 42(3), 259–269. https://doi.org/10.1109/17.403744
- Kausch, C. (2007). A risk-benefit perspective on early customer integration. New York: Physica-Verlag Heidelberg.
- Kolossovski, E. (2019). What Makes a Great High-Tech Product? Retrieved September 7, 2022, from https:// uxdesign.cc/what-makes-a-great-high-tech-product-9f7197e5bda3#:~:text=A%20high%2Dtech%20 product%20is%20a%20subset%20of%20product%20that,often%20requires%20high%20R%26D%20 investments.

- Krishnan, V. (2013). Operations Management Opportunities in Technology Commercialization and Entrepreneurship. *Production and Operations Management*, 22(6). https://doi.org/10.1111/poms.12012
- Lind, J. (2006). Boeing's Global Enterprise Technology Process. Research-Technology Management, 49(5), 36–42. https://doi.org/10.1080/08956308.2006.11657396
- Loutfy, R., & Belkhir, L. (2001). Managing Innovation at Xerox. Research-Technology Management, 44(4), 15–24. https://doi.org/10.1080/08956308.2001.11671436
- Lusthaus, C., Adrien, M.-H., Anderson, G., Carden, F., & Montalván, G. P. (2002). Organizational assessment: A framework for improving performance. Washington, D.C.: Inter-American Development Bank. Retrieved from http://www.iadb.org
- Meier, A., Kock, A. (2022). Agile R&D Units' Organization Beyond Software—Developing and Validating a Multidimensional Scale in an Engineering Context. *IEEE Transactions on Engineering Management*, 69(6), 3476–3488. https://doi.org/10.1109/TEM.2021.3108343
- Milbergs, E., & Vonortas, N. (2006). *Innovation Metrics: Measurement to Insight*. Center for Accelerating Innovation and George Washington University. Retrieved from www.innovationecosystems.com
- Miller, M., Bustamante, C., Roesch, R., Boshell, F., & Ayuso, M. (2015). Renewable Energy Technology Innovation Policy (p. 60). The International Renewable Energy Agency (IRENA),. Retrieved from The International Renewable Energy Agency (IRENA), website: https://www.irena.org/-/media/Files/IRENA/ Agency/Publication/2015/IRENA_RETIP_2015.pdf
- Möller, K., Menninger, J., & Rober, D. (2011). Innovationscontrolling: Erfolgreiche Steuerung und Bewertung von Innovationen. Stuttgart: Schäffer-Poeschel. Retrieved from https://books.google.com.tr/books ?id=NSTRDwAAQBAJ&printsec=frontcover&dq=Innovationscontrolling:+Erfolgreiche+Steuerung+u nd+Bewertung+von+Innovationen&hl=tr&sa=X&redir_esc=y#v=onepage&q=Innovationscontrolling% 3A%20Erfolgreiche%20Steuerung%20und%20Bewertung%20von%20Innovationen&f=false
- Moser, M. R. (1985). Measuring Performance in R&D Settings. *Research Management*, 28(5), 31–33. https://doi.org/10.1080/00345334.1985.11756920
- Myers, S., & Marquis, D. G. (1969). Successful Industrial Innovations (No. NSF 69-17). Washington National Sicence Foundation. Retrieved from National Sicence Foundation website: https://play.google.com/ store/books/details?id=umKyAAAIAAJ&rdid=bookumKyAAAIAAJ&rdot=1
- O'brien, R. M. (2007). A caution regarding rules of thumb for variance inflation factors. *Quality & Quantity*, 41(5), 673–690. https://doi.org/10.1007/s11135-006-9018-6
- Office of Technology Assessment (OTA). (1982). *Technology and Handicapped People* (No. NTIS order #PB83-172056). U.S. Government Printing Office, Washington, Retrieved from https://www.princeton.edu > ~ota > disk3
- Pleschak, F. (1997). Development Problems of Small Technology-Based Firms and Ways of Overcoming Them. In K. Koschatzky (Ed.), *Technology-Based Firms in the Innovation Process*. Heidelberg: Physica-Verlag HD. https://doi.org/10.1007/978-3-642-52135-5
- Richardson, J. M. (2017). Naval Research And Development.pdf. Retrieved from https://apps.dtic.mil/sti/ pdfs/AD1091630.pdf
- Richrath, S., Plano, C., & Nesbitt, M. (2016). Transportation Performance Management Capability Maturity Model (No. OMB No. 0704-0188; p. 70). Spy Pond Partners.
- Robbins, S. P., & Barnwell, N. (2006). Organisational Structures and their Discourses. In *Dimensions of organisation structure*. In Organisation theory: Concepts and cases (5th ed.). Australia: Frenchs Forest,

NSW: Pearson Education.

- Romasanta, A. K. S., Van der Sijde, P., & Van Muijlwijk-Koezen, J. (2020). Innovation in pharmaceutical R&D: Mapping the research landscape. *Scientometrics*, 125(3), 1801–1832. https://doi.org/10.1007/ s11192-020-03707-y
- Safiih, M. L., & Azreen, N. M. A. (2016). Confirmatory Factor Analysis Approach: A Case Study of Mathematics Students' Achievement in TIMSS. 41–51. Malaysian Journal of Mathematical Sciences. Retrieved from http://einspem.upm.edu.my/journal
- Schulz, A. P., Clausing, D. P., Fricke, E., & Negele, H. (2000). Development and integration of winning technologies as key to competitive advantage. *Systems Engineering*, 3(4), 180–211. https://doi. org/10.1002/1520-6858(2000)3:4<180::AID-SYS2>3.0.CO;2-H
- Sheasley, W. D. (1999). Leading the Technology Development Process. *Research-Technology Management*, 42(3), 49–55. https://doi.org/10.1080/08956308.1999.11671284
- Song, M., & Montoya-Weiss, M. M. (2001). The effect of perceived technological uncertainty on Japanese new product development. Academy of Management Journal, 44(1), 61–80. https://doi.org/10.2307/3069337
- Stern, P. (2008). Vinnova's focus on impact: A Joint Approach for Logic Assessment, Monitoring, Evaluation and Impact Analysis. Vinnova - Swedish Governmental Agency for Innovation Systems. https://doi. org/10.22163/fteval.2008.146.
- Sun, S. (2012). Marketing and Design In Organisational New Product Development (Master Thesis). Victoria University of Wellington.
- Sweeney, M. (1990, September 10). *Determining a technology strategy for competitive advantage*. Presented at the 4th annual conference of the British Academy of Management, Glasgow, Scotland. Glasgow, Scotland.
- Tatikonda, M. V., & Rosenthal, S. R. (2000). Technology novelty, project complexity, and product development project execution success: A deeper look at task uncertainty in product innovation. *IEEE Transactions on Engineering Management*, 47(1), 74–87. https://doi.org/10.1109/17.820727
- Turkishtime. (2021). 2020 Yılı Ar-Ge harcamalarına göre en büyük 250 şirket. Retrieved from https://turkishtimedergi.com/
- United Nations Development Programme. (2008). *Capacity Development Practice Note*. UNDP Practice Note: Capacity Development. Retrieved from www.capacity.undp.org
- United Nations Framework Convention on Climate Change (UNFCCC. (2017). Enhancing financing for the research, development and demonstration of climate technologies.
- United States Department Of Energy. (2007). Stage-Gate Innovation Management Guidelines. U.S. Department Of Energy. Retrieved from U.S. Department Of Energy website: https://www1.eere.energy.gov/manufacturing/financial/pdfs/itp_stage_gate_overview.pdf
- United States Government Accountability Office (GAO). (2006). *Stronger Practices Needed to Improve DOD Technology Transition Processes* (No. GAO-06-883). Retrieved from https://www.gao.gov/assets/ a251503.html
- Walwyn, D., & Chan, A. (2019). Research guide for masters and doctoral students in the Graduate School of Technology Management. University Of Pretoria Graduate School Of Technology Management. Retrieved from https://www.up.ac.za/media/shared/26/2022/research-guide-for-post-grad-students-issue-26of-january-2019.zp166976.pdf
- Whitney, D. E. (2007). Assemble A Technology Development Toolkit. Research-Technology Management,

50(5), 52-58. https://doi.org/10.1080/08956308.2007.11657462

Wobig, E. (2015). *PwC Perspective: Unlocking product development value in M&A*. PricewaterhouseCooper. Retrieved from http://www.pwc.co.uk/innovation



Istanbul Business Research, 53(2), 185-200

DOI: 10.26650/ibr.2024.53.1232448 http://ibr.istanbul.edu.tr/ http://dergipark.org.tr/ibr

Istanbul Business Research

Submitted: 12.01.2023 Revision Requested: 29.09.2023 Last Revision Received: 17.08.2024 Accepted: 26.08.2024

RESEARCH ARTICLE

Dynamics of Digital Financial Inclusion in Türkiye

Behice Canatan¹ , Ayşe Akboz-Caner², Gökhan Özbilge³

Abstract

This study examined the dynamics of different levels of digital financial inclusion (DFI) in Türkiye within the framework of individuals' basic financial behaviours and socioeconomic conditions and investigated factors that may affect the utilisation rate of DFI. Accordingly, findings from the 2017–2021 World Bank Global Findex dataset were estimated using the ordered logit model. The concept of DFI, which is becoming increasingly important, underscores that financial systems should have two basic conditions. First, the ability of those who can access financial systems digitally should be as high as possible. Second, these actors with access to financial systems should also have sufficient utilisation rates of the relevant financial instruments. In this respect, by focusing on the determinants of this second condition, the study fills a gap in the literature and provides clues for policy recommendations. These findings provide empirical evidence for our motivation. In this respect, the results clearly show that socioeconomic factors, such as education level and income level, as well as financial behaviours, such as formal savings and borrowing, should be taken into account when expanding DFI.

Keywords: Digital financial inclusion, Financial behaviours, Socioeconomic factors, Ordered logit, Global findex

Introduction

Financial inclusion is a multidimensional concept that affects the accessibility and availability of financial products and services (Ahmad et al., 2021). This concept explicitly refers to a high proportion of both those who have access to the financial system and those who use the services offered by this financial system (WB, 2014; Demirgüç-Kunt et al., 2017; WB, 2019). Building on the same basic structure, digital financial inclusion (DFI) refers to financial systems in which a high proportion of individuals access and use financial services digitally (Lyman & Lauer, 2015). Thus, ever-evolving technologies can create opportunities for those excluded from the financial system and those with limited access (Niu et al., 2022) and help individuals make choices that promote financial stability (Ozili, 2018).



¹ Corresponding Author: Behice Canatan (Phd Candidate), Tarsus University, Department of Finance & Banking, Mersin, Turkiye. E-mail: behice_canatan@tarsus.edu.tr

² Ayşe Akboz-Caner (Phd Candidate), Tarsus University, Department of Finance & Banking, Mersin, Turkiye. E-mail: ayse_akboz@tarsus.edu.tr

³ Gökhan Özbilge (Res. Asst.), Istanbul Gelişim University, Department of International Trade And Finance, Istanbul, Turkiye. E-mail: gozbilge@gelişim.edu.tr

To cite this article: Canatan, B., Akboz Caner, A., & Ozbilge, G. (2024). Dynamics of digital financial inclusion in Turkiye. Istanbul Business Research, 53(2), 185-200. http://doi.org/10.26650/ibr.2024.53.1232448

Over time, innovations in digital technologies play an essential role in increasing the slow penetration of financial inclusion (Tiwari et al., 2019) and in developing and transforming digital financial markets (Heckel and Waldenberger, 2022). With such innovation, digitised financial inclusion can provide financial services to a wide range of households (Akanfe et al., 2020) and increase their likelihood of participating in the financial market (Liao & Zhou 2020; Shen et al., 2022; Zhou & He 2020).DFI is central to reducing economic and social negatives, especially during times of crisis, such as during the current COVID-19 pandemic. Increasing financial access in different segments of society can also contribute to financial repair (Tay et al., 2022).DFI can improve the width and depth of financial services, decrease service costs, and reach more populations (Chen et al., 2022). Apart from all of its benefits, the significant increase in loan supply and recent household debt with digital loan platforms, due to the widespread use of DFI, has drawn the attention of researchers (Feng et al., 2019; Chen et al., 2020; Yue et al., 2022). Studies in this direction have shown that DFI encourages more households to borrow through formal channels (Li et al., 2022; Yue et al., 2022).

Examining the definitions and characteristics of the inclusions mentioned above makes it easier to move towards the main idea of this study. This research also helps the existing literature in several ways. First, it provides a microeconomic perspective, focusing on the individual characteristics of financial actors that may affect their use of DFI. Because, as will be recalled, DFI is by definition a financial system with a high proportion of both those who can access and use digital financial services. As in this study, these actors can include individuals (Chen et al., 2022) and guide their financial choices. In this context, this study examines basic financial behaviours such as savings, debt, and socioeconomic factors, that affect individuals' DFI levels. The existing literature generally focuses on financial inclusion (Sanderson et al., 2018; Sarigul, 2020; Teker et al., 2021; Avcı, 2022; Girón et al., 2022), whereas studies on DFI (Naumenkova, 2019; et al., 2021; Li et al., 2022) are relatively new and limited. In other words, research on the factors that may affect the utilisation rate of DFIs has been ignored, and the focus has been on access.

Second, this study contributes to the existing literature in several ways. First, an index was created using variables (debit cards, credit card ownership, making bill payments online, buying something online, using mobile phones or the internet to access financial institution accounts, and using mobile phones or the internet to check account balances) included in the studies (Akanfe et al., 2020; PAL, 2020; Nandru et al., 2021; Ozili, 2022) as indicators of DFI, each of which also contains different information. Although various studies (Nandru et al., 2021) have analysed and correlated relevant variables, these indicators provide only partial and therefore incomplete information when used individually and separately. Therefore, considering the current deficiencies in the literature, an additional DFI index was created in this study. Using the DFI index created by principal component analysis (PCA) instead of using the variables separately strengthens the measurement capability.

Most importantly, despite the recent global spread of DFI, more research has yet to be conducted in Türkiye. However, Türkiye has the infrastructure to ensure DFI (PAL, 2020; Yıldırım, 2020; CBFO, 2021). Here, one of the important points is decisions that affect the participation rate of individuals in DFI. Moreover, including basic financial behaviours such as savings and debt, in the analysis is important for identifying which financial behaviours will affect the actors covered by DFI and how and in what way. On the other hand, given the increasing use of digital financial services in recent years, determining the impact of financial behaviours on DFI. This is because DFI not only increases household savings and the amount of loans taken from financial institutions, which may increase the likelihood that individuals fall into debt traps. Thus, this study analyzes the impact of DFI from a more detailed perspective. For this purpose, the ordered logit model was applied using data obtained from the World Bank's Global Findex database for 2017–2021.

The remainder of this article is organised as follows: Section 2 reviews previous studies on financial inclusion and DFI, Section 3 provides datasets and methodology, Section 4 presents empirical results and discussions, and Section 5 presents concluding remarks.

Literature Review

This section provides an overview of the literature on financial inclusion and DFI. Although the main factors determining financial inclusion have been the focus of many studies, DFI has not been adequately explained, especially in Türkiye. This study contributes to the DFI literature by addressing this gap and expanding the existing literature's scope.

When studies on financial inclusion in the current literature were examined, it was seen that they generally focused on the relationship between financial inclusion and macroeconomic indicators. Lenka and Sharma (2017) used autoregressive distributed lag (ARDL) to examine the impact of financial inclusion on economic growth in India. They emphasised the positive impact of financial inclusion on economic growth. Fernandes et al. (2021) investigated the impact of digital financial services on financial inclusion with ARDL in Mozambique. Gündüz & Özyıldırım (2020) revealed the role of participation banking on financial inclusion in Türkiye. Sarıgül (2021) investigated factors determining financial inclusion in Türkiye using a co-integration test. Chen et al. (2022) examined the relationship between financial inclusion, economic growth, income inequality, and poverty using panel data from China. They found that financial inclusion has a long-term effect on poverty reduction in China, whereas the effect of economic growth is relatively weak. Kouladoum et al. (2022) sought to uncover the relationship between digital technology and financial inclusion in Africa through generalised method of moments (GMM). They emphasised that the rate of financial inclusion in Africa has increased as digital technologies have increased. Demir et al. (2022) sought to reveal the relationship between quantile regression and financial technologies, financial inclusion, and income inequality in 140 countries. The determinants of financial inclusion and regional inequality in Türkiye were examined by Takmaz et al. (2022) using an ordered logit model. Canatan & İpek (2023) researched factors affecting household financial inclusion in Türkiye. These studies considered macroeconomic variables in various dimensions, such as economic growth, income inequality, poverty, digital technologies, and their financial use. In this context, it is understood that these studies emphasised that financial inclusion supported by digitalisation can positively contribute to income, income inequality, economic growth, and regional inequalities.

Although the literature on DFI is relatively new, international studies have been increasing rapidly, whereas national studies still need to be comprehensive. Ahmad et al. (2021) examined the relationship between FDI and economic growth in China using panel data analysis. Dai (2021) conducted a study on the development and supervision of robo-advisors of complex systems and DFI in China. Keliküm (2021) revealed the relationship between DFI, informal economy, and poverty through GMM in Africa. Kumar (2021) explored the role of DFI on access, use, and quality of digital finance in India. Liu et al. (2021) investigated the relationship between DFI and economic growth. Nandru et al. (2021) reported on the determinants of DFI in India. Chen et al. (2022) used panel data to investigate the impact of DFI on household poverty in China. Li et al. (2022) examined the impact of DFI on household debt in China using panel data. Li et al. (2022) aimed to determine the effect of household consumption on DFI using spatial econometrics in China. Luo et al. (2022) studied the effect of DFI on household consumption in China using a fixed-effect model. Ozturk and Ullah (2022) examined DFI, economic growth, and environmental sustainability in GMM and OBRI economies. Saxena and Thakur (2022) investigated the factors affecting DFI by ANOVA in India. These studies generally recommended expanding DFI participation to achieve higher economic growth and reduce poverty and income inequality.

As a result, there are still some limitations in the relevant literature. More than macroeconomic indicators should be considered for DFI. Therefore, this study identified and evaluated factors affecting DFI by creating a micro-based DFI index to fill Türkiye's research gap. As this study's main contribution emphasises, to say that DFI is fully present, the rate of people using digital financial systems must be high. Only in this way can the DFI rate be fully increased and the above-mentioned macroeconomic developments be achieved. Thus, these findings are expected to help DFI become more efficient and broad.

Data Set

The Global Findex dataset is one of the microdata sets that measures individuals' savings, debt, payment, remittance, and risk management (+15) in the most detailed way (WB, 2018).

In this study, the 2017–2021 Global Findex Data Set (2011–2014 were not included because they did not have variables presenting the DFI index) was used as a pool to determine the impact of socioeconomic conditions and financial behaviours that affect DFI in Türkiye.

In this regard, a composite DFI indicator was created for Türkiye, which included DFI indicators via PCA. The formation of PCA follows the strategy of the World Bank (2022b), data set availability, and the relevant literature (Akanfe et al., 2020; Nandru et al., 2021; Ozili, 2022) and (PAL, 2020). For the DFI index, a triple-categorical dependent variable was formed that included items like debit cards, credit card ownership, making bill payments online via the internet, buying something online using the internet, using mobile phones or the internet to access financial institution accounts, and using mobile phones or the internet to check account balances. In addition, this indicator considered the need to use system contents and access financial systems. Furthermore, debt and savings variables were used to measure the effects of using the financial system for the same purpose. The PCA methodology and DFI construction are described in detail in the next section.

In addition, the choice of the independent variables debt and savings is based on their potential impact on the utilisation rate of DFI. An important point is to use borrowers that are family or friends rather than informal borrowers. One reason for this is that individuals in the sample may be reluctant to report informal borrowing (as Lyons et al. (2022) noted, factors such as legal and regulatory institutions and cultural factors, which are particularly emphasised in this study, are important in Türkiye and similar societies). Another reason is that the effects of borrowing from family or friend networks are well known in Türkiye (Yükseker, 2010), especially for sociological aspects). Finally, choosing socioeconomic variables is important in guiding their financial behaviour and understanding how they can guide their utilisation rates while following relevant literature. The variables included in the dataset and their summary statistics are detailed in Table 1.

Table 1	
Summary Statistics	
Variable	Description and Summary Statistics
DEI	Digital Financial Inclusion Index, N:1294, Min:1-Max:3
DFI	1:Low: 435(%33.62); 2:Med: 465(%35.94); 3:High: 394(%30.45);
	Income group, N:1294, Min:1-Max:5
Income	1:Poorest: 146(%11.28), 2:Second: 155(%15.69), 3:Middle: 221(%17.08),
	4:Fourth: 297(%22.95), 5:Richest: 475(%36.71)
	Education level, N:1294, Min:1-Max:3
Education	1:Completed primary or less: 188(%14.53), 2:Secondary: 766(%59.20),
	3:Completed tertiary or more: 340(%26.28)
Gandar	Gender, N:1294, Min:1-Max:2
Gender	1:Male: 785(%60.66), 2:Female: 509(%39.34),
Wartsform	Labour force inclusion status, N:1294, Min:1-Max:2
worktorce	1:In-workforce: 967(%74.73); 2:Out-of-workforce: 327(%25.27);

Variable	Description and Summary Statistics				
Age	Age, N:1294, Min:17-Max:87				
	Mean: 37.84, standard deviation: 13.86				
Saved FI	Saved in past 12 months: Using an account at a financial institution, N:1294,				
	Min:1-Max:2				
	1:Yes: 345(%26.66), 2:No: 949(%73.34)				
Saved IS	Saved in past 12 months: Using an informal savings club, N:1294, Min:1-Max:2				
	1:Yes: 106(%8.19), 2:No: 1188(%91.81)				
Borrowed FI	Borrowed in past 12 months: From a financial institution, N:1294, Min:1-Max:2				
	1:Yes: 347(%26.82), 2:No: 947(%73.18)				
Borrowed FF	Borrowed in the past 12 months: From family or friends, N:1294, Min:1-Max:2				
	1:Yes: 465(%35.94), 2:No: 829(%64.06)				

Finally, remarkable descriptive statistics regarding the variables in the dataset can be observed. In this regard, only 30.45% of the respondents in the DFI index are at a high level. In comparison, the percentage of those at low and medium levels is 69.56%. These rates are preliminary evidence for our research question and emphasise that for a sample like Türkiye, where digital financial infrastructure is available, the usage rate of tools offered by the relevant infrastructure should be high. In addition, the fact that the rate of those who borrowed from a family/friend network was 35.94%, whereas the rate of those who borrowed from a financial institution was 26.82% strengthens the argument that was made before for the variable under consideration. In conclusion, inadequate saving rates and other essential points highlighted in this paragraph emphasise that DFI must consider individual factors.

Methodology

The ordered logit model is widely used when the dependent variable has more than two ordered responses. This approach provides a methodology for finding independent variables that explain ordinal variables, considering the measurement uncertainty of the data (Ballerin et al., 2016). The ordered logit model can be written as in Eq.(1) and represents the dependent variable reflecting the DFI categories (DFI1, DFI2, and DFI3):

$y_i^* =$	$\beta' x_i + \epsilon_i$	i = 1,, n	(1)
$y_i =$	0 if - $\infty < y_i^* \le \mu_0$	(low level of DFI: DFI1)	
=	$1 \text{ if } \mu_0 \!\! < \!\! y_i^* \!\! \leq \!\! \mu_1$	(medium DFI: DFI2)	(2)
=	2 if $\mu_1 < y_i^* \le \infty$	(high DFI: DFI3)	

Here, μ is the threshold points and indicates the bounds of the dependent variable, *xi* represents the explanatory variable matrix, εi is the error term, and β is the vector of the parameter to be estimated. In the *xi* explanatory variables matrix in Eq.(1) and adapted to the

study, there are gender, age, educational status, employment status, income status, and saving and borrowing behaviors of the individual. In the next section, in Table 5, where the ordered logit model estimation results are shown, the odds ratio values are estimated according to the reference group and express the effect of the variables on DFI proportionally. In the logit models, the interpretation is made based on the odds ratio In logit models, interpretation is based on odds ratios and marginal effects (İpek, 2020).

Empirical Results

PCA is a data analysis tool that preserves the information of many interrelated variables and is used to reduce the number of variables. The most significant components (principal components) are determined using the eigenvalues and eigenvectors of the covariance matrix of the data matrix. The index created by PCA is based on projections of the principal components of the data (Abeyasekera, 2003; Vyas and Kumaranayake, 2006). In this study, a DFI index was constructed using a weighted combination. Where $w_i, ..., w_n$ are the weights of the components on the index. X_n is the projection of the nth principal component:

$$Index = w_i X_i + \dots + w_n X_n \tag{3}$$

$$DFI=0.45*(PC1)+0.40*(PC2)+0.27*(PC3)+0.37*(PC4)+0.44*(PC5)+0.46*(PC6)$$
(4)

f = f = f = f = f = f = f = f = f = f =						
	PC1	PC2	PC3	PC4	PC5	PC6
Eigenvalue	0.3012	0.5514	0.8921	3.1678	0.6445	0.4428
Proportion	0.0502	0.0919	0.1487	0.5280	0.1074	0.0738
Cumulative	1.0000	0.8760	0.6767	0.5280	0.7841	0.9498
Eigenvectors						
Component 1	0.4550	0.4024	0.2738	0.3761	0.4421	0.4681
Component 2	-2.2381	-0.2194	0.8478	0.3314	-0.1211	-0.2277
Component 3	-0.2309	0.5124	0.3364	-0.6796	0.2960	-0.1464
Component 4	-0.4130	0.6196	-0.2563	0.5177	-0.0758	-0.3257
Component 5	0.2756	0.3757	0.1653	-0.1369	-0.8327	0.2097
Component 6	0.6606	0.0537	0.0053	0.0036	0.0553	-0.7467

Table 2The results of Principal Component Analysis

Eigenvalues are a metric that measures how much each principal component explains the variance in the data set in PCA. Large eigenvalues indicate that the component is significant and reflects the underlying structure in the data set (Demšar et al., 2013). PC4 (first principal component) has a variance of 3.16 and explains 52% of the total variance (3.16/6). PC3 (second principal component) has a variance of 0.89, i.e. 14% of the total variance (0.89/6). As a result, we can also say that the first two principal components explain the sum of the variances of the individual components or about 66% of the total variance.

Table 3

The suitability of PCA was measured using the Keiser–Meyer–Olkin (KMO) test and Bartlett's test of sphericity. To test the sample adequacy, the results of the KMO test and Bartlett's test of sphericity were used and are given in Table 3. If the KMO index is 0.80, it is considered meritorious, 0.70 middling, and below 0.50 is unacceptable (Kaiser & Rice, 1974; Wu, 2021). The results supported the use of PCA in this study.

KMO and Bartlett's Test Results					
Variable]	кмо		
Overall			0.83		
PC1: Check account			0.80		
PC2: Buying something online		0.88			
PC3: Credit card		0.84			
PC4: Debit card		0.88			
PC5: Making bill payments online			0.86		
PC6: Account			0.79		
Dantlatt's	Chi-square	df	p-value		
Dai ucu s	2593.308	15	0.000		

The findings of the ordered logit model used to study the DFI effect are included in this section. The parallel assumption, which is the most critical assumption in the ordered logit model, was tested with the help of Wolfe Gould and Brant's (1990) tests. The hypothesis that the coefficients of independent variables are the same across all dependent variable categories must be tested. More technically, this means that the odds ratios of a category in all logistic models remain the same against a variable created by collapsing all categories except itself (Williams, 2016). Thus, a more generous interpretation technique can be used. Therefore, parallel assumption tests were applied.

Table 4Parallel Regression Tests

Test	Chi2	df	P> Chi2
WolfeGould	11.88	13	0.537
Brant	12.48	13	0.489
Score	12.03	13	0.525
Likelihood Ratio	12.24	13	0.508
Wald	12.12	13	0.518
Information Criteria	ologit	gologit	difference
AIC	2477.93	2491.69	-13.76
BIC	2555.41	2636.32	-80.91

Table 4 presents the results of the Wolfe–Gould and Brant tests, in which the parallelism assumption was tested for the model in general. It was concluded that the probability value of the Ho hypothesis was greater than 0.05; therefore, the assumption of parallelism was provided for the overall model. When the model goodness of fit, Akaike Information Criterion (AIC), and Bayesian Information Criterion (BIC) values were compared, it was observed that the ordered logit model (ologit) had lower information criterion values. Lower AIC and BIC information criteria were selected.

Ta	bl	e	5

Ordered Logit Model Estimation Results

	0.1.4s	Marginal Effect			
Variable	Daus	Low Med		High	
	Kallo	DFI	DFI	DFI	
Fomalo	1.017	003	.000	.002	
remaie	(.132)	(.023)	(.003)	(.019)	
In Workforce	1.749***	101***	.016***	.084***	
In workjorce	(.253)	(.028)	(.008)	(.020)	
100	.967***	.005***	000***	004***	
Age	(.004)	(.000)	(.000)	(.000)	
Constant dama a dama a dama a dama a dama a dama a dama a dama a dama a dama a dama a dama a dama a dama a dama	1.521*	084*	.025*	.058**	
Secondary education	(.251)	(.033)	(.012)	(.022)	
	4.311***	268***	.022*	.245***	
Completed tertiary or more	(.892)	(.037)	(.013)	(.035)	
S 10/20	1.352	058	.018	.039	
Second %20	(.300)	(.042)	(.014)	(.029)	
	1.740**	106**	.029*	.076**	
Middle %20	(.370)	(.040)	(.012)	(.028)	
E (1.0/20	2.462***	169***	.037**	.132***	
Fourth %20	(.516)	(.039)	(.012)	(.029)	
D: 1 (0/20	2.231***	152***	.036**	.115***	
Richest %20	(.458)	(.039)	(.012)	(.028)	
Saved using an account at a	2.522***	167***	.012*	.154***	
financial institution	(.369)	(.025)	(.005)	(.025)	
Saved using an informal saving.	s .913	016	.002	.013	
club	(.213)	(.042)	(.008)	(.034)	
Borrowed from a financial	2.914***	192***	.013*	.179***	
institution	(.417)	(.024)	(.005)	(.025)	
Borrowed from a financial	.931	012	.002	.010	
institution	(.121)	(.023)	(.004)	(.010)	
	.101*				
Cutl	(.403)				
	1.964*				
Cut2	(.407)				
Model Diagnostics		The Goodness of Fit Tests			
Number of $obs = 1,294$		McFadden=		0.161	
LR chi2(13) =	388.99	McFadden(adjusted) =		0.148	
Prob > chi2=	0.0000	AIC=		2060.489	
Pseudo R2=	0.1608	BIC=		2137.971	
Log likelihood=	-1015.2444	Hosmer-Lemeshow (25)=		49.408 (0.377)	
0		. (-/		1	

*p < 0.05, **p < 0.01, ***p < 0.001 and values in parentheses represent standard errors.

Because there was no violation of the parallelism assumption, the ordered logit model was suitable for this study. Additionally, when the fit tests presented in Table 5 were examined, we found a fit between the dataset and the model. The cut-off point, which expresses the μ threshold values in Eq.(2), is the cut-off point for the latent variable used to distinguish the low-level DFI category from the medium and high categories when the independent variables are 0. Table 5 indicates that the lower and upper threshold values for each statistically signi-

ficant category do not equal one. From this viewpoint, it can be concluded that no categories need to be combined in the model, and the categories established are consistent (lpek, 2022).

According to the statistically significant odds ratio results, individuals in the workforce are 1.74% more likely to participate in DFI than those not in the workforce. Increased workforce participation will create opportunities for DFI. With membership in the workforce, account ownership and access to financial transactions will increase, while expanding DFI will contribute. When the age variable among the continuous variables was analysed, it was seen that the likelihood of participating in the DFI decreased as individuals got older. According to the estimation results, education level is the most influential factor on DFI. More educated individuals are more likely to engage in DFI. This finding highlights the importance of education for providing access to digital financial services and products, in other words, for spreading DFI. Digital financial transactions are structured in a way that literate and educated adults can easily adapt, but illiterate people are excluded (Matthews, 2019). An increase in income level increased the probability of participating in DFI. This finding revealed the importance of providing financial services that meet the wishes and needs of low-income group members (Arnold & Gammage, 2019). It can also provide new tools for low-income households to access digital opportunities and help them better manage their financial behaviours while reducing transaction costs. In addition, opportunities can be offered to increase productivity and efficiency to increase income.

Although not statistically significant, the negative coefficient of the odds ratio indicated that, as expected, informal savers are less likely to participate in DFI. Similarly, although statistically insignificant, women are more likely to participate in DFI than men. It is important to consider these implications. In particular, for the Turkish sample, integrating informal savings into the financial system may be easier when women actively participate in the financial system. This is because women in Türkiye are more likely than men to engage in informal savings, i.e. having more savings under their pillow (Bank ING, 2021; Özbilge, 2022). Borrowers from family and friend networks were 0.93% less likely to engage in DFI. Individuals who borrow from nonfinancial institutions are likely less likely to engage in DFI. Due to the social nature of kinship ties, informal debts may be obtained mostly from a network of family and friends (Li et al., 2022). Another finding is that formal savings are 2.52% times more likely to be involved in DFI. Individuals who borrow from a financial institution are 2.91% more likely to engage in DFI. These findings demonstrate the significant impact of financial behaviours on DFI. This study also highlights the importance of individuals' strategies for digitally making their savings and borrowing behaviours (Niankara, 2023).Note that the household debt-income balance must be maintained while expanding DFI to prevent excessive debt growth (Luo & Li, 2022).

When the other variables were constant and the *marginal effects* showing the effect of the factor affecting DFI according to each DFI level compared to the average were examined, men, individuals with higher education and income levels, those who participate in the work-

force, formal savings, and debt holders were more likely to participate in DFI. Considering the rapid increase in digital financial transactions with technology (Jain et al., 2021) and the interactive relationship between finance and digitalisation (Bunje et al., 2022), it is vital to increase education, workforce participation, and income among the masses to ensure DFI and reach underserved segments. Additionally, the studys results highlight the current level of use of DFI and two marginalised groups: family/friend borrowers and informal savers. In particular, the most significant effects on high levels of DFI are from savings from financial institutions and debt variables with positive coefficients of 0.154 and 0.179, respectively. This provides concrete evidence that DFI may cause imbalances in household debt management.

On the other hand, it provides clues that financial institutions can make saving options more attractive and can also help individuals direct their savings. As a result, it allows us to observe the effects of financial behaviour on DFI. Thus, a framework that can provide a prediction for policy recommendations that can be made on the utilisation rate of DFI can be obtained.

Conclusion and Discussion

This study discusses the factors affecting DFI in Türkiye within the framework of financial behaviours and socioeconomic factors that may affect the usage rate. For this purpose, the effect of these determinants was analysed using the ordered logit method on the 2017–2021 Global Findex Dataset. The results obtained provide a perspective, especially for developing countries such as Türkiye, where digital infrastructure exists in terms of financial systems. However, the rate of DFI use needs to be increased. This perspective contributes to the encouragement of individuals to participate in DFI, especially when the possible effects of financial behaviours are known. The results reveal the importance of decisions made, especially regarding debt and savings, when coordinating the development of DFI.

According to the findings, after education level, the utilisation of financial institutions for saving and borrowing are the most important determinants of DFI. However, given the high borrowing costs, DFI may encourage households to overloan while enabling more convenient and efficient access to financial resources. In this context, policies at the household level should make careful efforts to improve and promote DFI and increase its use. Specific improvements should be made by controlling household debt levels.

Because of the analysis conducted in the study, it was found that 34% of the participants had a low DFI, 36% had a medium DFI, and 30% had a high DFI. Notwithstanding its benefits, DFI for Türkiye shows that it is not sufficiently accessible to vast segments of the population, which points to a gap between availability, accessibility, and use. In particular, informal savings and traditional options, such as borrowing from family/friend networks. Thus, DFI rates will significantly improve as households move towards formal channels. Moreover, and most importantly, it is anticipated that growing the supply of financial products and services alone will only help expand DFI if holistic efforts are made to improve it. The findings that increased workforce participation, education, and income levels. Furthermore, increase DFI necessitate a holistic perspective. Individuals' compliance with financial systems depends primarily on their ability to meet favourable socioeconomic conditions.

Although this study provides important implications for optimising the impact of DFI in Türkiye and determining the factors on which its use depends, it has some limitations. The effect of heterogeneity across units cannot be explained due to the horizontal cross-sectional structure of the dataset. Only situations specific to COVID-19 can be identified and compared with the findings of this study. Based on these limitations, future research could examine more complex financial behaviours and determine whether other aspects, such as digital financial literacy, could benefit, especially in developing DFI. Additionally, where access to datasets showing cultural and behavioural differences is available, other developing countries can be included in the analysis, and the relevant field can be expanded to compare theory and empirical results.

References

- Abeyasekera, S. (2003). Household surveys in developing and transition countries: design, implementation and analysis. *United Nations Statistics Division*.
- Ahmad, M., Majeed, A., Khan, M. A., Sohaib, M., Shehzad, K. (2021). Digital financial inclusion and economic growth: Provincial data analysis of China. *China Economic Journal*, 14(3), 291-310.
- Arnold, J.,& Gammage, S. (2019). Gender and financial inclusion: the critical role for holistic programming. Development in Practice, 29(8), 965-973.
- Akanfe, O., Valecha, R., & Rao, H. R. (2020). Design of an inclusive financial privacy index (INF-PIE): A financial privacy and digital financial inclusion perspective. ACM Transactions on Management Information Systems (TMIS), 12(1), 1-21.
- Avcı, B. S. (2022). Financial Inclusion and Economic Growth: An Analysis for Turkey. Karamanoğlu Mehmetbey Üniversitesi Sosyal Ve Ekonomik Araştırmalar Dergisi, 24(42), 241-254.
- Ballerini, L., Lovreglio, R., Hernández, M. D. C. V., Gonzalez-Castro, V., Maniega, S. M., Pellegrini, E., & Wardlaw, J. M. (2016). Application of the ordered logit model to optimising Frangi filter parameters for segmentation of perivascular spaces. *Procedia Computer Science*, 90, 61-67.
- Bank, I. N. G. (2022). Turkey's savings trends survey 2022/3. Quarter report. Retrieved from https://www. tasarrufegilimleri.com/Docs/2022 3 Ceyrek.pdfAccessed November 14, 2022

Peer-review: Externally peer-reviewed.

Conflict of Interest: The authors have no conflict of interest to declare.

Grant Support: The authors declared that this study has received no financial support.

Author Contributions: Conception/Design of study: B.C.; Data Acquisition: G.Ö., A.A.C.; Data Analysis/Interpretation: B.C.; Drafting Manuscript: B.C., A.A.C.; Critical Revision of Manuscript: B.C., G.Ö., Final Approval and Accountability: B.C., G.Ö., A.A.C.

- Bunje, M. Y., Abendin, S., & Wang, Y. (2022). The multidimensional effect of financial development on trade in Africa: The role of the digital economy. *Telecommunications Policy*, 46(10), 102444.
- Canatan, B., & İpek, Ö. (2023).Factors affecting household financial participation. Academic Review of Economics and Administrative Sciences, Nigde Omer Halisdemir University, 16(1), 265–276.
- CBFO (Presidency of The Republic of Türkiye Finance Office), (2021). The state of finctech ecosystem in Türkiye. Retrieved from https://www.cbfo.gov.tr/sites/default/files/2021-12/the state-of-fintech-ecosystem-in-turkiye-2021.pdfAccessed November 23, 2022
- Chen, R., Chen, H., Jin, C., Wei, B., & Yu, L. (2020). Linkages and spillovers between internet finance and traditional finance: evidence from China. *Emerging Markets Finance and Trade*, 56(6), 1196-1210.
- Chen, Y., Yang, S., & Li, Q. (2022). How does the development of digital financial inclusion affect the total factor productivity of listed companies? Evidence from China. *Finance Research Letters*, 102956.
- Chen, Z., Zhu, H., Zhao, W., Cao, B., & Cai, Y. (2022). Dynamic nonlinear connectedness between the financial inclusion, economic growth, and China's poverty alleviation: Evidence from a Panel VAR Analysis. *Complexity*, 2022.
- Chen, S., Liang, M., & Yang, W. (2022). Does digital financial inclusion reduce China's rural household vulnerability to poverty: An empirical analysis from the perspective of household entrepreneurship. *SAGE Open*, 12(2).
- Demirgüç-Kunt, A.,& Singer, D. (2017). Financial inclusion and inclusive growth: A review of recent empirical evidence. World Bank Policy Research Working Paper, (8040).
- Demir, A., Pesqué-Cela, V., Altunbas, Y., & Murinde, V. (2022). Fintech, financialinclusionandincome inequality: A quantileregressionapproach. *The European Journal of Finance*, 28(1), 86-107.
- Demšar, U., Harris, P., Brunsdon, C., Fotheringham, A. S., & McLoone, S. (2013). Principal component analysis on spatial data: An overview. *Annals of the Association of American Geographers*, 103(1), 106-128.
- Dai, W. (2021). Development and supervision of robo-advisors under digital financial inclusion in complex systems. *Complexity*, 2021.
- Fernandes, C., Borges, M. R., & Caiado, J. (2021). The contribution of digital financial services to financial inclusion in Mozambique: an ARDL model approach. *Applied Economics*, 53(3), 400-409.
- Gündüz, M., & Özyıldırım, Y. (2020). The role of islamic banking in the development of financial inclusion: Calculation of Islamic Banking Financial Inclusion Index. *BDDK Bankacılık ve Finansal Piyasalar Dergisi*, 14(2), 119-144.
- Girón, A., Kazemikhasragh, A., Cicchiello, A. F., & Panetti, E. (2022). Financial inclusion measurement in the least developed countries in Asia and Africa. *Journal of the Knowledge Economy*, 13(2), 1198-1211.
- Heckel, M., & Waldenberger, F. (2022). Financial Systems in the Digital Age: Perspectives from Europe and Japan. In The Future of Financial Systems in the Digital Age (pp. 1-18). Springer, Singapore.
- İpek, E. (2020). The costs of disability in Turkey. Journal of Family and Economic Issues, 41(2), 229-237.
- İpek, Ö. (2022). The determinants of subjective well-being of older adults in Turkey. *Çankırı Karatekin Üniversitesi İktisadi ve İdari Bilimler Fakültesi Dergisi*, 12(1), 84-103.
- Jain, P., Upadhyay, D., & Purswani, G. (2021). Digital financial inclusion: Strategic issues and imperatives. In Financial Inclusion in Emerging Markets (pp. 297-309). Palgrave Macmillan, Singapore.
- Kaiser, H. F., & Rice, J. (1974). Little jiffy, mark IV. Educational and psychological measurement, 34(1),

111-117.

- Kelikum, I. (2021). Digital financial inclusion, informal economy and poverty reduction in Africa. Journal of Enterprising Communities: People and Places in the Global Economy, 15(4), 626-640.
- Kouladoum, J. C., Wirajing, M. A. K., & Nchofoung, T. N. (2022). Digital technologies and financial inclusion in Sub-SaharanAfrica. *Telecommunications Policy*, 102387.
- Kumar, U. (2021). Digital finance and digital financial inclusion in India. Gyanshauryam, International Scientific Refereed Research Journal, 4(5), 121-133.
- Lenka, S. K., & Sharma, R. (2017). Does financial inclusions pureconomic growth in India? The Journal of Developing Areas, 51(3), 215-228.
- Li, G., Lv, X., & Han, X. (2022). Digital financial inclusion and household debt in China. Applied Economics Letters, 1-5.
- Li, Y., Long, H., & Ouyang, J. (2022). Digital financial inclusion, spatial spillover, and household consumption: Evidence from China. *Complexity*, 2022.
- Liu, Y., Luan, L., Wu, W., Zhang, Z., &Hsu, Y. (2021). Can digital financial inclusion promote China's economic growth? *International Review of Financial Analysis*, 78, 101889.
- Lu, X., Guo, J., & Zhou, H. (2021). Digital financial inclusion development, investment diversification, and household extreme portfolio risk. *Accounting & Finance*, 61(5), 6225-6261.
- Luo, J., & Li, B. Z. (2022). Impact of digital financial inclusion on consumption inequality in China. Social Indicators Research, 163, 529-553.
- Lyman, T., & Lauer, K. (2015). What is digital financial inclusion and why does it matter? CGAP Blog, March, 10. Retrieved from https://www.cgap.org/blog/what-digital-financial-inclusion-and-why-does-itmatter Accessed November 28, 2022
- Lyons, A. C., Kass-Hanna, J., & Fava, A. (2022). Fintech development and savings, borrowing, and remittances: A comparative study of emerging economies. *Emerging Markets Review*, 51, 100842.
- Mahmood Ahmad, Abdul Majeed, MuhammadAsifKhan, Muhammad Sohaib &KhurramShehzad (2021) Digital financial inclusion and economic growth: provincial data analysis of China, *China Economic Journal*, 14(3), 291-310.
- Matthews, B. H. (2019). Hidden constraints to digital financial inclusion: the oral-literate divide. *Development in Practice*, 29(8), 1014-1028.
- Nandru, P., Chendragiri, M., & Velayutham, A. (2021). Determinants of digital financial inclusion in India: Evidence from the World Bank's global findex database. *Journal of Economic Structures, Pré-impression*.
- Naumenkova, S., Mishchenko, S., & Dorofeiev, D. (2019). Digital financial inclusion: EvidencefromUkraine. *Investment Management & Financial Innovations*, 16(3), 194.
- Niankara, I. (2023). The impact of financial inclusion on digital payment solution uptake within the Gulf Cooperation Council Economies. *International Journal of Innovation Studies*, 7(1), 1-17.
- Niu, G., Jin, X., Wang, Q., & Zhou, Y. (2022). Broadband infrastructure and digital financial inclusion in rural China. *China Economic Review*, 76 (2022), 101853.
- Ozili, P. K. (2018). Impact of digital finance on financial inclusion and stability. *Borsa Istanbul Review*, 18(4), 329-340.
- Ozili, P. K. (2022). Digital financial inclusion. *In Big Data: A game changer for insurance industry* (pp. 229-238). Emerald Publishing Limited.

- Ozturk, I., & Ullah, S. (2022). Does digital financial inclusion matter for economic growth and environmental sustainability in OBRI economies? An empirical analysis. *Resources, Conservation and Recycling*, 185, 106489.
- Özbilge, G. (2022). The impact of digital banking services on saving and borrowing behavior: The case of Turkey. Master Thesis. Retrieved from https://tez.yok.gov.tr/UlusalTezMerkezi/tezSorguSonucYeni. jspAccessed December 17, 2022
- Sarigul, H. (2020). The determining factors of financial inclusion in Turkey. Journal of Economics Finance and Accounting, 7(3), 223-235.
- Saxena P. and Thakur, K. S. (2022) Factors affecting digital financial inclusion (With Special reference to Madhya Pradesh Region). *International Research Journal of Management Science & Technology*. 13(3), 53-64.
- Sanderson, A., Mutandwa, L., & Le Roux, P. (2018). A review of determinants of financial inclusion. International Journal of Economics and Financial Issues, 8(3), 1-8.
- Shen, Y., Hu, W., & Zhang, Y. (2022). Digital finance, household income and household risky financial asset investment. *Procedia Computer Science*, 202, 244-251.
- PAL, (2020). Digital financial inclusion. Retrieved from https://media.iyzico.com/b/2021/05/PALRE-PORT_04_05_2021_ENG_RGB_v2.pdfAccessed December 17, 2022
- Takmaz, S., Sarı, E., & Alataş, S. (2022). Financial inclusion in Turkey: Unpacking the provincial inequality and its determinants. *Applied Economics Letters*, 1-9.
- Tay, L. Y., Tai, H. T., & Tan, G. S. (2022). Digital financial inclusion: A gateway to sustainable development. *Heliyon*, e09766.
- Teker, S., Teker, D., &Guzelsoy, H. (2021). Financial Inclusion: The case of Turkey. Press Academia Procedia, 14(1), 150-151.
- Tiwari, T., Srivastava, S. and Kumar, S. (2019). Adoption of digital payment methods in India, *International Journay of Electroniv Finance*.
- Tiwari, T., Srivastava, A., & Kumar, S. (2019). Adoption of digital payment methods in India. International Journal of Electronic Finance, 9(3), 217-229.
- World Bank Group. (2013). *Global financial development report 2014: Financial inclusion (Vol. 2)*. World Bank Publications.
- World Bank, (2018). The global findex database. Retrieved from https://www.worldbank.org/en/publication/ globalfindexAccessed December 17, 2022
- World Bank. 2019. "Financial Inclusion." Europe and Central AsiaEconomic Update (Spring), Washington, DC: World Bank. License: Creative CommonsAttribution CC BY 3.0IGO.
- World Bank, (2022a), Retrieved from https://www.worldbank.org/en/news/feature/2022/07/21/covid-19boosted-the-adoption-of-digital-financial-servicesAccessed December 17, 2022
- World Bank, (2022b), Retrieved from https://www.worldbank.org/en/topic/financialinclusion/publication/ digital-financial-inclusionAccessed December 17, 2022
- Williams, R. (2016). Understanding and interpreting generalized ordered logit models. *The Journal of Mathematical Sociology*, 40(1), 7-20.
- Wu, T. (2021). Quantifying coastal flood vulnerability for climate adaptation policy using principal component analysis. *Ecological Indicators*, 129, 108006.

- Yıldırım, A. C. (2020). Turkish banks and digitalization: Policy recommendations from a qualitative study. BDDK Bankacılık ve Finansal Piyasalar Dergisi, 14(2), 145-174.
- Yue, P., Korkmaz, A. G., Yin, Z., & Zhou, H. (2022). The rise of digital finance: Financial inclusion or debt trap? *Finance Research Letters*, 47, 102604.
- Yükseker, D. (2010). Debts and indebtedness as a social glue binding Turkish society. *Toplum ve Bilim*, 117, 6-18.
- Vyas, S.,& Kumaranayake, L. (2006). Constructing socio-economic status indices: how to use principal components analysis. *Health Policy And Planning*, 21(6), 459-468.
- Zhou Yuqing, He Guangwen. (2020) "The impact of the development of digital financial inclusion on the allocation of rural household financial assets". *Contemporary Economic Science*, 42(03): 92-105.



Istanbul Business Research, 53(2), 201-228

DOI: 10.26650/ibr.2024.53.1216752 http://ibr.istanbul.edu.tr/ http://dergipark.org.tr/ibr

Istanbul Business Research

Submitted: 09.12.2022 Revision Requested: 02.07.2024 Last Revision Received: 10.07.2024 Accepted: 26.08.2024

RESEARCH ARTICLE

Changes in Capital Structure in Seasoned Equity Offerings: BIST Implementation

Zeynep Kalaycıoğlu¹ (D), Ahmet Kurtaran² (D)

Abstract

This study evaluates changes in firms' capital structures after a seasoned equity offering by testing the effects of firm-specific factors on leverage ratios. The relationship between the determined dependent variables and firm-specific factors is associated with the predictions of pecking order and trade-off theory. In 3 different models in which leverage ratios are determined as dependent factors and firm-specific factors as independent variables, 10 basic hypotheses are proposed, and the relationship between the variables is examined using panel data analysis. According to the results obtained in the analyses, as the number of seasoned equity offerings increases, firms' leverage ratios decrease, and firms tend to use more debt because of the corporate tax base discount granted to seasoned equity offerings. In addition, it has been determined that rather than choosing to source resources through equity or borrowing, the view of benefiting from both sources of funds is dominant in the face of the resource requirements of seasoned equity offering firms.

Keywords: Capital structure, Leverage ratios, Firm-specific factors, Seasoned equity offerings, Panel data analysis

Introduction

Modern capital structure theories take into account some benefits and costs and seek an answer to the question, "Whether debt or equity issuance provides the most benefit with the least cost in the face of resource requirements?". These theories are consistent with the view that the issuance of equity capital is the last choice of funding source, despite differences in source preferences. Another common view of both theories is that of turning to sources of funds that provide the greatest benefit at the lowest cost. Many studies have stated that the source of funds that can provide the most appropriate choice as an external fund source among benefits and costs is borrowing. It is emphasized in many studies that interest expenses arising from debt are subject to corporate tax base deductions, making debt advantageous compared to equity capital. Borrowing is the most basic external, cost-effective fund source,



¹ Corresponding Author: Zeynep Kalaycıoğlu (Dr.), Samsun Girls Dormitory Directorate, Samsun, Turkiye. E-mail: zeynepyalcin2010@hotmail.com ORCID: 0000-0001-9635-8469

² Karadeniz Technical University Faculty of Economics and Administrative Sciences Business Administration, Trabzon, Turkiye. E-mail: kurtaran@ktu.edu.tr ORCID: 0000-0003-1780-2491

To cite this article: Kalaycioglu, Z., & Kurtaran, A. (2024). Changes in capital structure in seasoned equity offerings: BIST implementation. Istanbul Business Research, 53(2), 201-228. http://doi.org/10.26650/ibr.2024.53.1216752

and some firms turn to equity issuance as an external fund source. Moreover, it is observed that some firms periodically repeat their equity issuance with seasoned public offering (SEO) practices. This situation reveals that some firms prefer the tax advantage provided by the issuance of equity capital rather than borrowing. In this context, the reasons for firms' debt-equity changes in Turkey are considered factors worth investigating.

In this context, the reasons for firms' debt-equity changes in Turkey are considered factors worth investigating. In this study, the capital structure and modern capital structure theories are evaluated in terms of firm-specific factors, and their relationship with SEO is then outlined. By providing a literature review on SEOs; the purpose, importance, and scope of the study, the models and hypotheses created, and the method applied are explained. The findings related to the model were evaluated, and suggestions for future studies were made.

Capital Structure and Related Theories

Capital structure decisions are the first step in determining how a firm's assets are to be financed. The capital structure is how a firm finances its assets through a combination of equity and debt that includes various costs and benefits (Titman and Wessels, 1988: 1). Equity and debt composition indicates firm leverage. This combination works as a balance to minimize resource costs. The main objective is to minimize resource costs as much as possible, in line with the target leverage level.

It should be stated that capital structure decisions will be affected by a wide range of general economic conditions and firm-specific factors, and there cannot be a fixed formula that will show which factors affect the optimal leverage level target and in what direction. Therefore, different views will continue to be proposed in studies on this subject.

Evaluation of Modern Capital Structure Theories

Modern theories of capital structure outline steps towards maximizing firm value and shareholder wealth. Both theories that form a financial resource triangle between the use of a firm's internal funds, equity issuance, and borrowing have been the subject of many studies as rivals to each other. In both theories, the orientation towards the least costly resource is the main means of meeting financing needs. As a matter of fact, trade off theory argues that the optimal balance can be reached with the least cost at the target leverage level; the pecking order theory pushes aside orientation to external financing sources, taking into account the costs of borrowing. According to the theory, internal resources are the safest and least costly. Although trade-off theory directly constructs the target leverage level, the pecking order theory explains the leverage level that changes according to firm managers' financial resource preferences. Although the goal is to create a cheap financing source, the effect of the preferred
method on leverage ratios is undeniable. The firm's selection of financial resources and, thus, resource costs create leverage ratios. The combination of long-term debt and equity is a key component of the target leverage level.

According to both theories, while resource costs are of great importance, the priority order of debt differs depending on the perspective of the theories. According to the trade-off theory, the benefits of borrowing are based on the view that the tax advantage created by the corporate tax base reduction will also reduce the costs of representation between managers and shareholders. The costs of borrowing include costs related to bankruptcy and financial distress, and it is stated that these costs will increase the costs of representation. Tax advantage is the strongest pillar at the target leverage level. The pecking order theory, on the other hand, does not evaluate the benefits and costs of resources. The theory guides capital structure decisions by comparing the benefits and costs of resources. According to this theory, the costs of issuing equity capital create a huge cost pool alongside debt costs. Therefore, firms' last financial resource preference is equity issuance. The priority is internal funding sources, and the second priority is to meet the source requirement based on the tax shield effect of debt. The last option is equity issuance. However, the tax shield effect created by equity capital issuance under certain conditions is ignored.

In their study, Rajan and Zingales (1995) evaluated the question "Borrowing or financing through equity issuance?" according to firm-specific factors. Although the importance of firm-specific factors is emphasized for both theories in this study, the approaches of the theories to fundamental factors are also different from each other.

Although the trade-off theory sees market to book (M/B) as the representative of the firm's growth opportunities, the pecking order theory expresses M/B as the determinant of profitability due to differences in growth expectations. While the desire to take advantage of profitable and risky investment opportunities in the future increases debt appetite in some firms, the desire to reduce representation costs in others causes firms to abstain from debt. According to pecking order theory, the point to be considered in this case is the direction of the relationship between the amount of investment to be incurred in the face of growth opportunities and the variability in profitability. Frank and Goyal (2003) stated that leverage ratios increase as growth opportunities increase. However, considering firms that adopt a more conservative borrowing policy according to trade-off theory, while shareholders' profit share expectations increase because of high investment expenditures, agency costs also increase, and the firm becomes more sensitive to resource costs. For this reason, Titman and Wessels (1988) stated that leverage ratios decrease as growth opportunities increase. According to Rajan and Zingales (1995) and Cortez and Susanto (2012), firms with high growth opportunities tend to use less debt, although it is thought that growth opportunities may increase the future debt capacity of the firm.

According to trade-off theory, leverage ratios increase as firm size increases. This prediction is also expressed in Rajan and Zingales (1995) and Fama and French (2002). According to them, these firms can conduct long-term borrowing with much easier and cheaper costs. According to the pecking order theory, as the size of a firm increases, transaction costs and information asymmetry decrease; thus, firms' external financing requirements decrease, and due to increasing growth requirements, the tendency towards equity issuance increases. Frank and Goyal (2008) stated that leverage ratios decrease as firm size increases.

According to trade-off theory, a high ratio of tangible fixed assets is the main collateral indicator of debt supply. As a strong collateral indicator, it minimizes the representation cost of debt, that is, the risk of the lender. Therefore, leverage ratios are expected to increase as the ratio of tangible fixed assets increases. Scott (1979), Harris and Raviv (1990), Rajan and Zingales (1995), Shenoy and Koch (1996), Hirota (1999), Gaud et al. (2005), Huang and Song (2006), Antoniou et al. (2008), Frank and Goyal (2008), and Fan et al. (2012) confirmed these expectations. According to the pecking order theory, firms with high tangible assets already have sufficient internal financing resources and do not require external financing. Marsh (1982) and Harris and Raviv (1990) stated in their studies that leverage ratios decrease as the ratio of tangible fixed assets increases, according to the pecking order theory.

According to trade-off theory, the ability to pay off debt increases as profitability increases because firms with high profitability can obtain debt at a lower cost. Firms can act on their targets of higher leverage ratios because increased profitability also minimizes the costs of bankruptcy and financial distress. Mehrotra et al. (2005) stated that leverage ratios increase as profitability increases. Brealey et al. (2017), however, stated that some firms with high profitability and technology incur relatively less borrowing. Profitable firms' tendency to borrow less can be expressed as their tendency to use internal fund sources. In the pecking order theory, on the other hand, because the priority is always to use internal funding sources, internal funding sources increase as profitability increases. Therefore, as profitability increases, the tendency to borrow decreases, and leverage ratios decrease. Ross (1977), Friend and Lang (1988), Rajan and Zingales (1995), Megginson (1997), Booth et al. (2001), Drobetz and Fix (2003), and Huang and Song (2006) confirmed in their studies that leverage ratios decrease as profitability increases.

According to trade-off theory, firms with higher liquidity ratios are more capable of meeting short-term financial obligations. Therefore, they tend to borrow at higher rates. Bradley et al. (1984) stated that leverage ratios increase as liquidity increases. The pecking order theory, known for its prioritization of orientation to internal funds, argues that leverage ratios decrease as liquidity increases. Deesomsak et al. (2004), Mazur (2007), Sheikh et al. (2011), and Babu et al. (2014) supported these. Because a firm tends to meet its fund needs with existing internal resources. Therefore, according to this theory, firms having higher liquidity use less debt. Although many studies have examined the effect of ownership structure on capital structure, no clear consensus has been reached. Because of different ownership structures, capital structures are driven by different financial preferences. Friend and Lang (1988) and Berger et al. (1997) stated that borrowing is not preferred to avoid risk. According to Huang and Song's (2006), ownership structure and leverage ratios are not connected. In fact, a framework for the capital structure has been proposed in both theories. However, some firms operate outside this framework. Although theories regarding leverage ratios and firm-specific factors exist, the basic view is that firms interact differently with regard to equity issuance and debt demand.

SEO and Capital Structure Relationship

Although the initial public offering (IPO) causes the ownership structure to open up to the outside environment and the ownership to be dispersed at certain rates, it also offers some sanctions to the firm. These sanctions incur different costs depending on the firm's management practices. These costs will decrease or increase exponentially depending on the success momentum of the firm in its corporate investment and financing policies.

After the IPO, the firms that gained public company status may return to the public offering. One of the most important advantages of public offerings to firms is that they can initiate the public offering process repeatedly. After the IPO, the firm that gains the status of a publicly traded company increases its equity capital by the stipulated amount and offers shares related to the increased amount to the public by limiting all or part of the pre-emptive rights of the existing partners. This transaction is referred to as SEO. SEO is expressed as a long-term and low-cost alternative financing source that can guide firms' capital structure decisions.

The success of the firm's IPO will also affect the direction and success of subsequent SEO decisions. Indeed, Barclay et al. (2021) stated in their studies that SEOs have a wide range of effects, including a number of financial components, such as leverage ratios, dividend distribution decisions, stock repurchases, maturity structures, priority and convertible structures of securities, liquidity policies, and corporate investment decisions. No matter how wide the area of influence is, when considered in terms of capital structure theories, borrowing and fund requirements were preferred to SEOs, and it was stated that SEOs create additional costs and disadvantage for the firm. The need for additional funds for investment projects, debt repayments, a balance instrument to achieve target leverage ratios, firms' acquisitions or provision of R&D expenses, or funds for daily operating expenses, regardless of the purpose of debt financing, is a clear indication that the use of debt should stop at some point.

Jung et al. (1996) stated that the need to create additional funds to implement future investment projects or to repay debt is the basis for firms' SEO decisions. Regardless of the purpose, the target leverage level of a firm is dynamically affected before and after SEO implementation, and leverage ratios tend to increase or decrease. However, Iqbal et al. (2014) emphasized in their studies that the leverage structure and financial performance of firms that engage in SEOs differ from firms that perform very few SEOs throughout their lifecycle. In addition to these differences, the basic view is that leverage ratios decrease in the short term after SEOs and increase systematically after offers.

Method of Research

In this section, the literature on SEO practices is examined, the purpose, importance, and scope of the study are discussed, the models and hypotheses are explained, the method applied is explained, and the findings of the study are presented.

Literature Review

When the relevant literature is examined, it can be seen that very few studies exist on SEOs in Turkey. It has been observed that these studies highlighted the return performance of IPOs and the changes in financial performance. It can be seen that the international literature on SEOs has been handled very extensively. The main research topics of the studies in the international literature focused on the compatibility of assumptions about reasons for firms to turn to SEOs, the purpose of using funds obtained from SEOs, short and long-term return performance before and after SEO, changes in the financial performance of firms that go to consecutive SEOs and those that do not, and SEO and capital structure theories. In this study, basic studies that highlight the effects of SEOs on the capital structure are included.

Walker et al., (2016); in their studies that deal with the effect of specified uses of SEO revenues on the next SEOs, formed the sampling frame with 670 SEOs conducted by 276 firms operating in the USA between 1995 and 2012. According to the findings of this study, firms that disclose details about the appropriate use of SEO revenues reach higher returns in their next public offering transactions. In addition, it has been stated that firms that use funds collected through SEOs to cover the costs of investment projects have relatively more successful returns in their future public offering announcements.

Walker et al., (2019); in their studies on the provision of funds through equity issuance for firms in financial distress, formed a sampling frame with 3692 SEOs carried out by industrial firms in the United States between 1994 and 2015. According to the findings of this study, it has been stated that after the issuance of equity capital, firms in financial distress have a tendency to grow in their corporate ownership, R&D structures, and market values; they have better abnormal returns; they tend to invest following SEO; and their business performance is on the way to improvement.

Asad et al. (2020), in their studies on measuring market responses to capital structure adjustments through SEOs, formed a sampling frame with 1725 SEOs of 1016 firms operating in various industries in the United States between 2004 and 2013. According to the findings of this study, cash requirements and growth opportunities have a significant effect on SEO decisions; the desire to reach the optimal leverage ratio, and stocks' valuations have a stronger effect on the orientation to SEO. In addition, it has been stated that firms using excessive debt with SEOs approach the debt rate target and a more optimal capital structure and do not deteriorate the target capital structure for at least 3 years after the offering.

Le et al. (2020), in their studies on the factors affecting SEO decisions, formed a sampling frame with the SEO of 99 firms traded on the Vietnam Stock Exchange between 2014 and 2018. According to the findings of this study, an increase in corporate earnings and growth in the scale of a firm increase the tendency towards SEOs; an increase in dividend payment rates reduces this tendency.

Stamou et al. (2020), in their studies on the changes in the debt structures of firms that go to SEO consecutively and firms that go to SEO once, created a sampling frame with the SEO of 1033 firms operating in the London market between 1999 and 2015. According to the findings of this study, it has been stated that firms that go to consecutive SEOs have a higher debt ratio and do not tend to decrease their existing debt ratios despite obtaining additional funds due to the issuance of equity capital; thus, they maintain a higher debt level, they have a growth trend, and short-term debt levels are very low compared to firms that go the same way.

Li and Wang (2021), in their studies on the role of corporate social responsibility strategies on public offering costs, formed the sampling frame with 1163 SEOs conducted in 38 countries between 2002 and 2018. According to the findings of this study, it has been stated that corporate social responsibility practices reduce equity issuance costs; the higher the corporate social responsibility performance, the less uncertainty in the value of firm assets; thus, solid steps have been taken to strengthen the capital market with a stronger, more stable corporate structure.

Fu and Smith (2021) proposed an integrated theory of capital structure that addresses the beneficial aspects of these theories. They created a sampling framework with 8000 SEOs of 7072 industrial firms operating in the United States between 1970 and 2017. In this study, an integrated theory of the capital structure called "*strategic financial management*" has been proposed by considering the beneficial aspects of the theories. According to the proposed theory, it has been stated that strategic financial management theory represents a structure that is reached with the optimal leverage level and that turns to SEO choices to finance the firm's long-term investment opportunities and incentives.

Barclay et al. (2021), within the scope of corporate investment and financing policies, in their studies on the target leverage ratio and management of investment projects in SEOs,

formed the sampling frame with 8608 SEOs carried out by firms operating in the US and the public sector between 1970 and 2015. According to the findings of this study, a strong interaction between firms' future investment projects and capital structure decisions. They stated that the main reason that firms turn to SEOs is to generate large and new investment projects rather than reach the target leverage ratio in capital structure decisions.

Zhu and Li (2022), in their studies in which they investigated how SEOs and firm management affect corporate investment success, formed a sampling frame with SEO firms operating in China between 2009 and 2020. According to the findings of this study, SEOs generally cause a decrease in investment efficiency; it has been stated that the advanced age of management and the fact that they have an overseas background increase the efficiency of investments. Instead of increasing the number of female managers, it was suggested that they should be developed to increase their level of competence.

Chen and Liu (2022) formed a sampling frame with 1061 firms outgoing that examined the effect of increased capital from SEOs and the partial cancellation of issued shares, the effect of reduced capital on the financial structure, and the role of representation costs in this effect. According to these findings, capital decreases due to partial cancelation of issued stocks increase long-term firm performance, whereas SEOs decrease firms' performance regardless of the short and long-term. In addition, they stated that increased free cash flows due to SEOs increase the costs of representation and that representation cost, albeit partially, decrease firm performance.

Hernholm and Wormsen (2023); in their studies, they investigated the prediction that SEOs have a negative impact on firm market value. Between 2010 and 2020, they created the sample frame with 6930 SEO transactions from 630 US firms. Study; It was divided into 3 stages: pre-SEO, SEO year, and post-SEO, and the change in stock returns over time was observed. Although there was an increase in the stock returns of firms before and during the SEO period, there was a decrease in stock returns during the post-SEO period. Firms predicted to have an SEO orientation were evaluated separately, and no meaningful results were obtained. For this reason, it has been suggested that different factors may explain why SEOs reduce their return performance. According to the study, these factors are bankruptcy risk, company size, and debt ratio. For this reason, no convincing evidence has been found regarding the predictions that SEOs have a negative impact on stock returns and reduces market value.

Sanchez et al. (2023) investigated whether firms acting within the scope of corporate social responsibility tend to employ earnings manipulation practices in SEO transactions. A sampling frame was created with 4027 SEO transactions conducted by non-financial firms in France, Germany, Italy, and Spain, representing the largest EU economies between 2011 and 2020. According to the findings, there are differences between countries despite having

the same legal framework and currency. In all countries except Spain, there is cash flow manipulation before SEO. However, it has been stated that firms that attach importance to social responsibility principles are less involved in earnings manipulation.

Ashour and Atik (2024) conducted a comprehensive literature review on the subject, discussing earnings management practices before SEO, which were discussed in 1998. In this context, 846 articles on the subject were scanned between 1998 and 2022, and 38 on earnings management practices were examined. It has been stated that most articles on this subject are published in the United States. For this reason, a recommendation has been made to disseminate studies on the subject in other markets. The common point between the articles is that firms manipulated the pre-SEO earnings management approach as an opportunistic approach. As a matter of fact, pre-SEO earnings were manipulated upwards.

Gad et al. (2024) stated in their study that the ultimate purpose of seasoned equity offerings is to reduce the ratio of financial leverage through the issuance of equity capital and to evaluate new investment opportunities with newly acquired funds and emphasized the importance of the desired effect on financial performance. However, they stated that specific effects will arise along with financial effects. In this context, 903 seasoned equity offering transactions conducted by US non-financial firms between 2002 and 2017 are discussed. They suggested that firms that turn to seasoned equity offerings may cut or increase their workplace health and safety expenses to increase the success of their public offerings. According to the findings, as seasoned equity offerings approach, workplace health and safety expenses are limited, albeit in the short term. The firm-specific effects of this limitation are felt in the medium and long term as a decrease in financial performance and efficiency and failure in the next public offering attempt.

He et al. (2024) opposed the view that banks should focus on increasing equity capital in response to their fund requirements. On the contrary, they stated that banks voluntarily tend to have equity capital that exceeds the legal minimum. In this context, the aim of this study is to reveal banks' motives for acquiring SEOs. In this study, among 2141 US banks between 1985 and 2013, banks that tended to offer SEOs and those that did not were examined comparatively. According to these findings, there was a significant increase in the total assets of banks that opted for SEOs compared to those that did not engage in SEOs. This increase continued to increase 5 years after the start of SEO. It has been stated that banks that turn to SEO make more purchases than those that do not. Accordingly, funds obtained through SEO are invested in loans for sale and other types of loans. This situation shows that banks are turning to SEOs, especially for asset expansion.

Ho et al. (2024) stated that firms exposed to negative environmental and social events tend to decrease their SEO. In this context, a sample frame was created using 1999 SEO transactions of 991 firms from 25 countries between 2009 and 2017. Firm managers calculate the loss

of reputation of the institution and find that investors have higher return expectations in return for the loss of trust. Despite this, it has been stated that firms turning to SEO face restrictions even in obtaining funds through debt.

Purpose and Importance of the Research

Studies on the relationship between SEOs and capital structure in Turkey generally focus on the short and long-term return performance of IPOs and changes in financial performance. It can be seen that the international literature on SEOs has been handled very extensively. The main research topics of the studies in the international literature focused on the reasons for firms to turn to SEOs, the purpose of the use of funds obtained from SEOs, short- and long-term return performance before and after SEO, the difference between the capital structure of firms that perform consecutive SEOs and those that do not, and the compatibility of the assumptions of SEO and capital structure theories. This study is considered an important step in evaluating firms' capital structures after conducting SEOs by testing the effect of firm-specific factors on leverage ratios in SEOs, and it is thought that this study will make a significant contribution to the literature. In this context, the main purpose of this study is to evaluate changes in the capital structure of firms after conducting SEOs by testing the effect of firm-specific factors on leverage ratios.

Scope of the Research

In this study, an unbalanced panel data analysis was used to explain the effect of firm-specific factors on leverage ratios in SEOs, and the Stata/MP 14.0 package program was used for the analysis. In the application part of the study, sports services and financial leasing firms, real estate investment trusts, insurance firms, investment firms, institutions operating in the banking sector, and securities investment trusts; firms that do not go to the SEO despite being traded in the BIST All Index and firms that have zero public offering density, have been transferred, have missing data in their financial statements, and have reduced their capital within the scope of the period covered in the SIST All Index over the 16-year period between 2005 and 2020 were included in this study. It was determined that the 41 firms included in the analysis made 52 SEOs over a 16-year period. All numerical data obtained are published by the Public Disclosure Platform (KAP), Borsa Istanbul (BIST), and provided by FINNET Electronic Publishing. The calculation methods of the variables used in this study and their abbreviated symbols of the variables are summarized in Table 1. Please refer to the Appendices to Table 1.

Research Hypotheses

Within the framework of the purpose of this study, 10 basic hypotheses and 37 subhypotheses were developed. The hypotheses established in this framework are supported by the literature.

Symbol	Dependent Variables	Calculation Method
D _{ST} /E	Short-Term Debt to Equity Ratio	Short-Term Debt / Total Equity
D_{LT}/E	Long-Term Debt to Equity Ratio	Long-Term Debt / Total Equity
D/E	Debt to Equity Ratio	Total Debt / Total Equity
	Independent Variables	
AGR	Growth Asset Ratio	((Assets at The End of The Period – Assets at The Beginning of The Period)/Assets at The Beginning of The Period)*100
M/B	Market to Book Ratio	(Market Capitalization / Total Book Value
TA	Tangible Asset Ratio	Fixed Assets / Total Assets
NPM	Net Profit Margin	Net Profit / Net Sales
ROE	Return On Equity	Net Profit / Total Equity
QR	Quick Ratio	(Current Assets – Inventories – Other Current Assets) / Short- Term Debts
DPO	Density of Public Offering	Number of SEOs / Age of Firm
PMV _{log}	Public Market Value of Firms	Total Market Value * Free Float Ratio (Included in the Analy- sis by Taking Natural Logarithm)
FFR	Free Float Rate	Amount of Publicly Held Capital / Total Equity
GRE	Growth Rate of Equity	((Equity at The End of The Period – Equity at The Beginning of The Period)/Equity at The Beginning of The Period)*100
GRNS	Growth Rate of Net Sales	((Net Sales at The End of The Period – Net Sales at The Beginning of The Period) / Net Sales at The Beginning of The Period)*100
	Dummy Variables	
TD	Tax Discount	Firms that have increased their cash capital have been entitled to a discount in their corporate tax returns since July 2015, if they meet certain conditions. Firms are divided into two gro- ups those that receive tax deductions and those that do not.
ICC	Increasing Cash Capital	Firms are divided into two groups as those that increase cash capital and those that do not.

Table 1: Variables of the Study

Footnotes to the table: This table contains abbreviated symbols for all variables used in the research article and explanations for their calculation methods.

According to the pecking order theory, as profitability increases, demand for borrowing decreases and leverage ratios decrease. Therefore, as profitability ratio increase, leverage ratios are expected to decrease. Within the scope of profitability ratios, NPM and ROE are included in the analysis as independent variables. In this context, the first hypothesis of the study and the sub-hypotheses are as follows:

H₁: In SEOs, leverage ratios decrease as profitability ratios increase.

- H_{1a}: As ROE increases, D_{ST}/E decreases.
- H_{1b}: As NPM increases, D_{ST}/E decreases.
- H_{1c}: As ROE increases, D_{LT}/E decreases.
- **H**_{1d}: As NPM increases, D_{LT}/E decreases.

- **H**_{1e}: As ROE increases, D/E decreases.
- **H**_{1f}: As NPM increases, D/E decreases.

According to the pecking order theory, leverage ratios are expected to increase as growth opportunities arise. Within the scope of growth opportunities, M/B and PMV are included as independent variables. In this context, the second hypothesis of the study and the sub-hypotheses are as follows:

H₂: In SEOs, leverage ratios increase as growth opportunities increase.

- H_{2a}: As PMV increases, D_{ST}/E increases.
- H_{2b}: As M/B increases, D_{LT}/E increases.
- H_{2c}: As PMV increases, D_{LT}/E increases.
- **H**_{2d}: As PMV increases, D/E increases.

According to trade-off theory, the collateral nature of tangible assets paves the way for low-cost borrowing. Therefore, it is expected that borrowing will increase as tangible assets increase. Alves and Ferreira (2011) evaluated borrowing in two ways: long-term and short-term. According to them, as the ratio of tangible assets increases, the long-term debt ratio increases, whereas the short-term debt ratio decreases.

In this study, expectations overlap with the general assumption of the trade-off theory and the findings of Alves and Ferreira (2011). The TA representing the firm's asset structure is included in the analysis as an independent variable. In this context, the third hypothesis of the study and the sub-hypotheses are as follows:

H₃: In SEOs, leverage ratios increase as tangible asset ratio increases.

- H_{3a}: As TA increases, D_{ST}/E decreases.
- **H**_{3b}: As TA increases, D_{LT}/E increases.
- H_{3c}: As TA increases, D/E increases.

According to trade-off theory, leverage ratios are expected to increase as firm size increases. Because a firm's assets increase, it can engage in long-term borrowing at easy and cheap costs. According to the pecking order theory, as the size of a firm increases, the tendency to issue equity capital increases; thus leverage ratios are predicted to decrease. In this study, the GAR, GRE and GRNS were included in the analysis, representing the size of the firm. In line with the expectation trade-off theory in the findings of this study, the leverage ratios will increase as the GAR and GRNS increase; in line with the pecking order theory, the leverage ratios will decrease as the GRE increases. In this context, the sub-hypotheses established by the fourth and fifth hypotheses of this study are as follows:

H₄: In SEOs, leverage ratios increase as growth rates increase.

- H_{4a}: As GAR increases, D_{ST}/E increases.
- H_{4b} : As GRNS increases, D_{ST}/E increases.
- H_{4c} : As GAR increases, D_{LT}/E increases.
- H_{4d} : As GRNS increases, D_{LT}/E increases.
- H_{4e}: As GAR increases, D/E increases.
- H_{4f} : As GRNS increases, D/E increases.

H₅: In SEOs, leverage ratios decrease as GRE increases.

- H_{5a} : As GRE increases, D_{ST}/E decreases.
- **H**_{5b}: As GRE increases, D_{LT}/E decreases.
- **H**_{5c}: As GRE increases, D/E decreases.

According to the pecking order theory, leverage ratios are expected to decrease as liquidity ratios increase. The expectation regarding the findings of this study conforms to the pecking order theory. In this context, the sixth hypothesis of this study and the sub-hypotheses are as follows:

H₆: In SEOs, leverage ratios decrease as liquidity ratio increase.

- **H**_{6a}: As QR increases, D_{ST}/E decreases.
- H_{6b} : As QR increases, D_{LT}/E decreases.
- H_{6c}: As QR increases, D/E decreases.

In this study, FFR and DPO were included in the analysis, representing the ownership structure. Although no general conclusion could be drawn regarding the relationship between ownership structure and leverage ratios in academic studies, the expectation regarding the findings in this study is that leverage ratios decrease as FFR increases. This expectation is in line with the findings of Sayılgan and Sayman (2012) that as ownership density decreases, FFR increases, and Stulz (1988) and Mehran (1992) that leverage ratios increase as ownership density increases. Therefore, the expectation is that leverage ratios will decrease as FFR increases. Another expectation is that leverage ratios decrease will decrease as the density

of offer increases. The more the firm raises funds by investing in SEOs, the less its debt orientation will be. In this context, the sub-hypotheses established from the seventh and eighth hypotheses of this study are as follows:

H₇: In SEOs, leverage ratios decrease as FFR increases.

- H_{7a} : As FFR increases, D_{ST}/E decreases.
- H_{7b} : As FFR increases, D_{LT}/E decreases.
- H_{7c}: As FFR increases, D/E decreases.

H₈: In SEOs, leverage ratios decrease as DPO increases.

- H_{8a}: As DPO increases, D_{ST}/E decreases.
- H_{8b}: As DPO increases, D_{LT}/E decreases.
- H_{8c}: As DPO increases, D/E decreases.

According to trade-off theory, leverage ratios increase as the rate of firms benefiting from corporate tax base reductions increase. Since firms benefit from corporate tax base reductions over the funds they obtain from SEOs, firms that receive tax reductions are included in the analysis as a dummy variable, and the leverage ratios of firms that receive tax reductions are expected to increase. However, leverage ratios are expected to decrease due to the funds collected during the SEO period, and the years of the SEO are included in the analysis as a dummy variable. In this context, the sub-hypotheses established with the ninth and tenth hypotheses of this study are as follows:

H₉: SEO practices that receive tax reductions increase firms'leverage ratios.

- H_{9a} : SEO practices that receive tax reductions increase firms' D_{ST}/E .
- H_{9b} : SEO practices that receive tax reductions increase firms' D_{LT}/E .
- H_{9c} : SEO practices that receive tax reductions increase firms' D/E.

 H_{10} : Leverage ratios decrease during the year of SEO.

- H_{10a} : In the year of SEO, D_{ST} /E decreases.
- H_{10b} : In the year of SEO, D_{LT} /E decreases.
- H_{10c} : In the year of SEO, D/E decreases.

Econometric Model

Within the scope of the study, three dependent variables were used, and a separate panel data model was established for each. These models are separately analyzed for capital structure ratios, which attempt to express a firm's ability to meet its short and long-term debt obligations and its total debt obligations. In this context, three different models established to explain the relationship between leverage ratios and firm-specific factors in SEOs are statistically expressed as follows:

Model 1: One-Way Random Effects Model

$$\begin{split} D_{ST}/E &= \beta_0 + \beta_1 GAR_{it} + \beta_2 TA_{it} + \beta_3 NPM_{it} + \beta_4 ROE_{it} + \beta_5 QR_{it} + \beta_6 DPO_{it} \\ &+ \beta_7 PMV log_{it} + \beta_8 FFR_{it} + \beta_9 GRE_{it} + \beta_{10} GRNS_{it} + \beta_{11} TD_{it} + \beta_{12} ICC_{it} + u_{it} \end{split}$$

Model 2: One-Way Fixed Effect Model

$$\begin{split} D_{LT}/E &= \beta_0 + \beta_1 GAR_{it} + \beta_2 M/B_{it} + \beta_3 TA_{it} + \beta_4 NPM_{it} + \beta_5 ROE_{it} + \beta_6 QR_{it} + \beta_7 DPO_{it} \\ &+ \beta_8 PMV log_{it} + \beta_9 FFR_{it} + \beta_{10} GRE_{it} + \beta_{11} GRNS_{it} + \beta_{12} TD_{it} + \beta_{13} ICC_{it} \\ &+ u_{it} \end{split}$$

Model 3: One-Way Fixed Effect Model

$$D/E_{\text{III}} = \beta_0 + \beta_1 GAR_{it} + \beta_2 TA_{it} + \beta_3 NPM_{it} + \beta_4 ROE_{it} + \beta_5 QR_{it} + \beta_6 DPO_{it} + \beta_7 PMV log_{it} + \beta_8 FFR_{it} + \beta_9 GRE_{it} + \beta_{10} GRNS_{it} + \beta_{11} TD_{it} + \beta_{12} ICC_{it} + u_{it}$$

Research Findings

Descriptive statistics, which present the information about the variables in tabular form by showing the highest and lowest values, mean value, number of observations, and standard deviation, represent raw data. The descriptive statistics are presented in Table 2. Please refer to the appendices to Table 2.

Table 2: D	escriptive Statistics				
Variables	Number of Obser- vations	Average	Standard Devi- ation	Smallest Value	Greatest Value
GAR	656	0.18	0.36	-0.94	4.33
M/B	653	1.77	8.37	-177.54	42.39
TA	656	0.46	0.19	0.01	0.98
D _{ST} /E	656	0.97	10.09	-202.74	42.47
NPM	656	0.00	0.25	-3.63	0.64
ROE	650	-0.01	0.92	-15.72	12.36
QR	656	1.18	1.19	0.02	9.11

215

Variables	Number of Obser- vations	Average	Standard Devi- ation	Smallest Value	Greatest Value
DPO	656	0.25	0.14	0.00	0.82
PMV _{log}	656	17.83	1.50	14.91	23.12
FFR	656	0.38	0.21	0.04	0.98
D _{LT} /E	656	0.48	2.26	-46.30	21.38
D/E	656	1.45	11.53	-206.12	63.84
GRE	640	0.19	0.62	-0.81	9.30
GRNS	656	0.15	0.33	-0.79	4.69

Footnotes to the table: This table contains descriptive statistics for the variables used in the analysis.

These variables are the dependent variables used in the research article: "D_{ST}/E: Short-Term Debt to Equity Ratio, D_{LT}/E: Long-Term Debt to Equity Ratio, D/E: Debt to Equity Ratio". These independent variables are used in the research article: "AGR: Growth Asset Ratio, M/B: Market to Book Ratio, TA: Tangible Asset Ratio, NPM: Net Profit Margin, ROE: Return On Equity, QR: Quick Ratio, DPO: Density of Public Offering, PMVl_{og}: Public Market Value of Firms, FFR: Free Float Rate, GRE: Growth Rate of Equity, GRNS: Growth Rate of Net Sales".

Average of the GAR 0.18; M/B 1.77; TA 0.46; the D_{ST}/E 0.97; NPM 0.00; ROE 0.01; QR 1.18; DPO 0.25; the PMV, which was included in the analysis by taking its natural logarithm, was 17.83; FFR 0.38; D_{LT}/E 0.48; D/E 1.45; GRE 0.19; and GRNS 0.15. It can be seen that the standard deviation values of all variables deviate from the mean values. This situation can be interpreted as the values of the variables within a group are not suitable for a homogeneous distribution, and the values are at different points from their mean values. It is necessary to determine appropriate analysis management by testing whether the dataset is homogeneous.

The correlation coefficient, which indicates the direction and strength of the relationship between variables, takes the value between -1 and +1. If the coefficient is positive, it indicates the same direction between the two variables; if it is negative, it indicates the opposite relationship between the two variables. The fact that the correlation coefficient between the two variables is close to 1 in absolute value indicates a strong relationship between them. For a significant correlation coefficient, the coefficient between variables is close to 0. Variables with a correlation coefficient of 0.60 or higher were excluded from the same model in this study. The correlation matrix used in this study is presented in Table 3. Please refer to the appendices to Table 3.

Table 5: Res	suits of Q	Correlat	ion Ana	Iysis										
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
(1) GAR	1.000													
(2) M/B	0.026	1.000												
(3) TA	-0.030	-0.019	1.000											
(4) D _{ST} /E	0.031	0.920	-0.049	1.000										
(5) NPM	0.180	0.093	-0.213	0.042	1.000									
(6) ROE	0.090	0.004	-0.038	-0.041	0.216	1.000								
(7) QR	0.012	0.007	-0.197	-0.023	0.207	0.088	1.000							
(8) DPO	0.039	-0.071	-0.182	-0.063	-0.020	-0.075	-0.004	1.000						
(9) PMV _{log}	0.105	0.102	0.090	-0.006	0.174	0.045	0.129	0.023	1.000					
(10) FFR	-0.038	-0.033	0.073	-0.014	-0.160	-0.042	0.109	-0.068	0.069	1.000				
(11) D _{LT} /E	0.050	0.482	0.110	0.567	-0.004	-0.058	-0.070	-0.020	0.043	-0.059	1.000			

Table 3: Results of Correlation Analysis

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
(12) D/E	0.037	0.899	-0.021	0.987	0.036	-0.047	-0.034	-0.060	0.003	-0.024	0.693	1.000		
(13) GRE	0.311	0.042	0.018	-0.057	0.070	0.056	-0.012	0.066	0.142	0.036	-0.040	-0.060	1.000	
(14) GRNS	0.464	0.040	-0.062	0.029	0.201	0.140	-0.048	0.026	0.120	-0.036	0.061	0.037	0.105	1.000
Footnotes to t	the table:	This tal	ole includ	les the re	sults of t	he correl	ation ana	lysis for	the varia	bles used	l in the a	nalysis.		

These variables are the dependent variables used in the research article: " D_{ST}/E : Short-Term Debt to Equity Ratio, D_{LT}/E : Long-Term Debt to Equity Ratio, D/E: Debt to Equity Ratio". These independent variables are used in the research article: "AGR: Growth Asset Ratio, M/B: Market to Book Ratio, TA: Tangible Asset Ratio, NPM: Net Profit Margin, ROE: Return On Equity, QR: Quick Ratio, DPO: Density of Public Offering, PMVlog: Public Market Value of Firms, FFR: Free Float Rate, GRE: Growth Rate of Equity, GRNS: Growth Rate of Net Sales".

According to Table 3, a high correlation between the M/B variable and the $D_{ST}/E - D/E$ variables. For this reason, M/B is not included among the independent variables in models 1 and 3. The high correlation between the D_{ST}/E and D/E was not considered because these variables were excluded from the same model.

In the application phase of the study, panel data were used, and all stages required by the panel data model were applied sequentially. The steps taken during the selection phase of the appropriate panel data model are illustrated in Chart 1.



Chart 1: Steps to Apply the Panel Data Model

Cross-section dependence was determined by applying the Pesaran (2004) CD test, which is appropriate for small time dimensions and large unit sizes. The hypotheses for testing cross-sectional dependence are expressed as follows:

- H_0 : There is no dependence between units.
- H_1 : There is a dependence between units.

The cross-section dependence test is expressed as follows for the unbalanced panel:

$$CD = \sqrt{\frac{2}{N(N-1)}} \left[\sum_{i=1}^{N-1} \sum_{j=i+1}^{N} \sqrt{T_{ij}} \rho_{ij} \right]$$

According to this formula, "N" refers to the number of units. According to this formula, "N" refers to the number of units. "N(N-1)" calculates the correlation of each unit with all units other than itself. "Tij" correlation coefficient is the calculated number of observations; "Pij" refers to the correlation coefficient of the unit residues. Table 4 presents the crosssectional dependency test results. Please refer to the Appendices to Table 4.

Variables	Test Statistics	Probabilty Value	Variables	Test Statistics	Probabilty Value
GAR	7.64	0.00	QR	5.93	0.00
GRNS	12.07	0.00	DPO	66.8	0.00
M/B	41.30	0.00	PMV _{log}	70.34	0.00
TA	0.95	0.34	FFR	23.18	0.00
D _{ST} /E	11.83	0.00	D/E	14.89	0.00
NPM	3.23	0.00	$\mathbf{D}_{\mathrm{LT}}/\mathbf{E}$	14.66	0.00
GRE	6.20	0.00	ROE	2.64	0.00

Table 4: Cross-Section Dependency Test Results for the Variables

Footnotes to the table: This table contains the cross-section dependency test results.

These variables are the dependent variables used in the research article: "D_{ST}/E: Short-Term Debt to Equity Ratio, D_{LT}/E: Long-Term Debt to Equity Ratio, D/E: Debt to Equity Ratio". These independent variables are used in the research article: "AGR: Growth Asset Ratio, M/B: Market to Book Ratio, TA: Tangible Asset Ratio, NPM: Net Profit Margin, ROE: Return On Equity, QR: Quick Ratio, DPO: Density of Public Offering, PMVI_{og}: Public Market Value of Firms, FFR: Free Float Rate, GRE: Growth Rate of Equity, GRNS: Growth Rate of Net Sales".

According to the results of the cross-section dependency test, the H_0 hypothesis, which is the main hypothesis, was established as "there is no dependence between units". There is cross-sectional dependence in all variables except for the TA variable, and the H_0 hypothesis suggesting inter-unit independence is rejected. For this reason, while the second generation unit root test was applied to all variables with cross-sectional dependence; the first generation unit root test was applied to the TA variable, which has no cross-sectional dependence.

During the unit root tests, the Im–Pesaran and Shin (IPS) Panel Unit Root Test was used. The main feature of the IPS test is that it is based on the average of different unit root statistics. In the IPS test, the hypothesis that there is a unit root in at least one unit is tested against the hypothesis that there is a unit root in all units. The hypotheses established according to the IPS unit root test are expressed as follows:

- **H**₀: All units contain unit roots.
- **H**₁: Some units are stationary.

The IPS panel unit root test is expressed as follows:

$$\bar{t}_{bar} = \frac{1}{N} \sum_{i=1}^{N} t_{iT}$$

According to the formula, \bar{t} represents the arithmetic mean of the t-values, and the unit root is calculated according to the \bar{t}_{bar} statistic. Panel unit root test results are presented in Table 5. Please refer to the Appendices to Table 5.

Second Generation	P Value	IPS Statistics	Second Generati-	P Value	IPS Statistics	
CAD	0.00	12.01	DDO	0.00	2.57	
GAR	0.00	-13.91	DPO	0.00	-3.5/	
GRNS	0.00	-15,16	PMV _{log}	0.00	-3.98	
M/B	0.00	-11,04	D/E	0.00	-30.47	
D _{ST} /E	0.00	-34,65	$\mathbf{D}_{\mathrm{LT}}/\mathbf{E}$	0.00	-13,26	
NPM	0.00	-5.78	ROE	0.00	-10.49	
QR	0.00	-5.69	FFR	0.01	-2.22	
First Generation	D Value	IDS Statistics				
Variables	r value	115 Statistics				
TA	0.00	0.00				
IA	0.00	0.00				_

Table 5: IPS	Unit Root	Test Results
--------------	-----------	--------------

Footnotes to the table: This table contains the unit root test results.

These variables are the dependent variables used in the research article: "D_{ST}/E: Short-Term Debt to Equity Ratio, D_{LT}/E: Long-Term Debt to Equity Ratio, D/E: Debt to Equity Ratio". These independent variables are used in the research article: "AGR: Growth Asset Ratio, M/B: Market to Book Ratio, TA: Tangible Asset Ratio, NPM: Net Profit Margin, ROE: Return On Equity, QR: Quick Ratio, DPO: Density of Public Offering, PMVl_{og}: Public Market Value of Firms, FFR: Free Float Rate, GRE: Growth Rate of Equity, GRNS: Growth Rate of Net Sales".

According to the IPS unit root test results, the H_0 hypothesis was rejected in all variables, and the series was stationary. After determining stationarity, the Hausman test was applied in accordance with the fixed coefficient panel data model of the models for which homogeneity was determined, and the fixed and random effects models were estimated separately. The heteroscedasticity, autocorrelation, and correlation between units were determined, and the type of resistant estimator suitable for the models was determined.

In Table 6, model 1; in the presence of heteroscedasticity and autocorrelation, the estimation results show that robust standard errors with clustered standard errors are observed in the random-effects model. The F statistic is significant and R² is 31.4%. In other words, the independent variables explain 31.4% of the change in D_{ST}/E , which is the dependent variable. The TA, QR, and DPO have a negative relationship with D_{ST}/E , whereas the PMV has a positive relationship. One-unit changes in TA, QR, and DPO decrease by 2.66, 0.23, and 3.43 units, respectively. A 1% increase in PMV increases the D_{ST}/E by 0.34%.

In Table 6, model 2; in the presence of heteroscedasticity, autocorrelation, and inter-unit correlation, resistant standard error estimation results made with clustered standard errors are shown in the fixed-effects model. The F statistic is significant and R² is 46.6%. In other words, the independent variables explain 46.6% of the change in D_{LT}/E , the dependent variable. GAR, M/B, TA, GRNS, and tax deduction have a positive relationship with D_{LT}/E , whereas DPO and PMV have a negative relationship. One-unit changes in GAR, M/B, TA, and GRNS lead to increases of 0.18, 0.16, 1.65, and 0.21 units, respectively; A one-unit change in DPO causes a 1.9-unit decrease. A 1% increase in PMV reduced D_{LT}/E by 0.2%. The increase in D_{LT}/E of firms that receive tax deductions is 0.48 units higher than that of firms that do not receive tax deductions.

Table 6: Results of Panel I	Jata Analysis	on The Effec	t of Firm-Sp	ecific Factors	on Leverag	ge Katios in	SEOS					
Estimator 1ype		Kandom	Effects			Fixed	Effects			Fixed E	flects	
Resistive Estimator	Ar	ellano, Froot	, and Roger	8		Driscoll 8	and Kraay			Driscoll an	id Kraay	
Dependent Variable		Model 1:	D _{ST} /E			Model 2	2: D _{LT} /E			Model 3	: D/E	
Independent Variable	Coef.	St. Err.	t-value	p-value	Coef.	St.Err.	t-value	p-value	Coef.	St.Err.	t-value	p-value
GAR	0.29	0.34	0.88	0.38	0.18	0.05	3.27	0.00***	0.29	0.42	0.68	0.498
M/B			ı	'	0.16	0.15	10.70	0.00^{***}				
TA	-2.66	1.27	-2.07	0.038^{**}	1.65	0.41	4.03	0.00^{***}	0.00	1.39	0.00	0.999
NPM	0.76	1.44	0.52	0.6	0.085	0.10	0.84	0.406	0.80	1.11	0.72	0.473
ROE	-3.94	2.23	-1.76	0.078^{*}	-0.12	0.13	-0.95	0.350	-4.05	1.82	-2.22	0.033^{**}
QR	-0.23	0.93	-2.48	0.013^{**}	0.22	0.01	1.49	0.145	-0.12	0.58	-2.08	0.044^{**}
DPO	-3.43	1.71	-2.01	0.044^{**}	-1.91	0.86	-2.23	0.032^{**}	-7.17	4.06	-1.76	0.085^{*}
PMV _{log}	0.34	0.13	2.66	0.008^{***}	-0.2	0.07	-2.84	0.007***	0.48	0.11	4.23	0.000^{***}
FFR	-1.74	1.02	-1.71	0.088^{*}	0.37	0.39	0.92	0.364	-1.99	1.22	-1.63	0.110^{*}
GRE	-0.29	0.27	-1.07	0.287	-0.08	0.048	-1.77	0.085^{*}	-0.46	0.22	-2.11	0.041^{**}
GRNS	0.42	0.23	1.83	0.067*	0.21	0.044	4.87	0.00^{***}	0.56	0.27	2.05	0.047 * *
ICC ₁	-0.15	0.24	-0.06	0.952	-0.10	0.07	-1.37	0.177	0.02	0.21	0.11	0.911
TD_1	-0.11	0.39	-0.28	0.778	0.48	0.23	2.06	0.046^{**}	0.19	0.38	0.51	0.612
Fixed	-1.57	1.71	-0.92	0.358	3.19	1.15	2.77	0.008^{***}	-3.88	1.97	-1.96	0.056^{*}
Number of Observa-		. 64					010			. 61		
tions			5								2	
Number of Units		:41					41				-	
\mathbf{R}^2		: %31	4.				9.6			: 24	S	
P probability value		: 0.0	0			0	00.			: 0.0	00	
Homogeneity Test	Swamy:	0.078, Proba	bility Value:	0.938	Swamy:	-0.371, Pro	bability Val	ue: 0.711	Swamy:	0.845, Probe	ability Valu	e: 0.398
	Della	J.102, FIODAL Tast Statisti	ounty value: v	710.0	Della: -	U.YU /, FIUU Tact Static	anniy van tie: 2256	10:01	Della: I	./J/, FIODat	ounty value	6/0.0
Hausman Test		Probability Va	cs. 17.42 alue: 0.135			Probability	Value: 0.04	-	Ι	Probability V	/alue: 0.04	
Levene, Brown, and	W0:	8.56, Probabi	ility Value: 0	00.		, .			ł			
Forsythe Test for	W50:	4.19, Probab	ility Value: 0	.00		Lest Statisti	CS: 1003-57	0 -		est Statistics	5: 45042.22 22.0400	
Heteroskedacity	W10:	5.76, Probab	ility Value: 0	.00		Frodability	value: 0.00		-	rrooaomy v	value: 0.00	
Durbin-Watson and		Durbin-Wat	son: 0.68			Durbin-W	atson: 0.92			Durbin-Wat	tson: 0.67	
Baltagi-Wu LBI Tests		Baltaoi-W	u: 1.21			Baltaoi-	Wu: 1.29			Baltaoi-W	/u: 1.22	
for Autocorrelation		D D				Q				D		
Inter-Unit Correlati-					Tact Static	tice. 5 25 1	r vyhahilitar	000.ettle/	Tact Statict	ice. 8 00 Dr	whahility V	00.0.0016
on-Pesaran Test					T CSL STALLS	1,02,0.2,1	TUUAUIIIIY	value. 0.00	10101010101	JCS. 0.77, FI	υυαυπική ν	aluc. 0.00
Footnotes to the table: This 1	able contains r	esults of panel o	lata analysis o	n the effect of f	irm-specific 1	factors on lev	erage ratios i	n SEOs.				
*** p<.01, ** p<.05, * p<.1 re	ters to statistica	al significance a	it the 1%, 5%,	and 10% levels	, respectively							
These variables are the dependent	lent variables ı	ised in the resea	arch article: "1	O _{ST} /E: Short-Te	erm Debt to	Equity Ratio,	D _{LT} /E: Long	r-Term Debt to	Equity Ration	o, D/E: Debt	to Equity Ra	tio". These
independent variables are used	l in the researcl	n article: "AGR	: Growth Asse	t Ratio, M/B: N	tarket to Boc	k Ratio, TA:	Tangible Ass.	et Ratio, NPM	: Net Profit]	Margin, ROE:	Return On J	Equity, QR:
Quick Ratio, DPO: Density of	Public Offerin	g, PMV1 _{og} : Pul	olic Market Va	lue of Firms, F	FR: Free Flo	oat Rate, GRI	E: Growth Ra	tte of Equity, C	FRNS: Grow	th Rate of Net	Sales". Th	ese dummy

variables are used in the research article: "TD: Tax Discount, ICC: Increasing Cash Capital".

220

In Table 6, model 3; in the presence of heteroscedasticity, autocorrelation, and inter-unit correlation, resistant standard error estimation results made with clustered standard errors are shown in the fixed-effects model. The F statistic is significant and R² is 24.3%. In other words, the independent variables explain 24.3% of the change in D/E, which is the dependent variable. ROE, QR, and GRE have a negative relationship with D/E, whereas PMV and GRNS have a positive relationship. An increase of 1% in the PMV coefficient increases 0.48% in D/E. A one-unit change in GRNS leads to a 0.56' unit increase in D/E; A one-unit change in ROE, QR, and GRE causes a decrease of 4.05, 0.12, and 0.46 units, respectively. Please refer to the appendices to Table 6.

Evaluation of Findings

The expectations regarding the hypotheses put forward in this part of the study and the compliance of the analysis results with these expectations are summarized in Table 7. Please refer to the Appendices to Table 7.

The pecking order theory suggests that as profitability increases, the need for borrowing decreases. The analysis results are consistent with this theory. Ross (1977), Friend and Lang (1988), Rajan and Zingales (1995), Megginson (1997), Booth et al. (2001), Drobetz and Fix (2003), and Huang and Song (2006) support this theory. According to the findings of this study, as firms' ROEs increase, their D/E decreases. H_{1e} hypothesis has been accepted as "*As ROE increases, D/E decreases*". According to a general opinion, the primary preference

	Mode	l 1: D _{ST} /E	Mode	l 2: D _{LT} /E	Mod	el 3: D/E
Variables		С	ompliance of F	indings with Hypot	heses	
	Expectation	Conclusion	Expectation	Conclusion	Expectation	Conclusion
AGR	Positive	Meaningless	Positive	H _{4c} /Acceptance	Positive	Meaningless
M/B	-	-	Positive	H _{2b} /Acceptance	-	-
ТА	Negative	H _{3a} /Acceptance	Positive	H _{3b} /Acceptance	Positive	Meaningless
NPM	Negative	Meaningless	Negative	Meaningless	Negative	Meaningless
ROE	Negative	Meaningless	Negative	Meaningless	Negative	H _{1e} /Acceptance
QR	Negative	H _{6a} /Acceptance	Negative	Meaningless	Negative	H _{6c} /Acceptance
DPO	Negative	H _{8a} /Acceptance	Negative	H _{8b} /Acceptance	Negative	Meaningless
PMV log	Positive	H _{2a} /Acceptance	Positive	H _{2c} /Rejected	Positive	H _{2d} /Acceptance
FFR	Negative	Meaningless	Negative	Meaningless	Negative	Meaningless
GRE	Negative	Meaningless	Negative	Meaningless	Negative	H _{5c} /Acceptance
GRNS	Positive	Meaningless	Positive	H _{4d} /Acceptance	Positive	H _{4f} /Acceptance
ICC ₁	Negative	Meaningless	Negative	Meaningless	Negative	Meaningless
TD_1	Positive	Meaningless	Positive	H _{9b} /Acceptance	Positive	Meaningless

Table 7:	Compatibility	of Unbalanced Panel	Data Analysis Test	Results with Hypotheses
			2	21

Footnotes to the table: This table summarizes the Compatibility of Unbalanced Panel Data Analysis Test Results with Hypotheses. These variables are the dependent variables used in the research article: " D_{ST}/E : Short-Term Debt to Equity Ratio, D_{LT}/E : Long-Term Debt to Equity Ratio, D/E: Debt to Equity Ratio". These independent variables are used in the research article: "AGR: Growth Asset Ratio, M/B: Market to Book Ratio, TA: Tangible Asset Ratio, NPM: Net Profit Margin, ROE: Return On Equity, QR: Quick Ratio, DPO: Density of Public Offering, PMVlog: Public Market Value of Firms, FFR: Free Float Rate, GRE: Growth Rate of Equity, GRNS: Growth Rate of Net Sales". These dummy variables are used in the research article: "TD: Tax Discount, ICC: Increasing Cash Capital". in SEOs has been internal funds, and debt payments have remained below the accumulated earnings, thus reducing the total debt ratio.

According to the pecking order theory, as firm's PMV increases, growth and investment opportunities also increase, and there will be a greater tendency to borrow. According to the results of the analysis, which is consistent with the theory, as firm's PMV increased, the D_{ST}/E and D/E increased. The hypotheses H_{2a} "As PMV increases, D_{ST}/E increases" and H_{2d} "As PMV increases, D/E increases" are accepted. According to these findings, the relationship between the D_{LT}/E and PMV is negative, and the hypothesis of H_{2c} "As PMV increases, D_{LT}/E increases" can not be confirmed. This finding does not agree with the general opinion proposed by the pecking order theory. Titman and Wessels (1988), Rajan and Zingales (1995), and Cortez and Susanto (2012) stated that firms that adopt conservative leverage policies tend to borrow less despite high growth opportunities. For this reason, it has been suggested that as growth opportunities increase, borrowing requirements decrease. The findings are in line with this prediction. As PMV increases, firms' D_{LT}/E of declines. It is observed that firms prefer short-term borrowing by structuring their long-term debts using funds obtained when the PMV increases after SEOs.

According to these findings, the relationship between the D_{LT}/E and M/B is consistent with pecking order theory. The H_{2b} hypothesis is accepted as "As M/B increases, D_{LT}/E increases". This finding is also expressed in Rajan and Zingales (1995), who suggested that leverage ratios decrease in firms with equity issuance.

These findings are in line with the prediction expressed in the studies of Mira (2005) and, Alves and Ferreira (2011) that tangible fixed assets that can be shown as collateral increase borrowing capacity and reduce the cost of long-term debt. H_{3a} "As TA increases, D_{ST}/E decreases" and H_{3b} "As TA increases, D_{LT}/E increases" hypotheses have been accepted. As firms' TA increased, the D_{LT}/E also increased, but D_{ST}/E decreased. These findings show that firms with increasing tangible assets prefer less costly, longer-term borrowing rather than short-term debt.

According to trade-off theory, when firms' growth rates are taken into consideration, growth based on assets and sales also causes trust and collateral factors to come to the fore. For this reason, depending on assets and sales growth rates, it is expected that firm's tendency to borrow money will increase. The analysis results agree with these expectations. As firm's GAR and the GRNS of firms increase, the D_{LT}/E also increases; as the GRNS increases, D/E also increases. For this reason, H_{4c} "As GAR increases, D_{LT}/E increases.", H_{4d} "As GRNS increases, D/E increases.", H_{4d} "As GRNS increases, D/E increases." hypotheses are accepted. Frank and Goyal (2003) and Delcoure (2007) stated that as the size of a firm increases, agency costs decrease; accordingly, long-term borrowing with easy and cheap costs is paved. This view is also expressed in Rajan and Zingales (1995), Fama and French (2002).

Rajan and Zingales (1995) stated that the need for borrowing decreased due to firms' growth tendencies due to the capital increases and preferred to turn to equity issuance due to increasing growth requirements. This view is in line with the pecking order theory. According to these findings, H_{5c} hypothesis "As GRE increases, D/E decreases" was accepted.

According to the pecking order theory, firms with high solvency tend to borrow less. This prediction was also supported by Deesomsak et al. (2004), Mazur (2007), Sheikh et al. (2011), and Babu et al. (2014). The analysis results are consistent with this prediction. For this reason, H_{6a} "As QR increases, D_{ST}/E decreases" and H_{6c} "As QR increases, D/E decreases" hypotheses are accepted. In SEOs, it can be stated that firms pay short-term debt with increased liquidity, but this view cannot be generalized for long-term debt. The relationship between the D_{LT}/E and QR could not be confirmed.

The relationship between FFR and leverage ratio could not be confirmed. This finding agrees with Huang and Song (2006) that ownership structure and leverage ratios are not connected.

According to these findings, as the DPO of the firms increased, the D_{ST}/E and D_{LT}/E decline. For this reason, H_{8a} "As DPO increases, D_{ST}/E decreases" and H_{8b} "As DPO increases, D_{LT}/E decreases" hypotheses are accepted. When firms' fund inflows from SEOs were repeated, their tendency to borrow decreased. Therefore, in SEOs, it can be stated that firms' tendency to default due to equity issuances has decreased. Short and long term funding requirements are met funds obtained from SEOs.

According to the findings of this study, the prediction that firms' leverage ratios decrease during the SEO period is meaningless. However, due to SEO practices, the D_{LT}/E of firms that receive tax deductions has increased. The increase in the D_{LT}/E for these firms is compatible with the trade-off theory. For this reason, H_{9b} *"SEO practices that receive tax reductions increase firms'* D_{LT}/E *"* hypothesis has been accepted. According to these findings, the D_{LT}/E of firms subject to corporate tax base reduction due to SEOs has increased. The tax factor, which has an important place in the optimal leverage level, can change the direction of firms' financial resource preferences. Although the first advantage that comes to mind is that the interest expenses incurred due to borrowing can be deducted from the corporate tax base when it comes to tax advantages, it is clear that SEOs made through capital increase also create a tax advantage.

Conclusion and Recommendations

Capital structure, which has been the subject of much research with the view that perfect competition conditions cannot always be met under all conditions and the tax factor cannot be ignored, continues research on capital structure decisions. In the studies carried out, the debt and equity preferences in firms' capital structure decisions and the existence of the optimal leverage level have formed the main point of this research. In many studies, the superior aspects of the theories have been tried to be revealed, and it has been emphasized that both theories are rivals of each other, but there are differences in the fund source preferences of firms due to firm, sector, or country conditions. Although the studies developed under the leadership of modern capital structure theories have not been able to reach a consensus on debt or equity capital preferences against firms' fund requirements, the main view is that both theories contribute significantly to the literature, and these theories should be evaluated as a whole.

When the findings of this study are evaluated, they are found to be generally compatible with modern capital structure theories. While the prediction that firms' long-term borrowing rates will increase according to trade-off theory has been confirmed, as the guarantee nature of a firm's tangible assets paves the way for low-cost and long-term borrowing, according to pecking order theory, the prediction that increased equity profitability ratios due to SEOs will decrease total borrowing rates has also been confirmed in the analysis results. While some analysis results are compatible with the predictions of the trade-off theory, others are compatible with the pecking order theory. It can be stated that the firms that go to SEO do not comply with the financial hierarchical order, and the view of benefiting from both funding sources is dominant rather than choosing between equity capital or financing through borrowing in the face of fund requirements. Therefore, instead of seeing the theories as rivals by focusing on the shortcomings or superiorities of both theories, the view that firms' fund requirement preferences can be changed by taking into account firm-specific factors and evaluating both capital structure theories as a complementary whole comes to the fore.

It can be interpreted that the efforts of firms to meet their fund needs by issuing equity capital are replaced by the corporate tax base reduction advantage provided by SEOs compared with the corporate tax base reduction advantage provided by borrowing. As a matter of fact, subjecting the amount calculated over the cash increase due to SEOs to corporate tax base reductions, SEOs are an important external fund source that provides financing with equity capital. Therefore, it is expected that the probability of repeating SEO at certain intervals in the future will increase considering firms' fund requirements.

According to the results of the analysis, the assets and sales of firms that engage in SEOs in Turkey have increased, and this increase and the benefit of corporate tax base reduction due to SEOs increase their tendency to engage in long-term borrowing. It can be stated that firms with increased SEO density have decreased short and long-term leverage ratios and pay their short-term debts by giving priority to internal fund resources due to the increased liquidity ratio.

In future studies, considering leverage ratios and country-specific factors, not limiting the sector to manufacturing firms only, and comparing firm data from different sectors in SEOs, the time interval, including before and after the SEO evaluation quarterly, and including a longer time period in the research may be important to review the findings more generally.

There are no academic studies in Turkey on the capital structure of SEOs. For this reason, regarding SEO transactions, whether the changes in the capital structures of firms before and after public offering transactions cause a difference when certain maturities are considered and when the purposes of use of the funds obtained through SEOs are considered, it is possible to make a more comprehensive evaluation.

Considering the low number of SEOs in Turkey, it is seen that some firms do not conduct SEOs after the IPO, while some firms conduct SEOs and repeat their SEOs at regular intervals. Changes in the capital structures of firms that did not participate in SEO after the IPO and those that did participate are also seen among the subjects worth researching.

References

- Alves, P. F. P., Ferreira, M. A. (2011). Capital Structure and Law Around The World. Journal Of Multinational Financial Management, 21, 119-150.
- Antoniou, A., Guney, Y., Paudyal, K. (2008). The Determinants of Capital Structure: Capital Market Oriented Versus Bank Oriented Institutions. *Journal of Financial and Quantitative Analysis*, 43(1), 59-92.
- Asad, F., Gulzar, S., Bangassa, K., Khan, M. J. (2020). Capital Structure Adjustment and Market Reaction Following Seasoned Equity Offerings. *International Journal Of Finance&Economics*, 25(3), 388-411.
- Ashour, M., Atik, A. (2024). Earnings Management Around Seasoned Equity Offerings: A Systematic Literature Review. Panukkale University Journal of Business Research (PIAR), 11(1), 291 – 314.
- Babu, N. S., Chalam, V. G. (2014). Determinants Of Capital Structure Of Indian Textile Industry-An Empirical Analysis. *International Journal Of Advance Research*, 2(2) ISSN 2320-9127.
- Berger, P. G., Ofek, E., Yermack, D. (1997). Managerial Entrenchment and Capital Structure Decisions. *The Journal Of Finance*, 52(4), 1411-1438.
- Barclay, M. J., Fu, F., Smith, C. W. (2021). Seasoned Equity Offerings and Corporate Financial Management. *Journal Of Corporate Finance*, 66(1), 1-29.
- Booth, L., Aivazian, V., Kunt, A. D., Maksimovic, V. (2001). Capital Structures in Developing Countries. *The Journal of Finance*, 56(1), 87-130.
- Bradley, M., Jarrell, G. A., Kim, E. H. (1984). On The Existence Of An Optimal Capital Structure: Theory

Peer-review: Externally peer-reviewed.

Conflict of Interest: The authors have no conflict of interest to declare.

Grant Support: The author declared that this study has received no financial support.

Author Contributions: Conception/Design of study: A.K., Z.K.; Data Acquisition: Z.K.; Data Analysis/Interpretation: A.K., Z.K.; Drafting Manuscript: Z.K.; Critical Revision of Manuscript: A.K.; Final Approval and Accountability: A.K., Z.K.

and Evidence. The Journal of Finance, 39(3), 857-878.

- Brealey, R., Myers, S. C., Allen, F. (2017). Principles of Corporate Finance. Mc Graw Hill Education, New York.
- Chen, Y. C., Liu, J. T. (2022). Seasoned Equity Offerings, Return of Capital and Agency Problem: Empirical Evidence From Taiwan. Asia Pacific Management Review, 27(2), 92-105.
- Cortez, M. A., Susanto, S. (2012). The Determinants Of Corporate Capital Structure: Evidence From Japanese Manufacturing Companies. *Journal Of International Business Research*, 11(3), 121-134.
- Deesomsak, R., Paudyal, K., Pescetto, G. (2004). The Determinants Of Capital Structure: Evidence From The Asia Pacific Region. *Journal Of Multinational Financial Management*, 14(4-5), 387-405.
- Drobetz, W., Fix, R. (2003). What Are The Determinants of The Capital Structure? Some Evidence For Switzerland. University of Basel, WWZ/ Department of Finance, Working Paper 4(03).
- Fama, E., French, K. R. (2002). Testing Trade-Off and Pecking Order Predictions About Dividends and Debt. The Review of Financial Studies, 15(1), 1-33.
- Fan, J. P. H., Twite, G. J., Titman, S. (2012). An International Comparison of Capital Structure and Debt Maturity Choices. *Journal of Financial and Quantitative Analysis*, 47(1), 23-56.
- Fu, F., Smith, C. W. (2021). Strategic Financial Management: Lessons From Seasoned Equity Offerings. Journal Of Applied Corporate Finance, 33(1), 22-35.
- Frank, M. Z., Goyal, V. K. (2003). Testing The Pecking Order Theory of Capital Structure. Journal of Financial Economics, 67(2), 217-248.
- Frank, M., Goyal, V. K. (2008). Trade-off and Pecking Order Theories of Debt, Espen Eckbo (Ed). Handbook of Empirical Corporate Finance, North-Holland, 135-197.
- Friend, I., Lang, L. H. P. (1988). An Emprical Test of The Impact of Managerial Self-Interest on Corporate Capital Structure. *The Journal of Finance*, 43(2), 271-288.
- Gad, M., Hass, L. H., Marangoni, C. (2024). Seasoned Equity Offerings and Firm Workplace Safety. S&P Global Market Intelligence. Available at SSRN: https://ssrn.com/abstract=4759539 or http://dx.doi. org/10.2139/ssrn.4759539
- Gaud, P., Jani, E., Hoesli, M., Bender, A. (2005). The Capital Structure Of Swiss Companies: An Empirical Analysis Using Dynamic Panel Data. *European Financial Management*, 11(1), 51-69.
- Harris, M., Raviv, A. (1990). Capital Structure and The Informational Role Of Debt. *The Journal Of Finan*ce, 45(2), 321-349.
- He, L., Li, H., Liu, H., Vu, T. N. (2024). Why Do Banks Issue Equity? Research in International Business and Finance, 69, 1-23.
- Hernholm, P., Wormsen, A. O. (2023). Are Seasoned Equity Offerings Predictable? Norwegian School of Economics Master Thesis in Economic and Business Administration Major in Financial Economics, 1-60.
- Hirota, S. (1999). Are Corporate Financing Decisions Different In Japan? An empirical Study On Capital Structure. Journal of The Japanese and International Economies, 13(3), 201-229.
- Ho, C. Y., Wu, E., Yu, J. (2024). The Price of Corporate Social Irresponsibility in Seasoned Equity Offerings: International Evidence, *The British Accounting Review*, 56(4), 1-19.
- Huang, G., Song, F. M. (2006). The Determinants Of Capital Structure: Evidence From China. China Economic Review, 17(1), 14-36.
- Iqbal, A., Akbar, S., Shiwakoti, R. K. (2014). The Long Run Performance Of UK Firms Making Multiple

Rights Issues. International Review Of Financial Analysis, 28, 156-165.

- Jensen, M. C., Meckling, W. H. (1976). Theory of The Firm: Managerial Behaviour, Agency Costs and Capital Structure. *Journal of Financial Economics*, 3(4), 305-360.
- Jung, K., Kim, Y. C., Stulz, R. M. (1996). Timing, Investment Opportunities, Managerial Discretion, and The Security Issue Decision. *Journal of Financial Economics*, 42(2), 159–185.
- Le, L. H., Nguyen, T. B. N., Pham, X. Q. (2020). What Influences Decision On Seasoned Equity Offerings Of Listed Vietnamese Companies? *The Journal Of Asian Finance, Economics and Business*, 7(5), 1-7.
- Li, Z., Wang, P. (2021). Flotation Costs Of Seasoned Equity Offerings: Does Corporate Social Responsibility Matter? *European Financial Management*, 1-38.
- Marsh, P. (1982). The Choice Between Equity and Debt: An Empirical Study. *The Journal of Finance*, 37(1), 121-144.
- Mazur, K. (2007). The Determinants Of Capital Structure Choice: Evidence From Polish Companies. International Advances In Economic Research, 13, 495-514.
- Megginson, W. L. (1997). Corporate Finance Theory. Addison-Wesley, 1. baskı.
- Mehran, H. (1992). Executive Incentive Plans, Corporate Control and Capital Structure. *The Journal Of Financial and Quantitative Analysis*, 27(4), 539-560.
- Mehrotra, V., Mikkelson, W.H., Partch, M.M. (2005). Do Managers Have Capital Structure Targets? Evidence From Corporate Spinoffs. *Journal of Applied Corporate Finance*, 17(1), 18-26.
- Modigliani, F., Miller, M. H. (1958). The Cost of Capital, Corporation Finance and The Theory of Investment. *The American Economic Review*, 48(3), 261-297.
- Pesaran, M. H. (2004). General Diagnostic Tests For Cross Section Dependence In Panels. *Institute for the Study of Labor (IZA) Discussion Paper*, 1240.
- Rajan, R. G., Zingales, L. (1995). What Do We Know About Capital Structure? Some Evidence From International Data. *The Journal of Finance*, 50(5), 1421-1460.
- Ross, S. A. (1977). The Determination of Financial Structure: The Incentive Signalling Approach. *The Bell Journal of Economics*, 8(1), 23-40.
- Sayılgan, G., Sayman, Y. (2012). The Effect of Partnership Structures of Manufacturing Companies on Financing Decisions: Istanbul Stock Exchange Application (1998-2009). , 12(48), *Istanbul Stock Exchange*, 1-16.
- Sanchez, M. G., Juan, A. I. S. S., Jimenez, E. M. I. (2023). Earnings Management in Socially Responsible Firms Around Seasoned Equity Offerings: Evidence from France, Germany, Italy and Spain. *Heliyon*, 9.
- Scott, H. J. (1979). Bankruptcy, Secured Debt and Optimal Capital Structure. *The Journal of Finance*, 34(1), 253-260.
- Sheikh, N. A., Wang, Z. (2011). Determinants Of Capital Structure: An Empirical Study Of Firms In Manufacturing Industry Of Pakistan. *Managerial Finance*, 37(2), 117-133.
- Shenoy, C., Koch, P. D. (1996). The Firm's Leverage-Cash Flow Relationship. *Journal of Empirical Finance*, 2(4), 307-331.
- Stamou, S. C., Huang, W., Coakley, J. (2020). Serial SEOs and Capital Structure. International Review Of Financial Analysis, 71, 1-15.
- Stulz, R. (1988). Managerial Control Of Voting Right: Financial Policies and The Market For Corporate Control. Journal Of Financial Economics, 20, 25-54.

- Titman, S., Wessels, R. (1988). The Determinants Of Capital Structure Choice. *The Journal Of Finance*, 43(1), 1-19.
- Walker, D. M., Yost, K., Zhao, J. (2016). Credibility and Multiple SEOs: What Happens When Firms Return To The Capital Market? *Financial Management*, 45(3), 675-703.
- Walker, D. M., Wu, Q. (2019). Equity Issues When In Distress. *European Financial Management*, 25(3), 489 -519.
- Zhu, D., Li, Y. (2022). Empirical Research on Seasoned Equity Offerings, Board Member Characteristics and Corporate Investment Efficiency. *Discrete Dynamics in Nature and Society*, 22, Article ID 3365840, Hindawi.



Istanbul Business Research, 53(2), 229-248

DOI: 10.26650/ibr.2024.53.1232448 http://ibr.istanbul.edu.tr/ http://dergipark.org.tr/ibr

Istanbul Business Research

Submitted: 12.12.2022 Revision Requested: 28.08.2024 Last Revision Received: 30.08.2024 Accepted: 31.08.2024

RESEARCH ARTICLE

The Impact of Fintech on Economic Growth: Evidence from Panels of Turkic and Southeast Asian Countries

Aydın Gürbüz¹ (), Meltem Kılıç² (), Emre Tatlıyer³ ()

Abstract

From the beginning of the 21. century, the whole world has witnessed huge changes because of technological improvements and revolutionary inventions. It can be inferred that technological developments affect mankind in terms of many aspects like cultural, psychological, social etc. and these impacts reflect on our lives much more immediately than in past times. One of these impacts can be observed in economic and financial platforms. Financial technology has recently attracted great interest. The cause of this situation may be said to be Fintech becoming a global phenomenon. Therefore, many researchers have tended to investigate the impact of technological improvements on economic growth. This study attempts to evaluate the impact of financial technological improvements on economic growth between 2000 and 2020 for Turkic and selected Asian countries. In obtaining results, panel data analysis has been used. The results indicate that internet usage rates and fixed broadband subscriptions have positive impacts on economic growth. However, mobile cellular subscriptions do not have any impact on economic growth. According to the findings, information-communication technologies and other macroeconomic factors positively affect economic growth.

Keywords

Turkic countries and selected Asian countries, Panel data analysis, Fintech, Economic growth, Communication technologies

Introduction

From the beginning of the millenium era, dazzling developments have been seen in almost every aspect of human life. It does not seem like wrong to claim that these developments have been observed, especially in communication and information technology fields. Because of technological improvements and revolutionary inventions, the whole world has witnessed huge changes and a paradigm shift. Communication and information technologies simplify information circulation and thus have direct effects on humankind in terms of many aspects like cultural, economical, psychological and social etc. (Kurniawati, 2021; Badwan & Awad, 2022).



¹ Aydın Gürbüz (PhD Student), Kahramanmaraş Sütçü İmam University, Department of Business Administration, Kahramanmaraş, Turkiye. E-mail: aydingurbuz46@gmail.com ORCID: 0000-0002-2428-2327

² Corresponding Author: Meltem Kılıç (Assoc. Prof. Dr.), Kahramanmaraş Sütçü İmam University, Department of International Trade and Logistics, Kahramanmaraş, Turkiye. E-mail: meltem.kilic@hotmail.com ORCID: 0000-0001-8978-9076

³ Emre Tatlıyer (Lecturer), Hatay Mustafa Kemal University, Antakya Vocational School, Department of Accounting and Taxation, Hatay, Turkiye. E-mail: emretatliyer@gmail.com ORCID: 0000-0001-6863-2520

To cite this article: Gurbuz, A., Kilic, M., & Taliyer, E. (2024). The impact of fintech on economic growth: evidence from panels of turkic and Southeast Asian countries. *Istanbul Business Research*, 53(2), 229-248. http://doi.org/10.26650/ibr.2024.53.1232448

If it is described, it can be stated that "*Fintech*" is a branch of service sector that benefits from mobile-based information technology to increase the efficiency of financial systems. Another definition that might be verbalised is "*internet finance*" or "*digital finance*", which comprises all shapes of financial transactions/operations, markets and intermediaries. When we look at the etymological origins, as a term, "*Fintech*" is a composition of "finance" and "technology" words and corresponds to industrial innovation that emerged from the combination of financial services and information technology (Schueffel, 2016). With respect to the contribution of economic growth and value creation; it may be seen that developments in the field of financial technology and communication-information technology, such as electronic banking, e-commerce, mobile devices, wireless networks, Internet of things, artificial intelligence, cloud computing, and blockchain, have enormous and revolutionary impacts recently (Sassi & Goaied, 2013; Song & Appiah-Otoo, 2022; Awais et al., 2023). It was not thought wrong to claim that; Fintech will give shape business processes/models, subvert conventional financial operations/products, and bring a brand new breath by providing a unique business perspective in the millennium (Bu et al., 2022; Awais et al. 2023).

Fintech and communication-information technologies (CIT) augment the accessibility of information, shape new communication methods, and, as a result, increase the efficiency of many distinct financial and economic activities (Kurniawati, 2021). In terms of innovation perspective, CIT can be seen as a broad innovation type of financial service and technological progress, combining technological-product-resource allocation innovations. Through the Fintech, both information transparency and effective distribution of resources have increased however information asymmetry that causes improper market functioning has been mitigated. Fintech may also facilitate turning savings into investment and effective capital use. Therefore, it may be claimed that economic growth of countries or households has effected in an affirmative direction via these technologies (Bu et al., 2022).

Due to rapid developments in communication and information technologies, the number of Internet users continues to grow worldwide. According to data from the International Telecommunication Union, as of 2022, nearly 5.3 billion people - approximately 66% of the world's population - have been using the Internet. This indicates a 24% increase since 2019 and indicates that 1.1 billion people were predicted to have come online during that period, as well (International Telecommunication Union, 2023). According to data of Brimco; market size of global digital transactions was \$5.44 trillion in 2020, and by year 2026, it was predicted to be worth \$11.29 trillion. It is also predicted that the Fintech industry will reach \$ 11.8 billion (by year of 2018) to approximately \$306 billion (by year of 2023). In 2021, more than a hundred million people benefited from the opportunity of "*proximity mobile payment*". According to Getapp, proximity mobile payments can be described as a type of payment method for goods or services, which a customer uses (via Near Field Communication-NFC technology) his/her mobile phone or other smart device. According to the same data of Brimco, in

year of 2020, 90% of smartphone users involved in a mobile payment and mobile payment market ran up to a value of more than \$1 trillion all over the world. It is projected that mobile transactions will continue to grow, and 60% of customers look to transact or communicate with their financial institutions via a single platform, for example, a mobile application. Additionally, since 2019, the rate of cash usage at all sales points has decreased by 42%.

Nowadays, in which we stand on the edge of 4th industrial revolution, Fintech makes it easier and cheaper for people to access financial services, via computer/Internet technologies and mobile phone applications, in comparison to traditional financial service types. Thanks to these technologies, more people can enter money and capital markets, and consequently economic-financial growth has increased globally (Shkarlet et al., 2018; Shen et al., 2021).

We believe that the Turkic countries of Turkey, Azerbaijan, Kazakhstan, Kyrgyzstan, Turkmenistan, and Uzbekistan, with their common cultural and historical ties, can form a meaningful group to compare the effects of Fintech on economic growth. These similarities allow for a consistent analysis of Fintech's effects on economic growth. In addition, countries such as China and India, which are among the world's largest economies, and Indonesia, Pakistan, Tajikistan, and Russia, which are emerging markets and have a large geographical and economic impact, provide a suitable basis for assessing the role of Fintech in economic growth on a global scale. China, in particular, a leader in mobile payment systems and digital banking, while in regions such as India and Indonesia, Fintech is seen as a critical tool for bringing financial services to the unbanked population through digital platforms. The impact of Fintech on economic growth can be understood through many factors, such as organisational innovations through the development of Information and Communication Technologies (ICT), productivity growth, efficiency of market transactions, new business opportunities, social and cultural changes, and education and knowledge management innovations. These factors allow us to assess Fintech's contribution to economic growth from a broad perspective. For this reason, we selected the Turkic countries and strategically and economically important countries such as China, India, Indonesia, Pakistan, Tajikistan, and Russia. Moreover, studies by Omer (2021) and Razzag (2024) revealed the importance of Fintech in these regions. In short, Fintech is thought to increase economic prosperity in these countries that are endowed with many resources.

In this context, Fintech's worldwide enourmous progress has strengthened our hands. Therefore, we think that in 2000-2020 period, a nexus between Fintech usage of the Turkic World countries and selected some Asian countries and economic growth worths researching on it for our aforementioned study. The rest of our study has been organised as; in the second part, existing literature, in the third part, variables and data set used in the study, and in the fourth part, findings of the research have been mentioned. In the fifth and last part, the results were evaluated and provided some advices.

Literature Review

Although the impact of the Fintech sector, which is named as communication and information technology, on economic growth has been accepted for many years, this impact has been felt much more seriously on a global scale in recent years. It is stated that the important elements that have made Fintech popular late are; technological progresses, changing customer demands and macroeconomic effects. Solow, (1956), In the study, also recognised as the "Solow Theory", Solow indicates that promoting technological developments will enhance income in the current labour market, and with that, innovations created by the real sector due to technological progress will have a significant impact on economic growth. On the other hand, Romer (1990) stated that technological change in the growth model is endogenous, and technological improvements guide investment decisions. Therefore, he asserted that technological improvements will direct economic growth. Bai and Ding (1998), held the view that technological innovations impact financial development in a positive manner and support economic growth. Pohjola (2001) also found that, given the importance of both human and physical capital and similar economic factors, both developing and developed countries will need information technology to promote economic growth. Jorgenson and Vu (2005), examined the influence of investments in information technology on economic growth in the world's 14 largest economies between 1989 and 2003. From that study, they find that technological investments greatly affect economic growth. Additionally, in their study, they determined that this effect was more prominent in the G7 countries and that developing Asian countries followed the G7 countries.

The vast majority of the studies in the literature show that the impacts of financial technologies (Fintech) on economic growth have a positive direction (Madden & Scott, 2000; Torero et al., 2002; Nasab & Aghaei, 2009; Yapraklı & Sağlam, 2010; Anthony & Patrick, 2013; Ishida, 2015; Salahuddin & Gow, 2016; Khan et al., 2020; Cumming & Schwienbacher, 2021). According to the research conducted by Garces and Daim (2012) in the United States, communication and information technology has an affirmative effect on economic growth. In the study of Yılmaz & Kırışkan (2017), conducted in Türkiye, it's claimed that communication and information technology is a crucial element for economic growth. Sadigov et al., (2020) stated that, advances in Fintech support economic growth by increasing real sector financing. According to Kurniawati (2021), communication and information technologies have a significant effect on economic development in high income level and middle income level Asian countries. Kurniawati also stated that improving CIT infrastructure in middle-income Asian countries is necessary to further expedite economic growth. Song and Appiah-Otoo (2022), clearly indicate that Fintech is a basic determinant of China's economic growth. Badwan and Awad (2022) stated that the growing use of financial technology in Palestine contributes to economic growth. Naeruz et al. (2022), also specified that financial technological advances have an affirmative effect on Indonesia's economic growth. Between 1995 and 2008, Türedi (2013) attempted to determine the effect of information and communication technology on economic progress in 53 countries, 23 developed and 30 developing. The study in which the sample was quite wide, Türedi benefited from the panel data analysis. The result of the empirical analysis indicates that the use of CIT in both developed and developing countries has a positive impact on economic growth. Erumban and Das (2016) targeted to reveal the role of CIT in routing economic growth in India between 1986 and 2011. In their studies, they divided the usage areas of CIT into manufacturing and service sectors. According to the results, it can be seen that the use of CIT in India enhances economic efficiency. Another important finding in the aforementioned study is that the weight of CIT used in the service sector is higher than that used in other sectors. Toader et al. (2018) discovered, that the prevalence of CIT in European Union countries between 2000 and 2017 is an important factor in economic growth. Romdhane et al. (2020), between 2001 and 2017, examined the relationship between Fintech usage and economic boost for 21 African countries. Findings demonstrate that financial and technological developments stimulate growth in African countries. Nair et al. (2020) examined the impact of CIT infrastructure on economic growth in OECD countries. According to the empirical findings, the progress of CIT makes a long-term contribution to economic growth. Kirayeva et al. (2021) evaluated the impact of financial technologies on economic growth in the period of 2004-2019. The country groups in their research comprise selected countries from the EU, CIS (Commonwealth of Independent States) and economies in transition. As a consequence of their research, they documented a significantly increased usage of Fintech in the countries examined. In the examined period, additionally, with the transformation in financial technology, researchers determined that there was downsizing in physical (numbers of bank branches) fields by transferring the financial transactions of the countries to the digital environment. As a result, they verified that a strong link exists between financial technologies and economic progress.

Bu et al. (2022) evaluated the effect of Fintech improvements in China on real economic progress using panel data analysis. Due to this study, it has been demonstrated that Fintech usage has crucial and encouraging dimensions to economic growth. Utami (2022) analysed Fintech's impact on economic growth in Indonesia between the years 2020-2022. The empirical results indicate that Fintech has a significant affirmative impact on economic development. In the research carried out by Gomes et al. (2022), the impact of Fintech on economic growth in OECD countries tried to be determined. In this research, 36 OECD countries were divided into groups according to their level of development between 2000 and 2019, and the GMM panel method was applied. The result of that study shows that Fintech usage has a significant effect on economic growth in OECD countries. In this respect, the cruciality of Fintech usage was emphasised to policy makers in terms of economic revival.

Even though it has been confirmed in the literature that the connection between Fintech and economic growth is affirmative, there are some studies that also stated the opposite or that there is no relationship between them. For example, Li et al. (2009) examined the nexus between fintech and economic growth in the Chinese economy. But, according to the results of their study, they could not identify any connection between Fintech and economic growth. Nabi et al. (2022) aimed to reveal the impact of CIT usage on economic growth for N11 (Next Eleven) countries in the 2000-2018 period. It has been identified that there is a negative correlation between the increasing use of CIT in N11 countries and economic growth in the examined period.

Although our study focuses on the nexus between Fintech and economic growth, we think that the impact of control variables, as well, in the model on economic growth need not be neglected. Studies in the literature claim that economic growth is positively affected by gross fixed capital formation (GFCF) (Ali, 2017; Boamah et al. 2018; Pasara & Garidzirai, 2020), general government final consumption expenditure (GOV) (Gomes et al. 2022; Poku et al. 2022), financial development (FG) (Salahuddin et al. 2016; Gow, Kurniawati, 2022).

This study focuses on the nexus between Fintech usage and economic growth in the Turkic and selected Asian countries. In the literature, it is seen that in most studies examining the linkage between Fintech and economic growth, a single criterion is used as the Fintech indicator. On the other hand, limited number of studies have focused on mobile cellular subscriptions, fixed broadband subscriptions and internet usage rate measurements as Fintech indicators. In addition, existing studies in the literature neglect the causal relationship between Fintech and economic progress. Finally, almost no studies have evaluated the impact of Fintech usage on economic growth from a wide perspective in countries in the Turkic world. Thus, our study not only enrichs the existing literature but also contributes to the literature by investigating the impacts of various Fintech indicators on economic growth for countries in the Turkic world and selected Asian countries.

In his neoclassical growth theory, Solow (1956) argued that information and communication technologies (ICTs) promote economic growth through technological advances and improved labour quality. However, Solow considered these technological developments as exogenous factors. Recent empirical studies, on the other hand, have examined the effects of the internalisation of technological innovations on economic growth, and these findings have been associated with endogenous growth theory (Ehrlich, 1990:9). In this context, the impact of Fintech, which is an important part of ICT, on economic growth can also be explained by endogenous growth theory. Therefore, researchers generally emphasised that ICT has a significant impact on the economic growth of countries (Choi and Yi, 2009; Andrianaivo and Kpodar, 2011; Farhadi, et al. 2012).

In this direction, we aim to test the impact of Fintech on the economies of the Turkic countries and some selected Asian countries in our study, based on the following hypotheses that we have developed inspired by various studies in the literature (Bahrini and Qaffas, 2019; Mayer, et al. 2020) that examined the effects of Fintech on economic growth: H₁: Mobile cellular subscriptions have an affirmative impact on the economic growth of Turkic and selected Asian countries.

H₂: Fixed broadband subscriptions have an affirmative impact on the economic growth of Turkic and selected Asian countries.

H₃: Internet usage rates have an affirmative impact on the economic growth of Turkic and selected Asian countries.

Data Explanation

Information and communication technologies are expressed using monetary and nonmonetary indicators. While monetary indicators comprise information and communication technology investments, non-monetary indicators are measured using numerical information such as mobile cellular subscriptions and fixed broadband subscriptions (Hu et al. 2021; Iqbal et al. 2022).

The aim of this research is to evaluate the effect of Fintech, which refers to information and communication technology, on the economic growth of countries in the Turkic world and some Asian countries between 2000 and 2020. Within the scope of the study, country data from Azerbaijan, China, Indonesia, India, Kazakhstan, Kyrgyzstan, Uzbekistan, Pakistan, Russia, Türkiye and Tajikistan were collected. Because Fintech data of Turkmenistan, a Turkic country, could not be accessed, this country was excluded from the sample. In addition, the reason why the study ended in 2020 is that, Fintech data of the countries within the scope of the study were published until this year. In the context of non-monetary indicators of CIT as a Fintech criterion; mobile cellular subscription (per 100 people) (LMOB), Internet use (% of Population) (LINDI), and fixed broadband subscriptions (per 100 people) (LFBROAD) were used. The dependent variable of the research was the gross domestic product per capita (Fixed 2015 US\$) (LGDP). The financial development index (LFG), gross fixed capital formation GDP (LGFCF) and general government final consumption expenditures (% of GDP) (LGOV) are also involved in the model as control variables. Among the data used in this study, the financial development index is provided by the International Monetary Fund's (IMF) database at www.data.imf.org, while other data are provided by the World Bank's database Data Bank's official website www.data. worldbank.org.

Model Specifications

For examining the impact of fintech on economic growth; the below defined econometric model was developed by benefiting the studies of Hafru, 2019; Adeleye & Eboagu, 2019; Kurniawati, 2022; Iqbal et al. 2022; and Gomes et al. 2022:

$$LGDP_{it} = \beta_0 + \beta_1 * LMOB_{it} + \beta_2 * LINDI_{it} + \beta_3 * LFBROAD_{it} + \beta_4 * LGFCF_{it} + \beta_5 * LGOV_{it} + \beta_6 * LFG_{it} + u_{it}$$
(1)

In this model, i denotes the country, t denotes the time, β denotes the coefficient, and u_it denotes the error term. Additionally, the symbol L in the model indicates that the variables are logarithmic series.

Methodology

Panel data regression and panel causality analyses were used to examine the connection between information and communication technologies (Fintech) and economic development in the Turkic world and selected Asian countries between 2000 and 2020. In order to perform panel data regression analysis and causality analysis, we first checked whether the series contained cross-sectional dependence and whether they were stationary. Cross-sectional dependence analysis of the series (Pesaran, 2004) was examined using the CD test. Whether there is cross sectional dependence in the CD (*Pesaran, General Diagnostic*) test, is tested with "*H0: There is no cross-sectional dependence* ($H_0:\rho_{ij}=0$)" (Tatoğlu, 2018). If the series include cross-sectional dependence, the stationarity of the variables is analysed using the second-generation panel unit root test and the cross-sectionally augmented LM, Pesaran and Shin (CIPS). The CIPS test designed by Pesaran is a simple alternative method expanded by standard augmented Dickey-Fuller (ADF) regressions with cross-sectional means of lagged levels and first differences of individual series (Pesaran, 2007). The CIPS stationarity test is applied as "H0: Series is not stationary (Contains unit root)." The cross sectional dependence and panel unit root test results are presented in Tables 3 and 4.

In static panel data regression analysis, pooled ECT, fixed effect and random effect models are frequently used. To select among these estimators; F Test which was developed by Moulton and Randolph, Breush-Pagan Lagrange Multiplier (LM) Test and Hausman Test were performed. The null hypothesis of this test is " H_0 : All unit and time effects are equal to zero ($H_0:\mu_i=\lambda_t=0$)." (Tatoğlu, 2020). Using the F Test, a decision is made between the pooled least squares and fixed effects models. The Breush-Pagan LM test is estimated with the hypothesis applied as parametric constraints. Using the LM Test, the regression is calculated by using the residuals of the model estimated by using the least squares test (Breush and Pagan, 1980).

In the Breush-Pagan LM test, hypothesis testing is performed in the form of " H_0 : *The variance of unit effects is zero* (H_0 : $\sigma_\mu^2=0$)." (Tatoğlu, 2020). Breush-Pagan LM test is made for deciding between the pooled least squares and random effects models. The Hausman test

is designed under the null hypothesis of misidentification and is developed for model specification. This test is performed to test the instrument variable, time section model and simultaneous equation model (Hausman 1978: 1251). The null hypothesis of this test is " H_0 : Difference between parameters is not systematical." (Tatoğlu, 2020). The Hausman test was used to select the appropriate model from the fixed-effects and random-effects models. These three estimator tests are presented in Table 5. After deciding on the regression model according to the results of the Hausman test, F test and Breush-Pagan test, assumption tests are run to determine whether there is a heteroscedasticity, autocorrelation and cross-sectional dependence problem in the model.

The main assumption tests in panel data models are the Levene, Brown and Forsythe Multiplier Test, Durbin-Watson and Baltagi-Wu local best invariant (LBI) tests and the Frees Test. The Levene, Brown and Forsythe tests are derived to test the inequality of variances. In other words, it tests the presence of heteroscedasticity. The null hypothesis of this test is established as " H_0 : The variances of the units are equal.". The tests applied to the auto-correlation problem are Durbin-Watson and Baltagi-Wu LBI tests. In these tests, a critical value close to 2 indicates that there is no auto-correlation problem. The test hypothesis is established as " H_0 : There is no auto-correlation." (Tatoğlu, 2020). Frees developed the test that examines the existence of a cross-sectional dependence. The model tests the hypothesis that there is no cross-sectional dependence (Frees, 1995). The results of the assumption tests are presented in Table 6.

According to the results of the panel data assumption tests, we found that the model exhibits heteroskedasticity, autocorrelation and cross-sectional dependence problems. Taking these three problems into account, a model estimation was made using the Driscoll-Kraay Robust Estimator. Driscoll-Kraay Test, which is a simple extension of the common non-parametric covariance matrix estimation technique, yields standard error estimates that are robust to all general forms of spatial and temporal dependence as the time dimension grows (Driscoll-Kraay 1998). In this study, the regression created by robust estimation is described in Model (1) and the results are displayed in Table 7.

After determining the impact of Fintech indicators on economic growth, the relation between the dependent variable and independent variables is evaluated with Dumitrescu-Hurlin Panel Causality Test. Dumitrescu-Hurlin's hypothesis testing is established as " H_0 : X is not the cause of Y." (Dumitrescu & Hurlin, 2012). The panel causality test results are summarised in Table 8.

Findings

The descriptive statistical analysis of the series used for the sample to determine the impact of Fintech on economic growth in the Turkic and selected Asian countries is displayed in Table 1. The results of the correlation analysis exhibiting the multicollinearity between dependent and independent variables are presented in Table 2.

Descriptive St	unsues				
Variables	Observation	Mean	Standart Deviation	Minimum	Maximum
LGDP	231	7.9219	0.9334	6.0645	9.3986
LMOB	231	3.5620	1.7393	-3.9904	5.1749
LINDI	231	2.5101	1.5589	-3.0241	4.4536
LBROAD	231	0.6851	2.9959	-8.8057	3.5244
LGFCF	231	3.2770	0.3159	2.2123	4.0602
LGOV	231	2.5348	0.2611	1.8767	3.0343
LFD	231	1.3716	0.6429	-2.8739	-0.3977

Table 1 Descriptive Statistics

In this study, there are 231 observations between 2000 and 2020. The variable with the highest average is LGDP at a rate of 7.92%, while the lowest is LBROAD at a rate of 0.68%. On the other hand, LBROAD had the highest standard deviation 2.99%, while LGOV had the lowest standard deviation 0.26%. The variable with the lowest minimum value is LBROAD and the variable with the highest minimum value is LGDP. While LGDP has the highest maximum value, LFD has the lowest maximum value. In addition, when having looked at the Fintech sub-metrics, LMOB variable has the highest average (3.56%) and LBROAD variable has the lowest average (0.68%). After explaining the descriptive statistics of the variables used in the model, the relationships between the variables were tested using Spearman's rank correlation test. Spearman's rank correlation test examines the relationship between two nonuniform heterogeneous series (Spearman, 1904). The results of this test are summarised in Table 2.

The Results of the Spearman's Rank Correlation Analysis							
	LGDP	LMOB	LINDI	LBROAD	LGFC	LGOV	LFD
LGDP	1.0000						
LMOB	0.7029*	1.0000					
LINDI	0.7502*	0.8873*	1.0000				
LBROAD	0.6832*	0.3431*	0.3933*	1.0000			
LGFCF	0.1724*	0.32483*	0.2259*	0.1231	1.0000		
LGOV	0.1304**	0.1700*	0.2522*	0.4365*	0.0936	1.0000	
LFD	0.6469*	0.2506*	0.3185*	0.3824*	0.2980*	0.1062	1.0000

, n 1 a c 1 . C . .

Note: * and ** indicate significance levels of 1% and 5%, respectively

To the result of the correlation analysis, there was a high correlation between LGDP and LMOB (0.7027 for a p<1%). There was a high correlation between LGDP and LINDI (0.7502 for a p < 1%) and between LGDP and LBROAD (0.6832 for a p < 1%). There are statistically significant but weak correlations between LGDP, LGFCF and LGOV. There is also a high correlation between LGDP and LFD at 0.6469. There was a very high correlation between LMOB and LINDI (0.8873 for a p<1%). There is a weak correlation between LMOB and LBROAD, LGFCF and LFD, whereas the correlation between LMOB and LGOV is very weak. The correlation coefficients between LINDI and LBROAD, LGFCF, LGOV and

Table 2
LFD indicate a weak relationship. There was a moderate correlation between LBROAD and LGOV (0.4365 for a p<1%). There was a weak correlation between LBROAD and LGFCF with LFD.

Following the correlation and descriptive statistics analyses, a cross-sectional dependence analysis and panel unit root tests used in series are performed. Pesaran CD cross-sectional dependance test results are summarized in Table 3.

	CD-Test	Prob.
LGDP	32.41*	0.000
LMOB	33.23*	0.000
LINDI	32.19*	0.000
LBROAD	30.53*	0.000
LGFCF	7.19*	0.000
LGOV	3.77*	0.000
LFD	12.55*	0.000

Note: * indicates significance at 1%

According to the result of the Pesaran CD test, it was detected that both dependent and independent variables consist of cross-sectional dependence. In other words, the null hypothesis was rejected and it was concluded that a cross-sectional dependence exists. In order to determine whether the series with cross-sectional dependence are stationary, the cross-sectional augmented panel unit root IPS (CIPS) test developed by Pesaran (2007), taking into account cross-sectional dependence, is used as the second generation panel unit root test.

Results of the	CIPS	(Intercept)	Unit Roo	t Test

Table 4

Variables	Level	Δ
LGDP	-2.178***	-2.350**
LMOB	-3.175*	-3.484*
LINDI	-2.300**	-3.921*
LBROAD	-3.094*	-2.707*
LGFCF	-2.766*	-3.013*
LGOV	-2.386*	-3.231*
LFD	-2.171***	-3.112*

Note: *, **, and *** indicate significance levels of 1%, 5%, and 10%, respectively.

Critical values at 1%, 5%, 10% significance are as follows: 2.45, 2.25, 2.14 (Pesaran, 2007).

To the results of the CIPS panel unit test, we determined that all of the series were stationary at different levels of significance. That is, it was concluded that all variables used in the study were I(0).

Since the series are stationary a level, it was decided to perform panel regression analysis in the study. The estimator tests are used to decide which panel regression analyses from the Pooled LS, Fixed-Effect Model and Random-Effect Model tests will be applied. The results of the panel data regression estimator tests are exhibited in Table 5.

Results of	of Panel Regression	n Estimator Tests				
	F Test		Breush-Pagan LM Test		Hausman Test	
	F. Statistical Value	Probability Value	Chibar2	Probability Value	Chi2	Probability Value
Model	818.35*	0.000	1971.74*	0.000	2.83	0.830
MT 4	1	1 610/				

Table 5

Results of Panel Regression Estimator Tests

Note: * indicates significance level of 1%

According to the results of the F test, the null hypothesis was rejected and it was determined that all unit and time effects were not equal to zero. This result shows that using the Pooled LS model is not suitable for the established model. According to the results of Breush-Pagan LM test, the hypothesis that the variance of unit effects is equal to zero is rejected. To the result of hypothesis testing using the LM test, it was decided that the Random-Effect Model must be preferred. According to the results of the Hausman test to decide between the Fixed-Effect Model and Random-Effect Model, the null hypothesis was accepted and it was inferred that the difference between the parameters was not equal. Additionally, it was decided that the most suitable test among the regression analyses was the Random-Effect Model estimator. In order to determine whether there is heteroskedasticity, autocorrelation and cross-sectional dependence in the Random-Effect Model assumption tests were performed.

Heteroskedasticity Test	eteroskedasticity lest							
	Test Statistics	Prob. Value						
W0	12.968*	0.000						
W50	10.250*	0.000						
W10	12.750*	0.000						
Cross-Sectional Dependance Test								
Frees	2.442**							
Results of Autocorrelation Test								
Modified from Bhargava et al. Durbin-Watson	0.228							
Baltagi-Wu LBI	0.373							

 Table 6

 Results of Panel Regression Assumption Tests

Note: * indicates significance level of 1%. ** Frees critical value alpha = 0.10:0.1231, 0.05:0.1611, 0.01: 0.2338

According to the consequences of the assumption tests, the null hypothesis for heteroscedasticity and cross-sectional dependence in the model was rejected. Based on the results obtained, it was decided that the variance of the units was not equal to zero and a correlation was observed between the units. Additionally, since Durbin-Watson and Baltagi-Wu LBI test values were far below 2, it was concluded that there was an autocorrelation problem in the model. From the results of the assumption test, we conclude that the model established comprises all three problems. In order to eliminate these problems, the model must be estimated using robust methods. The most suitable estimator, considering the heteroskedasticity, autocorrelation and cross-sectional dependance problems, is the Driscoll-Kraay Robust Estimator method. The coefficient results of the model established by considering the estimation method are explained in Table 7.

Table 7						
Results of the Random Effect Driscoll-Kraay Robust Estimator Test						
Coefficient Driscoll-Kraay Standard Error Probability Val						
LMOB	-0.0092	0.0179	0.613			
LINDI	0.1266*	0.0152	0.000			
LBROAD	0.0402*	0.0081	0.000			
LGFCF	-0.2925*	0.0493	0.000			
LGOV	-0.0579	0.0964	0.555			
LFD	0.2679*	0.0899	0.007			
CONSTANT	9.1379*	0.2752	0.000			
Wald Chi2	5857.77*					
Probability	0.0000					
R2	0.1259					

Note: *indicates significance level of 1%

Obtained Driscoll-Kraay robust estimator results demonstrate that the model is statistically significant. On the other hand, if we examine the impact of Fintech indicators on GDP we found that the impact of the LINDI variable on LGDP is positive and statistically significant at the 1% level. While this result shows that a 1% increase in internet usage rates increases the economic growth of the Turkic and selected Asian countries by 12%, it also confirms the hypothesis H_2 established in the hypothesis development section. Similarly, the effect of LBROAD on LGDP is positive and statistically significant at the 1% level. This result means that fixed broadband subscriptions have an increasing effect on the economic growth of the Turkic and selected Asian countries by approximately 4%, whereas the H_3 hypothesis established in the hypothesis established in the hypothesis development section will be accepted. On the other hand, unlike the existing studies, we reject the H_1 hypothesis established in the hypothesis development section and observe that mobile cellular subscriptions have no impact on the economic progress of the Turkic and selected Asian countries. Nevertheless, we think that it is not wrong to state that this situation does not undermine the economic consistency of our analysis.

In the model, after examining the nexus between Fintech indicators and economic growth, if we look at the impact of control variables on GDP, it has been revealed that the effect of gross fixed capital formation (LGFCF) on LGDP is negative and statistically significant, and financial development (LFD) has a positive effect on LGDP. Finally, we conclude that general government final consumption expenditures (LGOV) have no effect on LGDP. The discussion of the findings obtained within the scope of the study is presented in the conclusion section.

After examining the impact of Fintech indicators and control variables on economic growth in our study, we applied the Dumitrescu and Hurlin Panel causality test to examine

Table 8

the causal relationship between the variables. Table 8 presents the Dumitrescu and Hurlin Panel causal relationship results.

Results of the Dumitrescu and Hurlin Panel Causality Test						
	W-Stat	Zbar-Stat	Olasılık			
LMOB <=> LGDP	5.0827	3.3029	0.0010*			
	3.9289	1.9168	0.0553**			
LINDI <=> LGDP	4.7497	2.9028	0.0037*			
	9.9395	9.1374	0.0000*			
LGDP => LBROAD	2.8505	0.6213	0.5344			
	14.8341	15.0174	0.0000*			
	3.0137	0.8174	0.4137			
LGDP => LGFCF	6.7756	5.3366	0.0000*			
LCDD-> LCOV	3.0676	0.8821	0.3777			
LGDP => LGOV	7.0112	5.6196	0.0000*			
	1.2354	1.3189	0.1872			
LUDY -> LFD	6.0050	4.4108	0.0000*			

Note: *and ** indicate significance levels of 1% and 5% respectively

According to the causality findings, a bidirectional causal relationship between mobile cellular subscriptions (LMOB) and economic growth (LGDP) occurs. Likewise, there exists a bidirectional causal relationship between Internet usage rates (LINDI) and economic progress (LGDP). There is also a one-way causal relationship from economic progress (LGDP) to fixed broadband subscriptions (LBROAD). These results prove that a causal relation between Fintech and economic growth exists. On the other hand, our results show that, there's oneway causality from LGDP to LGFCF, LGDP to LGOV and LGDP to LFD.

Conclusions

This study is conducted in order to reveal the impacts of Fintech indicators, called information and communication technology, on the economic development of the Turkic countries and some selected Asian countries. In this direction, within the scope of the study, first of all the descriptive statistics of the series were evaluated and correlation analysis was performed. Then, the Pesaran CD test was applied to test the cross-sectional dependence of the series, and as a result of the test, it was specified that the series included cross-sectional dependence. The CIPS panel unit root test was then applied to determine the stationarity of the series in the model. To the results of the CIPS unit root test, it was determined that all series were stationary at level. In order to determine the appropriate panel data regression analysis for the stationary series, some estimator and assumption tests were performed. To the result of the estimator tests, it was determined that the most suitable model for the study was the Random Effect model. Due to the heteroskedasticity, autocorrelation and cross-sectional dependance problems arise in the Random Effects Regression Model, our model was tested using the Driscoll-Kraay Robust Estimator. According to the robust estimation results, internet usage rates and fixed broadband subscriptions have positive effects on the economies of the Turkic world and certain Asian countries. These findings in line with the results of recent studies indicating that increasing internet usage rates can positively affect economic growth. Salahuddin and Gow (2016) for South African countries, Gomes et al. (2022) for OECD countries Kurniawati (2022) for Asian countries found similar findings. In addition, there are studies suggesting that high levels of fixed broadband subscriptions positively impact economic growth. Toader et al. (2018) for European countries, Nair et al. (2020) for OECD countries, and Badwan and Awad (2022) for Palestine produced parallel results in this regard. In this context, according to the findings of both our study and other studies in the literature, improving the ICT skills of internet users, especially in developing countries, is critical in terms of supporting economic growth and development (Haftu, 2019). On the other hand, it is concluded that financial development, one of the control variables included in the model, has a positive effect on economic growth. This result is consistent with Pasara and Garidzirai (2020) and Nchake and Shuaibu (2022). However, it is observed that the effect of gross fixed capital investments, one of the control variables, on economic growth is negative. There may be several reasons for the negative impact of gross fixed capital investments on economic growth: Low capital formation resulting from slow growth in some countries in the sample of the study, differences in the amount of investment expenditures, inefficient use of investments or low productivity, overcapacity creation, high debt burden, global economic shocks and problems in investment quality may lead to these negative effects. Moreover, Collier and Gunning (1999) argued that one of the main reasons for poor growth performance is low investment.

In this study, causality connections between the variables were also examined with the Dumitrescu and Hurlin Test, and the results of the test showed that there was a bidirectional causality between internet usage rates and economic development. In addition, it was concluded that there's a bidirectional causality between mobile cellular subscriptions and economic development. These results indicate that internet usage rates and mobile cellular subscriptions will affect economic growth, which will affect internet usage rates and mobile cellular subscriptions. Finally, it's been determined that there's a one-way causal relationship from economic growth to fixed broadband subscriptions, financial development indicators, gross fixed capital investments and general government final consumption expenditures.

Recently, both the government and the real sector and scholars have shown great interest in information and communication technologies (Fintech). We believe that we have obtained some important findings by focusing on the Fintech sector in this study. According to these findings, information and communication technologies and other macroeconomic factors positively affect the economic growth of Turkic and selected Asian countries. In this respect, we think that new investments to be made in Fintech infrastructure and improvements to existing investments will be essential factors in guiding the economic development of the Turkic and selected Asian countries. On the other hand, Internet usage rates, which is one of the CIT indicators, make the biggest contribution to the economies of the Turkic and selected Asian countries, fixed broadband subscriptions make the lowest contribution. Therefore, these results allow policy makers to make important inferences. For example, it seems beneficial for countries to expand their CIT infrastructures and facilitate access to technology for the entire society, especially rural areas. The findings of this study confirm the interpretation that internet usage rates and fixed broadband subscriptions increase economic growth. From this perspective, we believe that improving the CIT infrastructure is of critical importance. Improvement of the CIT infrastructure will also contribute to the progress of the digital economy. Therefore, we believe that it will be beneficial for policymakers, the private sector and academics to act together and accelerate their studies in this field. From this point of view, incentives and projects should be developed by the Turkic and selected Asian countries to facilitate access to technology in rural areas through ICT investment incentive policies, and public awareness should be raised by organising training programmes on the use and benefits of ICT. In addition, cooperation between the public and private sectors should be established, joint projects and funds should be established for Fintech and ICT investments, and tax incentives should be provided to companies investing in these areas. Legal arrangements should be made for the rapid and secure growth of the Fintech sector and innovative projects should be encouraged by increasing R&D support. Finally, high-speed internet access can be expanded through policies to increase fixed broadband subscriptions.

This study examines the impact of CIT on economic growth in the Turkic and selected Asian countries. Given Fintech's global relevance, future research could explore the Fintechgrowth nexus among different country groups, especially between developed and developing nations. Sector-specific and city-based analyses, such as Fintech's effects on financial services, retail, health, and education, could reveal differences between metropolitan and rural areas and help to develop regional policies. Additionally, examining the impact of policies and regulations on the Fintech ecosystem can help governments to craft effective strategies. These research directions will provide a more comprehensive understanding of Fintech's impact on economic growth and will contribute significantly to the literature.

Grant Support: The authors declared that this study has received no financial support.

References

Peer-review: Externally peer-reviewed.

Conflict of Interest: The authors have no conflict of interest to declare.

Author Contributions: Conception/Design of study: A.G.; Data Acquisition: A.G., M.K.; Data Analysis/Interpretation: M.K., A.G., E.T.; Drafting Manuscript: M.K., A.G., E.T.; Critical Revision of Manuscript: E.T.; Final Approval and Accountability: M.K., A.G., E.T.

Adeleye, N., & Eboagu, C. (2019). Evaluation of ICT development and economic growth in Africa. Economic Research and Electronic Networking, 20(1), 31-53. https://doi.org/10.1007/s11066-019-09131-6

- Ali, G. (2017). Gross fixed capital formation and economic growth of Pakistan. Journal of Modern Developments in Social Sciences Research, 1(2), 25-34.
- Andrianaivo, M., & Kpodar, K. (2011). ICT, financial inclusion, and growth: Evidence from African countries. *IMF Working Papers*, WP/11/73, pp. 3-45.
- Anthony, U. E., & Patrice, N. (2013). The effects of information technology on global economy. *The Social Sciences*, 8(6), 606-609.
- Awais, M., Afzal, A., Firdevsi, S.& Hasnaoui, A. (2023). Is fintech the new path to sustainable resource utilisation and economic development? *Resources Policy*, 81.
- Badwan, N., & Awad A. (2022). The impact of financial technological advancement (FinTech) on the economic growth: Evidence from Palestine. Asian Journal of Economics, Business and Accounting, 22(23), 50-65.
- Bahrini, R., &Qaffas, A. A. (2019). Impact of information and communication technology on economic growth: Evidence from developing countries. *Economies*, 7(1). https://doi.org/10.3390/economies7010021
- Bai, Q., & Ding S. (1998). Sustainable development of finance. International Financial Research, 5, 28-32.
- Boamah, J., Adongo, F. A., Essieku, R., & Lewis, Jr. J. A. (2018). Financial depth, Gross fixed capital formation, and economic growth: Empirical analysis of 18 Asian Economies. *International Journal of Scientific* and Education Research, 2(4), 1-11.
- Breusch, T. S., & Pagan A. R. (1980). The lagrange multiplier test and its applications to model specification in econometrics. *The Review of Economic Studies*, 47(1), 239-253.
- Brimco. *Fintech*, (2022). Retrieved from. Retrieved from https://www.brimco.io/fintech-statistics-to-know/. (Data of Access: 01.02.2023)
- Bu, Y., Yu, X., & Li, H. (2022). The nonlinear impact of FinTech on the real economic growth: Evidence from China. *Economics of Innovation and New Technology*, pp. 1-18.
- Choi, C., & Yi, M. H. (2009). The effect of the internet on economic growth: Evidence from cross-country panel data. *Economics Letters*, 105(1), 39-41. https://doi.org/10.1016/j.econlet.2009.03.028
- Collier, P., & Gunning, J. W. (1999). Explaining African economic performance. Journal of Economic Literature, 37, 64–111. https://doi.org/10.1257/jel.37.1.64
- Conrad, A. (2021). Proximity mobile payment usage is the future of retail. June https://www.getapp.com/ resources/proximity-mobile-payment/. (Data of Access: 01.02.2023)
- Cumming, J. D., & Schwienbacher, A. (2021). FinTech Venture Capital. Thomas Liaw. (Eds.). The Routledge Handbook of FinTech. *Routledge International*, pp. 11-37.
- Driscoll, J. C., & Kraay A. C., (1998). Consistent covariance matrix estimation with spatially dependent panel data. *The Review of Economics and Statistics*, 80(4), pp. 549-560.
- Dumitrescu, E. I., & Hurlin, C. (2012). Testing for Granger Non-Causality in Heterogeneous Panels. Economic Modelling, 29, 1450-1460.
- Ehrlich, I. (1990). The problem of development: Introduction. *Journal of Political Economy*, 8(5), Part. 2, 1-11.
- Erumban, A. A., & Das, D. K. (2016). Information and communication technology and economic growth in India. *Telecommunications Policy*, 40(5), 412-431.
- Farhadi, M., Ismail, R., & Fooladi, M. (2012). Information and communication technology use and economic

growth. PloS One, 7(11), e48903. https://doi.org/10.1371/journal.pone.0048903

Frees, E. W. (1995). Assess cross-sectional correlation in panel data. Journal of Econometrics, 69, 393-414.

- Garces, E., & Daim, T. U. (2012). Impact of renewable energy technology on the economic growth of the USA. *Journal of the Knowledge Economy*, 3, 233-249.
- Gomes, S., Lopez, J. M., & Ferreira, L. (2022). The impact of the digital economy on economic growth: The case of OECD countries. *Revista de Administração Mackenzie*, 23(6) 1-31.
- Hafru G. G. (2019). Information communications technology and economic growth in Sub-Saharan Africa: A panel data approach. *Telecommunications Policy*, 43(1), 88-99.
- Hausman, J. A. (1978). Specification tests in econometrics. Econometrica, 46(6), 1251-1271.
- Hu, D., You, K., & Esiyok, B. (2021). Foreign direct investment among developing markets and its technological impact on host countries: Evidence from spatial analysis of Chinese investment in Africa. *Technological Forecasting and Social Change*, 166, 1-21.
- Hsing, Y. (2021). Impacts of mobile phone subscriptions, internet usage, and broadband penetration on economic growth in India. *International Journal of Applied Economics*, 18(2), 81-88.
- International Telecommunication Union. (2022). Statistics, 29 July 2022, https://www.itu.int/en/ITU-D/Statistics/Pages/stat/default.aspx. (Data of Access: 01.02.2023)
- Ishida, H. (2015). The effect of ICT development on economic growth and energy consumption in Japan. *Telematics and Informatics*, 32(1), 79-88.
- Iqbal, K., Sarfraz, M., & Khurshid (2022). Exploring the role of information communication technology, trade, and foreign direct investment to promote sustainable economic growth: Evidence from Belt and Road Initiative economies. *Sustainable Development*, 1-10.
- Jorgenson, D., W., & Vu K. (2005). Information technology and the world economy. Scandinavian Journal of Economics, 107(4), 631-650.
- Khan, H., Khan, U., Jiang, L. J., & Khan, M. A. (2020). Impact of infrastructure on economic growth in South Asia: Evidence from pooled mean group estimation. *The Electricity Journal*, 33(5), 1-8.
- Kireyeva, A. A., Kredina, A., Vasa, L., & Satpayeva, Z. T. (2021). Impact of financial technologies on economic development: Theories, methods and analysis. *Journal of International Studies*, 14(4), 286-303.
- Kurniawati, M. A. (2022). Analysis of the impact of information communication technology on economic growth: Empirical evidence from Asian Countries. *Journal of Asian Business and Economic Studies*, 29(1), 2-18.
- Li, Y., Jin, X., & Tian, W. (2019). Econometric analysis of disequilibrium relations between internet finance and real economy in China. *International Journal of Computational Intelligence System*, 12(2), 1454-1464.
- Madden, G., & Scott, J. S. (2000). Telecommunications and economic growth. International Journal of Social Economics, 27(7/10), 893-906.
- Mayer, W., Madden, G., & Wu, C. (2020). Broadband and economic growth: A reassessment. *Information Technology for Development*, 26(1), 128–145. https://doi.org/10.1080/02681102.2019.1586631
- Moulton, B. R., & Randolph, W. C. (1989). Alternative tests of the error components model. *Econometrica*, 57(3), 685-693.
- Nabi, A. A., Tunio, F., Azhar, M., Syed, M. S., & Ullah, Z. (2022). Impact of information and communication technology, financial development, and trade on economic growth: Empirical analysis on N11 Countries.

Journal of the Knowledge Economy, 1-18.

- Naeruz, M., Afiffudin, S., Ruslan, D., & Syafii, M. (2022). The impact of economic growth on technological developments, moneys and fluctuations interest rates and exchange rates in Indonesia. *E3S Web of Conferences*, 33(05008), 1-6.
- Nair, M., Pradhan, R. P., & Arvin, M. P. (2020). Endogenous dynamics between R&D, ICT, and economic growth: Empirical evidence from the OECD Countries. *Technology in Society*, 62, 2-12.
- Nasab, E. H., & Aghaei, M. (2009). The effect of ICT on economic growth: Further evidence. *International Bulletin of Business Administration*, 5, 46-56.
- Nchake, M. A., & Shuaibu, M. (2022). Investment in ICT infrastructure and inclusive growth in Africa. Scientific African, 17, 1-16.
- Pasara, M. T., & Garidzirai, R. (2020). Causality effects among gross capital formation, unemployment, and economic growth in South Africa. *Economies*, 8(2), 1-26.
- Pesaran, M. H. (2004). General diagnostic test for cross section dependence in panels. CESifo Working Paper, 1229, 46.
- Pesaran, M. H. (2007). A simple panel unit root test in the presence of cross section dependence. *Journal of Applied Econometrics*, 22, 265-312.
- Pohjola, M. (2001). Information technology and economic growth: Introduction and conclusions. Matti, P., (Eds.). Information Technology, Productivity, and Economic Growth: International Evidence and Implications for Economic Development. Oxford Academic.
- Poku, K. (2022). The influence of government expenditure on economic growth in Ghana: An Ardl approach. *Cogent Economics and Finance*, 10(1), 1-16.
- Romdhane, Y. B., Loukil, S., & Kammaoun, S. (2020). Economic African development in the context of FinTech. Akbar, R. Vd., (Eds.). Employing Recent Technologies for Improved Digital Governance. *IGI Global Publisher of Timely Knowledge*, pp. 273-286.
- Romer, P. M. (1990). Endogenous technological change. *Journal of Political Economy*, 98(5), Part 2, 71-102.
- Sadigov, S., Lyulyov, O., & Vasylieava, T. (2020). Economic growth and public governance: foresight scenarios. 55th International Scientific Conference on Economic and Social Development, Economic and Social Development, 1/4, 613-624.
- Salahuddin, M., & Gow, J. (2016). The effects of internet usage, financial development, and trade openness on economic growth in South Africa: A time series analysis. *Telematics and Informatics*, 33(4), 1141-1154.
- Schueffel, P. (2016). Taming the beast: A scientific definition of Fintech. *Journal of Innovation Management*, 4(4), 32-54.
- Shen, Y., Hu, W., & Hueng, C. J. (2021). Digital financial inclusion and economic growth: A cross-country study. *Procedia Computer Science*, 187,218-223.
- Shkarlet, S., Dbuyna, M., & Zhuk, O. (2018). Determinants of the financial services market functioning in the era of the informational economy development. *Baltic Journal of Economic Studies*, 4(3), 349-57.
- Song, N., & Appiah-Otoo, I. (2022). The impact of fintech on economic growth: Evidence from China. *Sustainability*, *14*(10), 1-17.
- Solow, R. M. (1956) A contribution to the theory of economic growth. The Quarterly Journal of Economics,

70(1), 65-94.

- Spearman, C. (1094). The proof and measurement of association between two things. *The American Journal of Psychology*, 15(1), 72-101.
- Sassi, S., & Goaied, M. (2013). Financial development, ICT diffusion and economic growth: lessons from MENA region. *Telecommun Policy*, 37(4-5), 252-261.
- Tatoğlu Y. F. (2018). Panel Zaman Serileri Analizi. Beta Yayıncılık, İstanbul.
- Tatoğlu Y. F. (2020). Panel Zaman Ekonometrisi. Beta Yayıncılık, 2020, İstanbul.
- Toader, E., Firtescu, B. N., Roman, A., & Anton, S. G. (2018). Impact of information and communication technology infrastructure on economic growth: An empirical assessment for the EU countries. *Sustainability*, 10(3750), 1-22.
- Torero, M., Chowdhury, S. K., & Bedi, A. S. (2006). Telecommunications infrastructure and economic growth: A Cross-country Analysis. Maximo, T., & Joachim, V. B. (Eds.). Information and Communication Technologies for Development and Poverty Reduction: The Potential of Telecommunications. *The John Hopkins University Press*, pp. 21-63.
- Türedi, S. (2013). Bilgi ve iletişim teknolojilerinin ekonomik büyümeye etkisi: Gelişmiş ve gelişmekte olan ülkeler için panel veri analizi. *Gümüşhane Üniversitesi Sosyal Bilimler Elektronik Dergisi, 4*(7), 298-322.
- Utami, Y. (2022). Analysis of the impact of financial technology (FinTech) on the Indonesian economy in the New Normal Era. *International Journal of Economics, Business and Accounting Research, 6*(4), 1-6.
- Yapraklı, S., & Sağlam, T. (2010). Türkiye'de bilgi iletişim teknolojileri ve ekonomik Büyüme: Ekonometrik bir analiz. Ege Akademik Bakış, 10(2), 577-598.
- Yılmaz, H. D., & Kırışkan, I. (2020). Türkiye'de telekomünikasyon altyapısı ve ekonomik büyüme. *bilig Türk Dünyası Sosyal Bilimler Dergisi*, 92, 55-84.



Istanbul Business Research, 53(2), 249-278

DOI: 10.26650/ibr.2024.53.1217731 http://ibr.istanbul.edu.tr/ http://dergipark.org.tr/ibr

Istanbul Business Research

Submitted: 12.12.2022 Revision Requested: 10.07.2023 Last Revision Received: 30.08.2024 Accepted: 31.08.2024

RESEARCH ARTICLE

Lean Management's Effect on Industry 4.0 Regarding Innovation

Serkan Naktiyok¹ (), Mehmet Burhanettin Coşkun² (), Fatma Temelli³ ()

Abstract

This study intends to investigate the relationship between lean management, Industry 4.0, and innovation perceptions in employees of companies in Turkey's general tourism sector and to determine whether these perceptions have an impact on how lean management influences perceptions of Industry 4.0. Businesses are compelled to participate in innovation activities due to the fourth industrial revolution, or industry 4.0, which is characterised by the rapid advancement and change of technology. On the other hand, there is a dearth of empirical research on the impact of lean management practises in enterprises on innovation and industry 4.0. Thus, by examining the connection between lean management, industry 4.0, and the innovation process as well as the impact of lean management on these three elements, the study seeks to advance the area.

The sample consists of 527 employees in employment in the tourism sector (hotels, agencies). Surveys were used to gather data, which structural equivalency modelling was used to assess. According to the study's findings, lean management has a favourable impact on innovation and industry 4.0. Another finding is that innovation affects Industry 4.0. The emergence of new and creative ideas in businesses will lead to the creation or development of advanced production technologies. The principal discovery of the study indicates that innovation functions as a mediator in the correlation between industry 4.0 and lean management.

Keywords: Lean Management, Innovation, Industry 4.0, Tourism businesses

Introduction

At the beginning of the twentieth century, scientific management ideas began to emerge in management principles and practices, and a discipline was formed under the name of the scientific management movement. Indeed, all systematic and scientific knowledge on the subject of this discipline has been systematised into theories and has entered the scientific maturation process since the middle of the twentieth century. The most prominent subject here was Frederick W. Taylor.



¹ Corresponding Author: Serkan Naktiyok (Assoc. Prof. Dr.), Atatürk University, Faculty of Economics and Administrative Sciences, Erzurum, Turkiye. E-mail: snaktiyok@atauni.edu.tr ORCID: 0000-0003-1226-2940

² Mehmet Burhanettin Coşkun (Assoc. Prof. Dr.), Osmaniye Korkut Ata University, Faculty of Economics and Administrative Sciences, Osmaniye, Turkiye. E-mail: burhanettincoskun@osmaniye.edu.tr ORCID: 0000-0001-7138-7669

³ Fatma Temelli (Assoc. Prof. Dr.), Ağrı İbrahim Çeçen University, Faculty of Economics and Administrative Sciences, Ağrı, Turkiye. E-mail: ftemelli@agri.edu.tr ORCID: 0000-0001-7436-5289

To cite this article: Naktiyok, S., Coskun, M. B., & Temelli, F. (2024). Lean management's effect on industry 4.0 regarding innovation. Istanbul Business Research, 53(2), 249-278. http://doi.org/10.26650/ibr.2024.53.1217731

Taylor asked three fundamental questions while theorising scientific management thought. The first is where the principles of scientific management differ from other classical understandings. Second is why and how better results will be achieved in scientific management than others. The third question Taylor asked is whether getting the right person to head the company is the most important issue or not. He stated that he developed the Taylor theory to find answers to these questions (Taylor, 1985). The Scientific Management Theory, which started the first studies on business, emerged at this stage.

The supporters of the theory believed that it was possible to conduct business management according to scientific principles, and in this process, under the leadership of Taylor, they suggested that management should be a discipline with clearly defined and unchanging principles instead of relying on vague ideas (Örnek, 1991).

In today's competitive conditions, lean management plays a significant role in establishing a sustainable advantage from the rational outputs of any production tool or technique. It is crucial for businesses that use high-cost production systems and need solutions to the crises in existing markets. Lean management techniques are typically built on eliminating all business endeavours that do not add value to the market for rival competitors' actions.

Meanwhile, the business and related service-production activities are handled holistically. Another issue that should be considered here is that employees' physical and intellectual contributions to service quality must be at the core of lean management.

Like lean management, innovation is one of the basic concepts that should increase service quality. The twenty-first century is still feeling the effects of the industrial revolution. Previously, businesses were conducted with economies of scale and high production volumes, which provided severe competitive advantages. However, today, the situation has begun to change with the emphasis for low cost, high quality, and convenience. Historically, these conceptual changes, encountered in the 1970s and 1980s, first emphasised low cost but later revealed that quality was more important for competition. In addition, the interaction gained momentum with the quick shift in the industries' competitive landscape, the significance of economic-based sustainability, and the prominence of environmentally friendly initiatives and activities.

Therefore, the intensity of the production processes in question, with the impact of changing importance levels of the concepts of low cost, high volume of production and quality, and from another point of view, the approaches such as the contingency approach, total quality management, and learning organisations, has been at the centre of the studies that will contribute to the creation of innovation, along with the novelties and thoughts formed within the extent of Industry 4.0, a significant process of change. Tourism has continued to increase over the past 60 years, making it one of the biggest and fastest-growing industries in the world; it is expected that there would be more than 1.8 billion foreign visitors worldwide by 2030. (UNWTO, 2017). Numerous development opportunities are available to heterogeneous tourism-related enterprises, such as tour operators, intermediaries, transfer firms, lodging facilities, and dining establishments. However, as technology advances, social, economic, political, and environmental issues will all have a significant impact on their future. One trend that undoubtedly has a big influence and being relevant in the growth of tourism is the advancement of technology. Industry 4.0's adoption of cutting-edge technology is well known for its disruptive power, affecting various industries, including travel and tourism. (Korže, 2019).

With the development of technology, like many other industries, tourism has seen a rise in the use of e-applications. The fourth industrial revolution, or Industry 4.0, is what we are currently experiencing, which has completely transformed numerous industries due to the automation, electrification, computerisation, and digitalisation of tasks (Kagermann et al., 2013). During the fourth industrial revolution, or Industry 4.0, the innovations that create added value in the tourism sector have become more critical than ever (Imanovic et al., 2016). Due to the significant contribution the tourist industry makes to the growth of national economies, the transition to the digitalisation era is essential for the sector to support the economics of the nation. Additionally, regularly monitoring and adjusting to the digitisation processes is how tourism organisations can outperform their rivals in terms of performance (Okatan & Yıldırım, 2021). Again, the effective use of technological opportunities is an important issue regarding giving tourists the possible service. As a result, using digital tools in tourism with the advancement of technology can be quite beneficial for travel agencies (Dülgaroğlu, 2021).

Every sector has some critical strategic factors ensuring its growth and sustainable development. The most important variables in the fiercely competitive tourism industry are innovation and the creation of new services. Companies in the service industry are compelled to innovate by changes and transformations, and innovation is essential to raising the calibre of services, particularly in the travel and tourism sector (Işık et al., 2019). The tourist industry is a large segment within the service sector, which is significantly affected by innovation activities. With the increase in competition, the essential way for a service industry to offer a long-term edge over competitors is to make a difference with innovation in its products or services. Therefore, tourism businesses that perform innovation activities will be able to create customer satisfaction, improve their company's performance and acquire a long-lasting competitive edge. Studies on applying lean management in many industries, including construction and health, occupy a notable place in the literature. However, there is no study focusing on the tourism sector. Therefore, addressing the applicability of lean management in the tourism sector will close a gap in the literature about its effects on production. This research's objective is to investigate the connection between Industry 4.0, lean management and innovation perceptions in the employees of businesses operating in the tourism sector in Turkey (hotels, agencies) and to reveal whether their innovation perceptions affect the effect of understanding Industry 4.0 through lean management. In this regard, it is intended to make a literary contribution by revealing employees' perceptions or awareness. The staff members were given a questionnaire for this reason. The current economy and production dynamics were considered in the subject selection; the modern methods seen as the requirements of the businesses, and the meaning areas formed for interaction were examined and defined for tourism businesses.

In the article, lean management, innovation and industry 4.0 were discussed conceptually, studies in the literature dealing with the relationship between these concepts were included, research data were collected, and the necessary analyses were made and reported. Then, the results were discussed, and the study was completed with a general evaluation and conclusion, which suggested more comprehensive future studies.

Lean Management

John Krafcik, an MIT researcher, used the concept of "lean" in 1988. He is the first engineer to be employed by the NUMMI automotive company, a partnership between the Japanese auto manufacturer Toyota and the American auto manufacturer GM. "Lean Management" is stated as "Toyota Production System" (TPS). In the beginning, "TPS" and later "Lean Administration" started to attract the attention of academia and the business world as a production and management methodology. TPS was also viewed as the driving force in industrial development in general and the creation of the Japanese miracle specifically in the growth of the automotive sector. This understanding of administration is thus also referred to as the "Japanese Administrative System".

Lean Thinking is an approach established to determine the added value, sequence the actions that will create value to achieve the best result, implement them instantly when requested, and perform the tasks more efficiently and effectively. In short, lean thinking means simplicity. The essence of this simplicity is the minimisation of human labour. On the other hand, lean management achieves the maximum return after this minimisation. This way becomes more effective in meeting customer demands (Womack & Jones, 2003). Lean management is a production theory that focuses on the timely delivery of the best quality product to the consumer and its production at the lowest cost.

Today, many businesses are trying to develop various improvement methods. Many effective techniques have been tried, including six sigma, kaizen, and total quality management. Although these methods are successful when they fit the setup of the businesses, many businesses cannot benefit from them. Lean management is mentioned worldwide for the first time in Womack and Jones' "The Machine That Changed The World". During promotional trips for the book, they encountered many companies eager to implement lean manufacturing, raising the question, "how do we do lean manufacturing?". The primary rule is to eliminate the waste completely and let the customer make all definitions (Womack & Jones, 2003).

Since the 1990s, the use of lean methods and the literature have been developing. Lean management has been successfully implemented in industries to increase service quality while maximising production and efficiency. Although the concept of lean has become synonymous with large companies, mass production and the automotive industry, it has also started to gain an important place in small and medium-sized businesses, or SMEs and the service sector. Numerous industries apply lean management, including healthcare, instruction, construction, maintenance, insurance and financial services, product development, IT-supported services, airlines, food and beverage services, logistics services, and public services, and continues to grow within the service industry (Cuatrecasas, 2004; Gupta et al., 2016; Perdomo-Verdecia et al., 2022).

"Lean management is a management type established for the realisation of lean manufacturing, where the delegation of authority is increased, everyone is responsible for their work, and zero hierarchy is targeted" (Tikici et al., 2006).

The objectives of lean management are to reduce costs by preventing waste and using the workforce most efficiently. Human resources act as the building block in the operation of processes and are not seen as a cost area. In this context, the way to prevent waste and ensure that processes operate at the highest level is to create cooperation and include human resources in the processes (Engelund et al., 2009).

There is more than one definition for lean manufacturing and lean management. Lean manufacturing is all systems and techniques that adopt the vision of optimising the offered value by eliminating waste and simplifying the product and service creation phase, and eventually increasing the profitability of the company (Sığırcı, 2001).

One method that emerged because of lean thinking is called lean management. Value, excellence, and value streams are all part of lean thinking (Deran & Beller, 2014). Therefore, lean management can be described as a management style that is closely related to the following principles:

1. *Principle of flow:* A simple transformation can be described as a journey involving two large movements. The first of these movements is to create flow, and the second is to create pull. The first focus should be on creating flow, and as with all important journeys, having a checklist is thought to help ensure that the organisation does not stray (Byrne, 2015).

- 2. Value principle: The starting points of lean thinking are values. Manufacturers create these values and customers can define them. The producer who created the values cannot always define them correctly. For example, they think that American firms create value in the short run through competition tactics and profit transfers from the supplier at the head of the chain. German companies with strong technical equipment define values in relation to the technical complexity of their products and technologies. In Japan, the issue of where values are created is important. Value is being able to reconsider from the standpoint of the client. For value to have meaning, it must be expressed regarding a good or service that can fulfil the requirements of clients in a certain time and at a certain price (Warnecke & Hüser, 1995).
- 3. *Value stream principle*: The next stage of lean thinking is to ensure that the valuecreating stages are in flux. To be more precise, it involves redefining the roles of functions, departments and firms in a manner that favourably influences the development of value for businesses. Although this idea contradicts the departmentalised accumulation mentality, which is generally accepted as true in our minds, it states that by working on a workpiece from raw material to final product, tasks can be performed much more accurately and efficiently. In short, it can be seen that things get a lot better when the focus is on the product and what the product requires, instead of the organisation or equipment, so that the activities required for the design, order and manufacturing stages take place in a continuous flow (Sığırcı, 2001).
- 4. *Pull principle:* Another principle of lean thinking is the concept of "pull". In its most general definition, pulling means that no product or service is produced in the previous processes without the request of the customer in the next processes. The best way to grasp the basic logic of pull thinking is to start with the customer's demand for a particular product and look back through the entire process from the product to the customer's arrival (Düren, 2002).
- 5. Principle of excellence: It becomes increasingly entrenched in the minds of those involved in the business that there is no end to the time, effort, place, error reduction and cost reduction process at the stage of offering a product that is very close to their customers' demands. Perhaps the best promoter of excellence is transparency. The first step in this lean method is the fact that suppliers, subcontractors, system integrators, distributors, employees and customers, in short, everyone can see everything. For this reason, it becomes easier to find healthier ways to create value, and there is very positive and instant feedback for the employees who make the improvement. As stated in this section, lean is the key point of the business and a strong supporter of efforts to improve (Çanakçıoğlu, 2019).

Additionally, even if Lean Management is best for standardised processes, businesses are increasingly being forced to adopt more adaptable strategies to fulfil the expectations of the clients (Handscomb et al., 2020). Thus, agile management should also be considered in addition to lean administration. In addition, whether or not the industry determines how much money can be saved by applying lean concepts. An analysis of a case study that appeared in the International Journal of Production & Operations Management (Lewis, 2000) specifically shed light on this issue.

Innovation

The concept of innovation has recently become a concept that attract the attention of researchers in various fields. In general, innovation is a progressive act of distinctive goods and services that add value to companies in an intensely competitive environment by developing new ideas. With the developing new conditions in the changing competitive environment, consumers tend to turn to new style products and services that are beyond the ordinary, that are developed formally, that bring constant added value, and that are revealed with their visuals.

Consumers show these needs and orientations in a fast and rapid transformation line. In such a challenging and extremely competitive environment, all companies, whether large or small, can maintain their competitiveness by innovating in their products or services and thus survive. For this reason, companies today can only meet the rapidly developing needs of consumers through innovation and manage the conditions of competition in their own direction. Accordingly, the innovation referred to as "innovation" not only does something new, advanced, harmonious and different, but also opens a new way that can create economic value for the benefit of the company (Işık & Keskin, 2013).

According to Schumpeter (1961), innovation means novelty in the basic sense. In this period, innovation was generally expressed as "the series of works that offer practical solutions to the needs and that can be used commercially and accepted as a novelty". Schumpeter described innovation as inventing a new production technique, producing a new product, or enhancing the quality of an already-existing one.

According to Myers & Marquis (1969), innovation is more than just a method, a fresh concept, an invention, tool, or result; it is the whole process consisting of many sub-processes.

In its broadest sense, innovation is the combination of creativity and practice (Lyons et al., 2007). Therefore, innovation is the use of a creative idea commercially. In this sense, it starts from the first stage and reaches the consumer at the end.

According to Kucmarski (1996), most today's managers see innovation as an essential tool for increasing incomes. On the other hand, the managers' tendency to consider innova-

tion as a fundamental business strategy has not increased. All managers must see innovation as a powerful determinant that increases competitiveness in building their strategy. Freeman and Soete (1997) suggest that businesses can draw a path by using the opportunities and alternatives specific to their companies while determining their strategies. For example, short, medium, and long-term strategies can be set; these companies can merge with other companies or switch to different markets with different products and methods. Because of all these different actions, they may encounter significant risks and dangers, especially in technology, materials, and chemistry. The majority of those who fail to manage these processes will not survive. They will face the necessity of re-entering the innovation cycle and have to manage it. This is because their competitors will lower product costs using the innovation cycle and will seize the market earlier. Therefore, they will realise they cannot move forward without implementing an agile, defensive or imitative innovation strategy.

According to the Oslo Manual, one of the first documents that come to mind for inventiveness, "An innovation is the implementation of a new or significantly improved product (good or service), or process, a new marketing method, or a new organisational method in business practises, workplace organisation or external relations" (Oslo Manual, 2005).

The Innovation Support Guide prepared by the Turkish Chamber of Commerce and Exchanges (TOBB) defines innovation as "transforming an idea into a sellable product or service, a new or improved manufacturing or distribution management, or a new social service method".

In addition, Çavuş and Akgemci (2008) draw attention to establishing a team and ensuring the participation of employees with a system that will make it easier to understand technological, social, and local changes to turn innovation into a fundamental skill.

Lean management and innovation approaches intersect within the body of businesses and organisations desiring to perform high-quality activities in the institutional sense. Employeecentred innovation management emphasises employees and their functional power and places considerable value on them; it agrees with the lean management approach. Accordingly, the strategy of creating a working environment that appeals to employees' physical and mental health in lean management is a significant management move towards more qualified activities. The steps realised to improve the operational performance and psychological state of the employees in line with the idea of innovation bring the two management approaches closer.

Industry 4.0

In the past decade, information and communication technologies (ICT) have become more prevalent, and a radical transformation has been witnessed in daily life. The size of computers

is decreasing and they are disappearing into almost all technical devices. Primarily, items converse via the Internet or World Wide Web. This tendency has spread to the industrial sector as well, which stands to gain more from the developments in computer science and ICT.

This movement is known as the "4th Industrial Revolution," or Industry 4.0, in Germany (Kagermann et al., 2013). Shorter product life cycles are closely linked to highly personalised goods and the development of modern factories into "smart factories," which aim to solve and defeat the difficulties posed by the current intense global competition (Weyer et al., 2015).

Industry 4.0 refers to the Fourth Industrial Revolution. It is acknowledged as the next development after the outsourcing boom of the 1990s and the lean revolution of the 1970s, and the digitisation of the manufacturing industry with mechanisation systems in the early 2000s (Uy & Rabo, 2019). The concept of "Industrie 4.0" originates from Germany's digitalisation programme aimed at boosting the manufacturing sector's competitiveness (Ghobakhloo et al., 2021). The idea of Industry 4.0 first surfaced in 2011 at a trade show in Hannover, Germany and brought a wave of development to the industry (Nascimento et al., 2021). It has received tremendous attention in academia since its publication in 2011. Many scientific advances have greatly enhanced our knowledge and comprehension of the Industry 4.0 phenomenon (Ghobakhloo et al., 2021. The goal of Industry 4.0 in Germany was to develop these systems through collaboration between universities and commercial businesses, with funding for research and development coming from the government (Zengin et al., 2021).

The Industrial Revolution is described as the transformation that took place between the end of the eighteenth century and the mid-1800s, in which small-scale workshops were transformed into industrial workshops carrying out large-scale production. It is also expressed as the technological change that allowed the workforce to turn into machines (Frederick, 2016). It is believed that Industry 4.0 marks the start of the fourth industrial revolution and displays the most recent automation technology trend that is becoming increasingly common in the manufacturing sectors (Sony, 2018). Every unit involved in manufacturing, whether directly or indirectly, is planned to work together, and information technologies and digital data software operate together in a cohesive way. (Schuh et al., 2014). Industry 4.0 designates the industrial sectors as having entered the fourth industrial revolution. This new revolution shows itself as an increase in digitisation and diversification and the association between the products, value chains, and business models (Banger, 2017).

As Industry 4.0 provides information about the developments related to future factories, it is seen as an intelligent technological system with a dramatic impact on the industrial sector; and it is the subject of much research (Pereira & Romero, 2017). Industry 4.0 has become a really interesting technology to achieve efficient, accurate, and precise business outcomes (Rajput & Singh, 2019). Meanwhile, Industry 4.0 technologies have created advantages such as higher flexibility, shortening delivery times, and adaptation to customer demands with

small batch sizes (Wang & Wang, 2016). The fundamental feature of Industry 4.0 is that it includes automatically self-configuring and self-optimising systems that allow more agile and low-cost operations in production (Asdecker & Felch, 2018).

In summary, the fourth industrial revolution idea, or Industry 4.0, significantly affects the country's economy, business world, society, and environment. The effects of Industry 4.0 and advancements in technology on companies operating in various sectors are increasing day after day. Industry 4.0 is a new process with many components, including cyber-physical systems, artificial intelligence, analytics, cloud technology, the objects with internet, and intelligent factory. It is predicated on the cooperation of humans with machines, thus aiming to bring customer-specific products to the market faster and more flexibly. In other words, companies are increasingly moving with the idea of Industry 4.0 as a crucial tactic for surviving in a cutthroat market.

Literature Research

Lean Management and Tourism

Today, lean management continues to evolve and is implemented in numerous service industries, including food, finance, and transportation, and mostly occurs within public services (Hadid & Mansouri, 2014; Leite & Vierira, 2015; Gupta et al., 2016; Narayanamurthy & Gurumurthy, 2016). Lean management (Vlachos & Bogdanovic, 2013; Farrington et al., 2018; Perdomo-Verdecia et al., 2022) practices have become priority issues in the tourism sector literature, and the number of studies needed is increasing. Below is a summary of a few few studies that have been done on the topic in the literature:

Vlachos and Bogdanovic (2013) looked into how lean thinking could be used in European hotel companies. With the dearth of lean practice literature in the hotel business, this study might be regarded as a crucial resource. The study claims that one can use the "Five Step Model" as a management tool to reduce waste and boost customer value. The underlying premise of the model is that there are five sequential sequences of actions from Value to Excellence.

Al-Aomar and Hussain (2019) developed a methodology for recognising, classifying, and prioritising lean approaches applied in a hotel supply chain. In order to characterise the lean techniques used in the United Arab Emirates' four- and five-star hotels, they gathered information using a questionnaire. Six primary lean application categories and 19 particular lean methods pertaining to a hotel supply chain were identified by the study. JIT, Kaizen, Quality, Inventory, Maintenance, and Standardisation are examples of the defined categories. The study's findings indicate that, of the lean application categories that were found, JIT and Kaizen have the highest priority. In addition, prompt customer service, an efficient enhancement

framework, and supplier delivery on schedule are the three lean approaches that are most appropriate for a hotel supply chain.

In their study on hotel lean management, "Where we are and where can we go?" Perdomo-Verdecia et al. (2022) attempted to provide answers to these queries. There were only 36 references retrieved, indicating the dearth of studies conducted in this field. With the popularity of VSM and 5S applications, the adoption of lean management solutions in hotels appears to be quite limited. The reference study unequivocally demonstrated the necessity for considerably more research on specific practices—some of which are underreported—in many hotel operations.

Innovation and Tourism

Innovation is a means to competitiveness, environmental sustainability, and economic success. Therefore, tourism-related firms should always be innovating. Furthermore, given how quickly rivals imitate winning concepts, innovation in the tourist industry is crucial to giving destinations a long-term competitive edge over rivals. As a result, competitors should find it difficult to embrace advances (Giotis & Papadionysiou, 2022).

The innovation theory of Schumpeter provides a basis for the literature on tourism and innovation (Alsos et al., 2014). The study, finding, creation, promotion, approval, and commercialisation of novel goods, systems, organisational configurations, and practices are all considered forms of innovation, according to the notion. This strategy is recognised as a widely acknowledged strategy for all other service sectors and the tourism industry (Hjalager, 2002; Nordin, 2003). Below is a summary of a few studies that address innovation in tourism and the connection between innovation and tourism in the literature:

Topsakal et al. (2018) used a questionnaire to gather information on how small and medium-sized tourism enterprises are embracing innovation ideas in Antalya (Kaleiçi), Turkey. The investigation's findings indicate that although small and medium-sized tourism enterprises want innovation to improve service quality, cost is the primary barrier to innovation.

Mendoza-Moheno et al.'s study from 2021 sought to pinpoint the social and technological frameworks that community-based tourism organisations use to sustain their socio-technical innovation system. The results of the study add to the discourse on stakeholder cooperation and offer an effective model of an organisation that satisfies social, environmental, and economic demands. The study's demonstration of the interaction between technical and social systems is one of its key contributions.

Giotis and Papadionysiou (2022) analyzed the empirical research to assess the influence that managerial and technological advances play in the tourism business. Within the context of the fundamental role that tourism benefits from advances in technology and management, this paper tries to offer a thorough summary of the empirical studies conducted to look into these topics. The findings of a few studies showed that a company's ability to foster innovation is increased when it has a pleasant workplace with suitable personnel, a management style that is appropriate, and managers and employees who work together. Additionally, several management implications have been put up to help managers create effective strategies for companies operating in the tourism sector.

Industry 4.0 and Tourism

Every day, industry 4.0 developments have an increasing impact on the economy, finance, and commercial sectors. The industrial revolution has sped up the growth of the tourism industry. Consequently, the tourism industry is among those where Industry 4.0 technologies have a bigger pragmatic effect. Given the service-oriented nature of the tourist industry, it can be argued that Industry 4.0 advancements will significantly alter how people perceive travel.

There is not much research in the literature on the connection between Industry 4.0 and the travel and tourist industry. A few of these studies are outlined below:

To better understand the present circumstances and future prospects of service automation and robot adoption by travel, tourism, and hospitality-related businesses, Ivanov et al. (2017) carried out a study that included a description of the difficulties that these businesses will have while implementing these technologies to assist visitors.

Osei et al. (2020) planned to thoroughly evaluate the expectations of business 4.0 for the lodging and tourism business by conducting a scientific projection to uncover the expectations of the fourth industrial revolution for the hotel industry. Remarks about the expectations (good and bad) for the lodging industry were included in the study. The discussion concludes with the practical and societal ramifications.

In his research, Şengel (2021) outlined the historical progression of the relationship between contemporary tourism trends and the industrial revolution. The goal of this study was to ascertain how the industrial revolution affected contemporary tourist movements from the outset to the present. The research findings indicate that the industrialisation initiatives that sparked the industrial revolution have had a major influence on the development of contemporary tourism movements.

Chang et al. (2022) conducted a study on the adoption of the blockchain technology for tourism based on UTAUT and the connectivity theory. Their work highlights the potential of using social media (YouTube) to increase user knowledge and potential interest in the domestic travel industry, as well as providing an actual implementation of blockchain technology.

Method

In this section, we used SPSS and AMOS to analyse the data collected through the survey technique. In this context, the statistical methods and techniques used have been explained.

Purpose of the Study

This study aims to reveal the role of the innovation perceptions of employees in businesses operating in the tourism sector in Turkey (hotels, agencies and restaurants) on the effect of lean management perceptions on Industry 4.0 perceptions.

Method of the Study

This study is a quantitative research in which measurements can be repeated and objective, and various statistical methods are used (Elitaş, 2011).

Universe and Sample of the Study

The universe of the study consists of employees employed in the tourism sector in Turkey (hotels, agencies). According to TUIK data, around 34.334.000 people are in the workforce in Turkey (TUİK, 2022). If the size of the universe is 250,000 or more, the minimum sample size should be 384 (Gürbüz & Şahin, 2017). Therefore, the participation of 527 employees in the study provides a sufficient sample size.

Assumptions and Limitations of the Research

The scales used in the study have been qualified to explain the factors (lean management, innovation, industry 4.0) that are the subject of the analysis. Reliability and validity analyses were also performed in this study for scales that showed high reliability and validity in previous studies; the reliability of the measurement tools used was determined to be high. Therefore, it is assumed that the questionnaire reflects the current situation regarding the perceptions of lean management, industry 4.0, and innovation within the organisation.

The human element is the basis of the study, and the data includes evaluations of the employees. Therefore, the general limitations of social sciences research are also valid for this study. The reliability of the findings is limited to employees' evaluations and the characteristics of the survey technique used in data collection.

Data Collection Tools

As the adopted research and measurement method was based on numerical (quantitative) data, the survey technique was used for data collection. The questionnaire starts with 5 qu-

estions prepared to collect the employees' demographic information (gender, age, education level, marital status, and working time). In the first part of the questionnaire, the 24-item Lean Management Scale developed by Derin (2008) was used to measure the lean management perceptions of the employees. The second part of the questionnaire consists of an 8-item scale used by Arpacı (2019) to determine the innovation process. The third part includes a 6-item scale used by Zengin et al. (2021) to measure Industry 4.0. The scales used are of the Likert type.

Research Hypotheses and the Model

Because of the literature review above, the following hypotheses were developed (5.1., 5.2, 5.3.).

The hypotheses of the study are as follows:

H1: Lean management has a positive impact on industry 4.0.

H2: Lean management positively affects innovation.

H3: Innovation has a positive impact on industry 4.0.

H4: Innovation plays a mediating role in the effect of lean management on industry 4.0.



Figure 1. Research Model

Data Analysis

In this study, statistical package programmes were used to analyse the data. The significance level was taken as p = 0.05. First, the reliability and validity of the scales were tested. The reliability of the scales was calculated through internal consistency using Cronbach's alpha coefficients. Exploratory and confirmatory factor analyses were used to test the reliability and validity of the questionnaires. First, correlation analysis was employed to test the hypotheses that form this study's basis and determine the relationship between lean management, industry 4.0, and innovation. In the study, the structural equation model was used to determine whether the variables were statistically significant. In this direction, the structural model was tested with the AMOS programme. The relationships between the variables were examined using goodness-of-fit statistics and hypotheses to test the model's validity. The bootstrap method was used to test the mediating roles between variables, and the 5-point Likert scale was used in this research. In the selection of the sample, a simple random method was used, which makes the chance of entering the sample equal for every employee in the phase.

Findings

Table 1

The findings regarding the demographic information of the employees participating in the study are shown in Table 1.

Demographic Information					
Variable	Category	f	(%)		
Candar	Female	156	29.6		
Gender	Male	371	70.4		
	21-30-year-old	110	20.9		
	31-40-year-old	206	39.1		
Age	41-50-year-old	158	30.0		
	51-year-old and more	53	10.1		
NF 1 1 C 1	Single	150	28.5		
Marital Status	Married	377	71.5		
	Elementary-High School	33	6.3		
	Associate's degree	89	16.9		
Education	Bachelor's degree	252	47.8		
	Postgraduate	153	29.0		
	1-5 years	223	42.3		
Working Time in	6-10 years	113	21.4		
the Tourism Sector	11-15 years	80	15.2		
	16-20 years	39	7.4		
	20 years and more	72	13.7		
	4,000-6,000-TL	95	18.0		
A	6,001-8,000 TL	79	15.0		
Annual Income	8,001-10,000	117	22.2		
	10,001 TL and more	236	44.8		
Managerial Po-	Yes	261	49.5		
sition	No	266	50.5		

First, an item analysis was conducted to test the reliability of the scales. The item-total correlation of all items was above 30, which is the cut-off point. Therefore, there was no need to remove any item from the scale in the first step (Çanakçı et al., 2019).

The second stage of the reliability and validity analysis was conducted to test the reliability and validity of the lean management, innovation, and industry 4.0 scales separately. For this purpose, exploratory factor analysis with Varimax rotation was performed using the principal component method.

Regarding the exploratory factor analysis results, the factor load of the Lean1, Lean6, Lean7, and Lean19 items in the lean management scale was below 0.40, and they were excluded from the analysis. KMO, which confirms the applicability of the descriptive factor analysis to the data, and the degree of sphericity (Barlett's Test of Sphericity), which show that significant factors can be obtained from the research data. The Cronbach alpha coefficients are displayed in the table below.

Table 2

Exploratory Factor Analysis Results

Variables	Dimensions	Scale Item	Factor Lo- ads	Cronback	ı's Alpha	КМО	Barlett Test of Sphericity (p)
		Lean5	.791				
		Lean4	.782				
	n 1	Lean9	.714	880			
	Lea	Lean3	.665	.000			
		Lean2	.656				
		Lean8	.604		.963		
<u> </u>		Lean20	.758	.956			
EN		Lean16	.733				
IW	n 2	Lean12	.731				
GF		Lean13	.715			060	8341.357;
NA		Lean22	.708			.909	p=0.000
MA		Lean24	.706				
Z		Lean23	.704				
JE A	Lea	Lean11	.703				
-		Lean17	.686				
		Lean14	.671				
		Lean21	.667				
		Lean10	.661				
		Lean15	.655				
		Lean18	.574				
			Eigenvalue: 7.4	492; Total Vai	riance: 64.434		

Variables	Dimensions	Scale Item	Factor Lo- ads	Cronbacl	h's Alpha	КМО	Barlett Test of Sphericity (p)
		Inno3	.879				
		Inno2	.862				
Z	п	Inno1	.858				
Ш	atic	Inno6	.826	021	021	0.022	3162.631;
INNOVA	nov	Inno8	.818	.931	.931	0.922	p=0.000
	Ц	Inno4	.815				
		Inno7	.771				
		Inno5	.741				
			Eigenvalue: 5.	409; Total Va	riance: 67.612		
		Ind_4_6	.902				
4.0	4.0	Ind_4_5	.864				
8V	iry	Ind_4_4	.833	804	804	870	1896.388;
ILSUD	lust	Ind_4_2	.784	.094	.094	.870	p=0.000
	Ind	Ind_4_3	.778				
Z		Ind_4_1	.697				
			Eigenvalue: 3.	961; Total Va	riance: 66.018		

Confirmatory Factor Analysis for the Lean Management Scale

After the exploratory factor analysis, confirmatory factor analysis was performed using the AMOS package programme on the lean management scale. Regarding the item loads obtained from the regression model, all the regression items' loads were higher than 0.50. The analysis showed that the modifications between Lean4-Lean5, Lean10-Lean18, Lean11-Lean12, and Lean20-Lean21 would increase the chi-square and fit values.



Figure 2. Modified Confirmatory Factor Analysis of the Lean Management Scale

Confirmatory Factor Analysis for the Innovation Scale

After the exploratory factor analysis, confirmatory factor analysis was performed on the innovation scale. Regarding the item loads obtained from the regression model, all the regression items' loads were higher than 0.50. The analysis showed that modifying Inno1-Inno2 would increase the chi-square and fit values. In addition, Inno6 and Inno7 were excluded from the analysis because they highly overlapped with other factors.



Figure 3. Modified Confirmatory Factor Analysis of the Innovation Scale

Confirmatory Factor Analysis for the Industry 4.0 Scale

After the exploratory factor analysis, confirmatory factor analysis was performed on the Industry 4.0 scale. Regarding the item loads obtained from the regression model, all the regression items' loads were higher than 0.50. The analysis showed that the modifications between Ind401-Ind405 and Ind403-Ind406 would increase the chi-square and fit values.



Figure 4. Modified Confirmatory Factor Analysis of the Industry 4.0 Scale

T J	Reference Value				
Index	Good Fit	Acceptable Fit			
CMIN/DF	$0 < \chi 2/sd \le 3$	$3 < \chi 2/sd \le 5$			
CFI	$.95 < CFI \le 1$	$.90 < CFI \le .94$			
TLI	$.95 < TLI \le 1$.90 < TLI≤ .94			
IFI	$.95 \le IFI \le 1$	$.90 < IFI \le .94$			
RMSEA	$0 \le RMSEA \le .05$	$.05 \leq RMSEA \leq .08$			
RMR	< 0.05	< 0.08			

Table 5				
Reference	Values	for the	Goodness	of Fit

T 11 2

Table 4

Industry 4.0

Source: Tasgin and Korucuk, 2018

Goodness of Fit Indices of the Models for Confirmatory Factor Analysis						
Variables	CMIN/DF	CFI	TLI	IFI	RMSEA	
Lean Management	3.3585	.948	.941	.949	.070	
Innovation	4.078	.989	.980	.989	.076	

2.807

CR (Composite Reliability) is a value that measures the internal performance of a factor. It is desirable for the value to be greater than 0.70. The CR value is constructed based on Cronbach's Alpha being considered as a better alternative. The AVE (Average Variance Extracted) value is a value that measures the similarity between the items of a factor. It is desirable for the value to be greater than 0.50.

.993

.985

.993

.021

The table below shows the CR composite reliability values of the lean management, innovation process and industry 4.0 scales. According to the table, the CR value is over 0.70, which shows that the scale is reliable. For the convergent validity of the scale, the mean self-variability is expected to be higher than 0.5 and the composite reliability ratio is expected to be greater than the mean self-variability (CR > (AVE) / AVE > 0.5).

It is evident from the figures in the table that AVE (self-variability) is greater than 0.5 and CR composite reliability) is greater than 0.7. These are the values at which there is enough convergence in the model (Hair et al., 2009).

Validity-Reliability Matrix					
Factors	CR	AVE	CR>0,70	AVE>0,50	CR>AVE
Lean Management	0,94	0,51	0,94>0,70	0,51>0,50	0,94>0,51
Innovation	0,95	0,68	0,95>0,70	0,68>0,50	0,95>0,68
Industry 4.0	0,92	0,66	0,92>0,70	0,66>0,50	0,92>0,66

Hypothesis Testing

Table 5

First, correlation analysis was used to test the hypotheses that formed the research's basis

RMR .047 .032

.059

and determine the relationship between lean management, innovation, and industry 4.0. The results of the correlation analysis are shown in Table 5. The table shows a positive and significant relationship between the organisation's lean management approach and innovation (.795**). At the same time, there is a positive and significant relationship between the lean management approach and industry 4.0 (.440**).

Table 6			
Relationship Between Variables			
Factors	1	2	3
1-Lean Management	1		
2-Innovation	.795**	1	
3-Industry 4.0	.440**	.498**	1
*n<.05, **n<.01			

Analysis of the Structural Equivalence Model

A structural equivalence model was used to determine the statistically significant relationships between the variables. For this purpose, the previously determined structural model of the research was analysed with the AMOS programme. The relationships between variables were examined using goodness-of-fit statistics and hypotheses to confirm the model's validity.

To test the effect of the mediating variables, the mediator was removed from the model, and the prediction power of the independent variable on the dependent variable was calculated. As a result, lean management has a 42.4% effect on industry 4.0. Accordingly, H1 is accepted. The estimations of the tested model are shown in Figure 5.



Figure 5. Standardised Estimation Results Showing the Effect of Lean Management on Industry 4.0

The fit indices of the model and the reference values specified in the literature are given in Table 6.

Index —	Referen	ce Value	M	D14
	Good Fit	Acceptable Fit	wieasurement	Kesun
CMIN/DF	$0{<}\chi 2/sd{\le}3$	$3 < \chi 2/sd \le 5$	3.089	Acceptable Fit
CFI	$.95 < CFI \le 1$	$.90 < CFI \le .94$.942	Acceptable Fit
TLI	$.95 < TLI \le 1$	$.90 < TLI \le .94$.935	Acceptable Fit
IFI	$.95 < IFI \le 1$	$.90 < IFI \le .94$.942	Acceptable Fit
RMSEA	$0 \leq RMSEA \leq .05$	$.05 \leq RMSEA \leq .08$.063	Acceptable Fit
RMR	< 0.05	< 0.08	.056	Acceptable Fit

Table 7

The Goodness of Fit Indices of the Models Showing the Effect of Lean Management on Industry 4.0

The effect of lean management on innovation is shown in Figure 6. Accordingly, lean management's effect on innovation is 87.8%. Thus, H2 is accepted. The estimations of the tested model are shown in Figure 6.



Figure 6. Standardised Estimation Results Showing the Effect of Lean Management on Innovation

Index –	Refer	ence Value	M (
	Good Fit	Acceptable Fit	Measurement	Result
CMIN/DF	$0{<}\chi 2/sd \leq 3$	$3 < \chi 2/sd \le 5$	3.360	Acceptable Fit
CFI	$.95 < CFI \le 1$	$.90 < CFI \le .94$.940	Acceptable Fit
TLI	$.95 < TLI \le 1$	$.90 < TLI \le .94$.933	Acceptable Fit
IFI	$.95 < IFI \le 1$	$.90 < IFI \le .94$.940	Acceptable Fit
RMSEA	$0 \leq RMSEA \leq .05$	$.05 \leq RMSEA \leq .08$.067	Acceptable Fit
RMR	< 0.05	< 0.08	.050	Acceptable Fit

 Table 8

 The Goodness of Fit Indices of the Models Showing the Effect of Lean Management on Innovation

The effect of innovation on Industry 4.0 is shown in Figure 7. Accordingly, the effect of innovation on Industry 4.0 is 51.4%. Thus, H3 is accepted. The estimations of the tested model are shown in Figure 7.



Figure 7. Standardised Estimation Results Showing the Effect of Innovation on Industry 4.0

Table 9

	· · · · · · · · · · · · · · · · · · ·	M 11 Cl ·	I FCC I	CT	11, 10
The goodness of F	it indices of the	Models Showing	пе Епесі ої	Innovation on	Industry 4.0
		· · · · · · · · · · · · · · · · · · ·			

Index –	Reference Value		M	D 14
	Good Fit	Acceptable Fit	Measurement	Kesun
CMIN/DF	$0 < \chi 2/sd \le 3$	$3 < \chi 2/sd \le 5$	3.984	Acceptable Fit
CFI	$.95 < CFI \le 1$	$.90 < CFI \le .94$.966	Acceptable Fit
TLI	$.95 < TLI \le 1$	$.90 < TLI \le .94$.955	Acceptable Fit
IFI	$.95 < IFI \le 1$	$.90 < IFI \le .94$.966	Acceptable Fit
RMSEA	$0 \leq RMSEA \leq .05$	$.05 \leq RMSEA \leq .08$.075	Acceptable Fit
RMR	< 0.05	<0.08	.063	Acceptable Fit

In the research model, lean management was considered an exogenous variable, and implicit variables such as innovation and industry 4.0 were considered endogenous variables. The innovation was added to the model as a mediating variable. The Bootstrap method was also used to reveal the validity of the mediating role while testing the relationships between the variables. In the Bootstrap method, the sample was increased by 1000, and the Bootfactor was set as 1. The maximum likelihood estimation method, which is accepted to be suitable for cases where the data show a normal distribution, was applied in this study. Figure 8 shows the estimation results of the model tested in the AMOS programme using the Bootstrapping method.



Figure 8. Displaying Standardised Estimation Results on the Model

Regarding the effect of lean management on industry 4.0 in the context of innovation, the innovation (mediator) was removed from the model first, and lean management's direct effect on industry 4.0 was examined. This effect was significant (.424 and p=000). Data related to this model are shown in Figure 2. Then, the innovation process, the mediating variable, was added to the model, and the analysis was repeated. In this case, the effect of lean management on industry 4.0 was significant (.186 and p=.001). As can be seen, the direct effect of lean management on industry 4.0 is significant without a mediator. On the other hand, when the mediator variable innovation is added to the model, the effect of lean management on Industry 4.0 decreased. Regarding the Bootstrap test results, the mediator variable's effect was significant. When the mediator variable was added to the model, the effect of lean management on industry 4.0 did not

disappear and decreased, which shows that innovation plays a mediating role in the interaction between lean management and industry 4.0. It is also possible to interpret this situation as lean management affecting industry 4.0 both directly and through innovation because the mediation effect was also significant according to the results of the Bootstrap test.

Regarding the goodness of fit indices of the model in Table 9, they indicate an acceptable fit.

Rejerence values for the Goodness of Fit						
Index -	Refer	rence Value	M (D K		
	Good Fit	Acceptable Fit	Measurement	result		
CMIN/DF	$0 < \chi 2/sd \le 3$	$3 < \chi 2/sd \le 5$	3.086	Acceptable Fit		
CFI	$.95 < CFI \le 1$	$.90 < CFI \le .94$.931	Acceptable Fit		
TLI	$.95 < TLI \le 1$	$.90 < TLI \le .94$.924	Acceptable Fit		
IFI	$.95 \le IFI \le 1$	$.90 < IFI \le .94$.931	Acceptable Fit		
RMSEA	$0 \leq RMSEA \leq .05$	$.05 \leq RMSEA \leq .08$.063	Acceptable Fit		
RMR	< 0.05	<0.08	.059	Acceptable Fit		

 Table 10

 Reference Values for the Goodness of F

Discussion and Conclusion

Due to the intense competition brought by globalisation, businesses face some difficulties in terms of cost, quality, and time. The increasing number of competitors increases the customers' decision-making options and causes harsh competitive conditions. To survive in this competitive environment, businesses want to determine efficient methods and processes the entire value chain. The lean management approach focuses on producing more with less time, efforts, and equipment and meeting customers' expectations. Uncertainty is a factor that makes management difficult in businesses; its presence complicates the management. Management eliminates this uncertainty and complexity by simplifying the processes and making them lean and elementary. Consequently, the decision-making processes of the business gain momentum as uncertainty and complexity decrease. From this perspective, lean management is an approach that focuses on creating more efficient working conditions by reducing nonvalue-adding activities and significantly increasing business performance. The lean management approach eliminates waste by continuously improving the production process. Businesses should adopt the lean management approach to improve their innovation processes and keep up with changing competitive conditions because developing new products and processes plays a vital role in gaining competitive power. Therefore, lean management can significantly contribute to the improvement of a business' innovation capacity. In addition, the concept of Industry 4.0, called the fourth industrial revolution, has emerged recently in the production sector. After a detailed review, it can be said that lean manufacturing methods and Industry 4.0 technologies are intertwined and act on each other.

The tourism sector is at a very different point today, with the development of technology. Industry 4.0, which includes many industries in the world, and the innovation process have also affected the tourism industry.

Among these innovative movements in the tourism sector, there are issues such as businesses designing beautiful websites, developing mobile applications, providing search and research opportunities and conveniences to their customers with social networks, web, agencies.

In addition, one of the most effective methods of providing quality service at low cost for businesses is to adopt the principles of "Lean Management". Like many businesses that consider themselves "Lean", tourism businesses are in a better position than their competitors in the current economy. Lean management practices in tourism businesses differ from lean practices in the production environment because they are service businesses. The intangibility of services, the difficulty of measuring service outputs, and the creation of value for the customer can complicate the applicability of lean management practices in tourism businesses.

This study addressed the simultaneous relationship between lean management, Industry 4.0, and innovation for the businesses operating in the tourism sector (hotels, agencies) in Turkey. The findings showed that lean management is effective in Industry 4.0. According to this result, it can be argued that lean management forms the basis for Industry 4.0 and, at the same time, is supplemented by Industry 4.0. Therefore, it is thought that integrating Industry 4.0 technologies into lean management processes will increase tourism businesses' productivity (Figure 8). Kolberg and Züehlke (2015) stated that Industry 4.0 applications support lean manufacturing approaches and can be a tool in realising lean manufacturing. Their results support the results obtained in this study.

Regarding another finding obtained in the study, lean management was found to be effective in innovation. According to this result, tourism businesses should implement a lean management approach to improve their innovation capacities. Therefore, it can be stated that lean management will improve the innovation processes of tourism businesses and provide high flexibility in offering new products or services to businesses. Abdallah et al. (2019) stated that innovation-oriented companies tend to benefit from lean management to improve their innovation capacity, which supports the results of this study.

Another finding is that innovation affects Industry 4.0. The emergence of new and creative ideas in businesses will lead to the creation or development of advanced production technologies. Industry 4.0 is an industrial revolution that includes high technological development and transformation; therefore, it is closely related to innovation (TÜSİAD, 2016). Tourism businesses should attach importance to innovation activities for adapting to the age of Industry 4.0. In addition, tourism businesses that want to increase their innovation level and adapt to

the Industry 4.0 age are thought to survive in a competitive environment and reach a better position than their competitors.

The study's main finding is that innovation is a mediator in the relationship between lean management and industry 4.0. In cases where lean management positively affects industry 4.0, the inclusion of innovation in the process will further increase the performance of industry 4.0. Especially in today's global world, the rapid change and development of technological activities have accelerated the efforts towards industry 4.0. In this direction, businesses will gain momentum in the transition to industry 4.0 by focusing on innovation and adopting lean management.

The deepening of global competition has expanded its sphere of influence. Industry 4.0, innovation, and especially lean management concepts in planning and implementing cost, activity, high production, new product development, market creation, generating new ideas, and R&D processes of the businesses and making them sustainable have an essential and critical place in the current dimension of the discussions. Businesses operating in the tourism sector are rapidly developing and changing. This development becomes more critical with the effect of lean management, innovation, and industry 4.0 practices. However, tourism businesses should focus more on lean management than on industry 4.0 and innovation, and lean methods should be adapted to the industry's specific needs.

As a result, this quantitative study in the professional and business sectors addressed the relations between lean management, industry 4.0, and innovation regarding the businesses operating in the tourism sector. Future studies can make comparative studies, including tourism and other sectors or the countries where the research will be conducted can be extended.

References

- Abdallah, A.B., Dahiyat, S.E. and Matsui, Y. (2019). Lean management and innovation performance: Evidence from international manufacturing companies. *Management Research Review*, 42(2), 239-262.
- Al-Aomar, R. and Hussain, M. (2019). Exploration and prioritization of lean techniques in a hotel supply chain. *International Journal of Lean Six Sigma*, 10(1), 375-396.
- Alsos, G. A., Eide, D. and Madsen, E. L. (2014). Introduction: Innovation in Tourism Industries. Chapters, 1-24. (In Handbook of Research on Innovation in Tourism Industries). Cheltenham, UK: Edward Elgar Publishing.

Arpaci, Y, O. (2019). Relationship between Outsourcing and Innovation in Businesses. (Unpublished docto-

Peer-review: Externally peer-reviewed.

Conflict of Interest: The authors have no conflict of interest to declare.

Grant Support: The authors declared that this study has received no financial support.

Ethical Approval: Ethical permission for the study was obtained in writing from the Atatürk University Social and Human Sciences Ethics Committee (Permission Date: 22.04.2022, Session Number: 8, Decision No: 112).
ral dissertation). Pamukkale University, Institute of Social Sciences, Denizli.

- Asdecker B. and Felch V. (2018). Development of an Industry 4.0 Maturity Model for The Delivery Process in Supply Chains. *Journal of Modelling in Management*, 13(4), 840-883.
- Banger, G. (2017). Industry 4.0 Extra. Ankara: Dorlion Publications.
- Byrne, A., (2015). Lean Transformation. Ed.: Melis İnan, İstanbul: Optimist Publishing.
- Chang, M., Walimuni, A. C. S. M., Kim, M. and Lim, H. (2022). Acceptance of tourism blockchain based on UTAUT and connectivism theory. *Technology in Society*, 71, 1-13.
- Cuatrecasas, L. (2004). A lean management implementation method in service operations. *International Journal of Services, Technology and Management*, 5(5/6), 532-544.
- Çanakçı S.D., Çanakçı T. and Geçgin, E. (2019). Destination Image of Kars Province and Eastern Express Experience. *International Journal of Society Research*. 11(18), 1876-1896.
- Canakcioglu, Mustafa, (2019). Tools to fight waste in lean thinking philosophy. *Journal of Social Sciences Research*, 8(3), 270-282.
- Çavuş, M. F. and Akgemci, T. (2008). The Effect of Employee Empowerment on Organizational Creativity and Innovation in Enterprises: A Study in Manufacturing Industry. *Selcuk University Journal of Social Sciences Institute*, 20, 229-244.
- Deran, A. and Beller, B. (2014): Value Stream Costing as a Tool of Lean Management in Hospitals and an Application in a Public Hospital. *Selcuk University Journal of Social Sciences Institute*, 32, 161-174.
- Derin, N. (2008). The Effect of Lean Management on Internal Image According to Employees' Perceptions: A Study in Private Hospitals in Turkey. (Unpublished doctoral dissertation). İnönü University Institute of Social Sciences, Malatya.
- Duren, Z. (2002). Management in the 2000s. Istanbul: Alfa Publications.
- Dülgaroğlu, O. (2021). Digitalization in Tourism: Smart Tourism Applications, Digital Tourism and Tourism 4.0. Journal of Tourism and Management, 2(1), 1-15.
- Engelund, Eva H. Breum, Gitte and Friis, Alan (2009). Optimization of large-scale food production using Lean Manufacturing principles. *Journal of Foodservice*, 20(1), 4-14.
- Elitaş, C. (2011). Research Methods in Accounting. Ankara: Gazi Bookstore.
- Farrington, T., Antony, J. and O'Gorman, K. D. (2018). Continuous improvement methodologies and practices in hospitality and tourism. *International Journal of Contemporary Hospitality Management*, 30(1), 581-600.
- Frederick, D. E. (2016). Libraries, Data and the Fourth Industrial Revolution (Data Deluge Column). Library Hi Tech News, 33(5), 9-12.
- Freeman, C. and Soete, L. (1997). The Economies of Industrial Innovation. London: MIT Press.
- Ghobakhloo, M., Iranmanesh, M., Maroufkhani, P., Grybauskas, A., Vilkas, M. and Petraitė, M. (2021). Industry 4.0, innovation, and sustainable development: A systematic review and a roadmap to sustainable innovation. *Business Strategy and the environment*, 30(2), 1-21.
- Giotis, G. and Papadionysiou, E. (2022). The Role of Managerial and Technological Innovations in the Tourism Industry: A Review of the Empirical Literature. *Sustainability*, 14, 1-20.
- Gupta, S., Sharma, M. and Vijaya Sunder, M. (2016). Lean services: a systematic review. International Journal of Productivity and Performance Management, 65(8), 1025-1056.
- Gürbüz, S. and Şahin, F. (2017). Research Methods in Social Sciences: Philosophy-Method-Analysis. An-

kara: Seçkin Publishing.

- Hadid, W. and Mansouri, S. A. (2014). The lean-performance relationship in services: A theoretical model. International Journal of Operations & Production Management, 34(6), 750-785.
- Hair, J., Black, W., Babin, B. and Anderson, R. (2009). Multivariate data analysis (7th b.). Upper Saddle River, N.J.: Prentice Hall.
- Handscomb, C., Jautelat, S., Rodriguez, M., Wienke, L. and Raedemaecker, S. (2020). Lean management or agile? The right answer may be both. McKinsey, Jul. 14, p. 6.
- Hjalager, A.N. (2002). Repairing Innovation Defectiveness in Tourism. Tourism Management. 23, 465-474.
- http://vetpro40.com/wp-content/uploads/2019/02/End.4.0.Rehber.pdf, Downloaded: 24.03.2022
- Işık, C. and Keskin, G. (2013). The Importance of Innovation in Establishing Competitive Advantage in Information Economies. *Atatürk University Journal of Economics and Administrative Sciences*, 27(1), 41-57.
- Işık, C., Küçükaltan, E. G., Taş, S., Akoğul, E., Uyrun, A., Hajiyeva, T., Turan, B., Dırbo, A. H. and Bayraktaroğlu, E. (2019). Tourism and innovation: A literature review. *Journal of Ekonomi*, 2, 88-154.
- Ivanov, S., Webster, C. and Berezina, K. (2017). Adoption of robots and service automation by tourism and hospitality companies. *Revista Turismo & Desenvolvimento*, 27/28, 1501-1517.
- Imanovic, S., Milojica, V. and Roblek, V. (2016). A holistic approach to innovations in tourism. Paper presented at the International Conference on Economic and Social Studies, Sarajevo, Bosnia and Herzegovina, 21–22 April.
- Kagermann, H., Helbig, J., Hellinger, A. and Wahlster, W. (2013). Recommendations for Implementing the Strategic Initiative, Industrie 4.0: Securing the Future of German Manufacturing Industry, Final Report of the Industrie 4.0 Working Group.
- Kagermann, H., Wahlster, W. and Helbig, J. (2013). Securing the Future of German Manufacturing Industry: Recommendations for Implementing the Strategic Initiative Industrie 4.0. Final Report of the Industrie 4.0 Working Group. Forschungsunion im Stifterverband f
 ür die Deutsche Wirtschaft e. V., Berlin.
- Kolberg, D. and Züehlke, D. (2015). Lean Automation enabled by Industry 4.0 Technologies. *IFAC-Paper-sOnLine*, 48(3), 1870-1875.
- Korže, S. Z. (2019). From Industry 4.0 To Tourism 4.0. Innovative Issues and Approaches in Social Sciences, 12(3), 29-52.
- Krafcik J. F. (1988). Triumph of the Lean Production System. Sloan management review, 30(1), 41-52.
- Kucmarski, T. D. (1996). Innovation: Leadership Strategies for the Competitive Edge. USA: NTC Business Books.
- Leite, H.R. and Vierira, G.E. (2015). Lean philosophy and its applications in the service industry: a review of the current knowledge. *Production*, 25(3), 529-541.
- Lewis, M. A. (2000). Lean Production and Sustainable Competitive Advantage. International Journal of Operations & Production Management, 20(8), 959–978. https://doi.org/10.1108/01443570010332971.
- Lyons, R. K., Chatman, J. and Joyce, C. K. (2007). Innovation in Services: Corporate Culture and Investment Banking. *California Management Review*, 50(1), 174-191.
- Myers, S. and Marquis, D. G. (1969). Successful Industrial Innovations: A Study of Factors Underlying Innovation in Selected Firms. National Science Foundation.
- Nascimento, I.T., Silva, M.E.A., Gama, G.R.S and Xavier, M.C. (2021). Application Of Lean Manage-

ment Tools In Industry 4.0: A Systematic Review. Independent Journal Of Management & Production (IJM&P), 12(9), 856-864.

- Mendoza-Moheno, J., Cruz-Coria, E. and Gonzalez-Cruz, T. F. (2021). Socio-technical innovation in community-based tourism organizations: A proposal for local development. *Technological Forecasting & Social Change*, 171, 1-12.
- Okatan, D. and Yıldırım, Y. (2021). Endüstri 4.0 Teknolojilerinin Turizm Sektörüne Yansımaları: Literatür İncelemesi. Journal of Tourism Intelligence and Smartness, 4(2), 168-185.
- Osei, B. A., Ragavan, N. A. and Mensah, H. K. (2020). Prospects of the fourth industrial revolution for the hospitality industry: a literature review. Journal of Hospitality and Tourism Technology. https://www.emerald.com/insight/1757-9880.htm.
- Narayanamurthy, G. and Gurumurthy, A. (2016). Leanness assessment: a literature review. International Journal of Operations & Production Management, 36(10), 1115-1160.
- Nordin, S. (2003). Tourism Clustering & Innovation: Paths To Economic Growth and Development. European Tourism Research Institute, MiD Sweden University, 14, 1-81.
- Oslo Manual (2005). Guide to Collection and Interpretation of Innovation Data. Ankara: TUBITAK Publications.
- Örnek, A. (1991). Public Administration, İstanbul.
- Perdomo-Verdecia, V., Sacristán-Díaz, M. and Garrido-Vega, P. (2022). Lean management in hotels: Where we are and where we might go. *International Journal of Hospitality Management*, 104, 103250.
- Pereira A.C. and Romero F. (2017). A Review of Meaning and The Implications of The Industry 4.0 Concept. Procedia Manufacturing, 13, 1206-1214.
- Rajput S. and Singh S.P. (2019). Industry 4.0-challenges to implement circular economy. *Benchmarking: An International Journal*, 28(5), 1717-1739.
- Schuh, G., Potente, T., Wesch-Potente, C., Weber, A. R. ve Prote, J. P. (2014). Collaboration Mechanisms to Increase Productivity in the Content of Industrie 4.0. Robust Manufacturing Conference (p. 51-56). Elsivier B:V.
- Schumpeter, J. A. (1961). The Theory of Economic Development. New York: Oxford University Press.
- Sığırcı, M. (2001). National Productivity Center. Key Magazine, 6(68).
- Sony, M. (2018). Industry 4.0 and Lean Management: a Proposed Integration Model and Research Propositions. *Production & Manufacturing Research*, 6(1), 416-432.
- Şengel, Ü. (2021). Chronology of the Interaction Between the Industrial Revolution and Modern Tourism Flows. Journal of Tourism Intelligence and Smartness, 4(1), 19-30.
- Tasgin, A. and Korucuk, M. (2018). Development of Foreign Language Lesson Satisfaction Scale (FLSS): Validity and Reliability Study. *Journal of Curriculum and Teaching*, 7(2), 66-77.
- Tikici, M., Aksoy, A. and Derin, N. (2006). Lean Management as One of the Radical Elements of Total Quality Management. *Electronic Journal of Social Sciences*, 5(15), 20-33.
- Topsakal, Y., Çelik, P. and Yüzbaşıoğlu, N. (2018). Innovation in Small and Medium-Sized Tourism Enterprises (Small Business): The Case of Kaleiçi, Antalya. *Turkish Studies*, 13(3), 737-752.
- Taylor, F. W. (1985). The Principles of Scientific Management, Easton, Hive Publishing, Co., Easton, USA.
- TÜSİAD (2016). Industry 4.0 in Turkey as An Imperative for Global Competitiveness An Emerging Market Perspective. İstanbul: TÜSİAD Publishing.

- Uy, A. O. O. and Rabo, J. S. (2019). IR4.0 Readiness of Accounting Education: Professional Quotient and CHED's accounting program standards. Presented at the DLSU Research Congress 2019 De La Salle University, Manila, Philippines.
- UNWTO (2017) Tourism highlights. 2017 Edition. Available at: https://www.e-unwto.org/doi/pdf/10.18111/9789284419029 (14.6. 2023).
- Vlachos, I. and Bogdanovic, A. (2013). Lean thinking in the European hotel industry. *Tourism Management*, 36, 354- 363.
- Wang L. and Wang L. (2016). Big Data in Cyber-Physical Systems, Digital Manufacturing and Industry 4.0. I.J. Engineering and Manufacturing, 4, 1-8.
- Warnecke, H. J. and Hüser, H. (1995). Lean production. Proceedings of the 12 International Conference on Production Research, 41(13), 37-43.
- Weyer, S., Schmitt, M., Ohmer, M. and Gorecky, D. (2015). Towards Industry 4.0 Standardization as the crucial challenge for highly modular, multi-vendor production systems. *IFAC-PapersOnLine*, 48-3 (2015), 579-584.
- Womack, J. P and Jones, D. T. (2003). Lean Thinking. (Translator: Yamak, O.). Istanbul: Optimist Publications.
- Zengin, Y., Naktiyok, S., Kaygın, E., Kavak, O. and Topçuoğlu, E. (2021). An Investigation of Industry 4.0 and Society 5.0 within the Context of Sustainable Development Goals. *Sustainability*, 13, 1-16 (2682); https://doi.org/10.3390/su13052682

AIM AND SCOPE

Istanbul Business Research (IBR) is an international academic and peer-reviewed journal published by Istanbul University Business School.

As an academic journal that strives for excellence, IBR exclusively publishes high-quality articles in English, covering a broad range of business-related topics. Our main objective is to maintain high academic standards and welcome exceptional contributions from scholars and professionals in the field. IBR publishes novel, theoretical, and research-based studies, including articles that feature robust qualitative and quantitative analysis.

Thank you for your interest and considering IBR as a venue for publishing your research. We encourage scholars and professionals to send high-quality articles that meet our rigorous academic standards.

IBR is published triannually (in April, August and December), providing a platform for multidisciplinary studies from the fields of business administration and economics. Its scope includes a wide range of subjects such as finance, production, marketing, accounting, strategy, human resources, organizational theory, and decision sciences. However, we also encourage studies that explore innovative or interdisciplinary approaches to the study of business-related issues.

EDITORIAL POLICIES AND PEER REVIEW PROCESS

Publication Policy

The subjects covered in the manuscripts submitted to the Journal for publication must be in accordance with the aim and scope of the journal. The journal gives priority to original research papers submitted for publication.

General Principles

Only those manuscripts approved by its every individual author and that were not published before in or sent to another journal, are accepted for evaluation.

Submitted manuscripts that pass preliminary control are scanned for plagiarism using iThenticate software. After plagiarism check, the eligible ones are evaluated by editor-in-chief for their originality, methodology, the importance of the subject covered and compliance with the journal scope.

Short presentations that took place in scientific meetings can be referred if indicated in the article. The editor hands over the papers matching the formal rules to at least two national/ international referees for evaluation and gives green light for publication upon modification by the authors in accordance with the referees' claims. Changing the name of an author (omission, addition or order) in papers submitted to the Journal requires written permission of all declared authors. Refused manuscripts and graphics are not returned to the author.

Copyright Notice

Authors publishing with the journal retain the copyright to their work licensed under the Creative Commons Attribution-NonCommercial 4.0 International license (CC BY-NC 4.0) (https:// creativecommons.org/licenses/by-nc/4.0/) and grant the Publisher non-exclusive commercial right

to publish the work. CC BY-NC 4.0 license permits unrestricted, non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

Open Access Statement

Istanbul Business Research is an open access journal which means that all content is freely available without charge to the user or his/her institution. Except for commercial purposes, users are allowed to read, download, copy, print, search, or link to the full texts of the articles in this journal without asking prior permission from the publisher or the author.

The articles in Istanbul Business Research are open access articles licensed under the terms of the Creative Commons Attribution-NonCommercial 4.0 International License (CC BY-NC 4.0) (https://creativecommons.org/licenses/by-nc-nd/4.0/deed.en/)

Article Processing Charge

All expenses of the journal are covered by the Istanbul University. Processing and publication are free of charge with the journal. There is no article processing charges or submission fees for any submitted or accepted articles.

Condition of Acceptance For Evaluation

Manusripts which do not comply with the principles and rules defined in the Information for Authors and which do not meet the requirements stated in the Manuscript Preparation section in below are not taken into consideration and will be returned to the corresponding author for the completion of the necessary arrangements.

Peer Review Process

Only those manuscripts approved by its every individual author and that were not published before in or sent to another journal, are accepted for evaluation.

Submitted manuscripts that pass preliminary control are scanned for plagiarism using iThenticate software. After plagiarism check, the eligible ones are evaluated by Editor-in-Chief for their originality, methodology, the importance of the subject covered and compliance with the journal scope. Editor-in-Chief evaluates manuscripts for their scientific content without regard to ethnic origin, gender, sexual orientation, citizenship, religious belief or political philosophy of the authors and ensures a fair double-blind peer review of the selected manuscripts.

The selected manuscripts are sent to at least two national/international external referees for evaluation and publication decision is given by editor-in-chief upon modification by the authors in accordance with the referees' claims.

Editor-in-Chief does not allow any conflicts of interest between the authors, editors and reviewers and is responsible for final decision for publication of the manuscripts in the Journal.

Reviewers' judgments must be objective. Reviewers' comments on the following aspects are expected while conducting the review.

- Does the manuscript contain new and significant information?
- Does the abstract clearly and accurately describe the content of the manuscript?
- Is the problem significant and concisely stated?
- Are the methods described comprehensively?
- Are the interpretations and consclusions justified by the results?
- Is adequate references made to other Works in the field?
- Is the language acceptable?

Reviewers must ensure that all the information related to submitted manuscripts is kept as confidential and must report to the editor if they are aware of copyright infringement and plagiarism on the author's side.

A reviewer who feels unqualified to review the topic of a manuscript or knows that its prompt review will be impossible should notify the editor and excuse himself from the review process.

The editor informs the reviewers that the manuscripts are confidential information and that this is a privileged interaction. The reviewers and editorial board cannot discuss the manuscripts with other persons. The anonymity of the referees is important.

PUBLICATION ETHICS AND PUBLICATION MALPRACTICE STATEMENT

Istanbul Business Research is committed to upholding the highest standards of publication ethics and pays regard to Principles of Transparency and Best Practice in Scholarly Publishing published by the Committee on Publication Ethics (COPE), the Directory of Open Access Journals (DOAJ), the Open Access Scholarly Publishers Association (OASPA), and the World Association of Medical Editors (WAME) on https://publicationethics.org/resources/guidelines-new/principlestransparency-and-best-practice-scholarly-publishing

All parties involved in the publishing process (Editors, Reviewers, Authors and Publishers) are expected to agree on the following ethical principles.

All submissions must be original, unpublished (including as full text in conference proceedings), and not under the review of any other publication synchronously. Each manuscript is reviewed by one of the editors and at least two referees under double-blind peer review process. Plagiarism, duplication, fraud authorship/denied authorship, research/data fabrication, salami slicing/salami publication, breaching of copyrights, prevailing conflict of interest are unnethical behaviors.

All manuscripts not in accordance with the accepted ethical standards will be removed from the publication. This also contains any possible malpractice discovered after the publication. In accordance with the code of conduct we will report any cases of suspected plagiarism or duplicate publishing.

Research Ethics

The journal adheres to the highest standards in research ethics and follows the principles of international research ethics as defined below. The authors are responsible for the compliance of the manuscripts with the ethical rules.

- Principles of integrity, quality and transparency should be sustained in designing the research, reviewing the design and conducting the research.
- The research team and participants should be fully informed about the aim, methods, possible uses and requirements of the research and risks of participation in research.
- The confidentiality of the information provided by the research participants and the confidentiality of the respondents should be ensured. The research should be designed to protect the autonomy and dignity of the participants.
- Research participants should participate in the research voluntarily, not under any coercion.
- Any possible harm to participants must be avoided. The research should be planned in such a way that the participants are not at risk.
- The independence of research must be clear; and any conflict of interest or must be disclosed.

- In experimental studies with human subjects, written informed consent of the participants who decide to participate in the research must be obtained. In the case of children and those under wardship or with confirmed insanity, legal custodian's assent must be obtained.
- If the study is to be carried out in any institution or organization, approval must be obtained from this institution or organization.
- In studies with human subject, it must be noted in the method's section of the manuscript that the informed consent of the participants and ethics committee approval from the institution where the study has been conducted have been obtained.

Author Responsibilities

It is authors' responsibility to ensure that the article is in accordance with scientific and ethical standards and rules. And authors must ensure that submitted work is original. They must certify that the manuscript has not previously been published elsewhere or is not currently being considered for publication elsewhere, in any language. Applicable copyright laws and conventions must be followed. Copyright material (e.g. tables, figures or extensive quotations) must be reproduced only with appropriate permission and acknowledgement. Any work or words of other authors, contributors, or sources must be appropriately credited and referenced.

All the authors of a submitted manuscript must have direct scientific and academic contribution to the manuscript. The author(s) of the original research articles is defined as a person who is significantly involved in "conceptualization and design of the study", "collecting the data", "analyzing the data", "writing the manuscript", "reviewing the manuscript with a critical perspective" and "planning/ conducting the study of the manuscript and/or revising it". Fund raising, data collection or supervision of the research group are not sufficient roles to be accepted as an author. The author(s) must meet all these criteria described above. The order of names in the author list of an article must be a co-decision and it must be indicated in the Copyright Agreement Form. The individuals who do not meet the authorship criteria but contributed to the study must take place in the acknowledgement section. Individuals providing technical support, assisting writing, providing a general support, providing material or financial support are examples to be indicated in acknowledgement section.

All authors must disclose all issues concerning financial relationship, conflict of interest, and competing interest that may potentially influence the results of the research or scientific judgment. When an author discovers a significant error or inaccuracy in his/her own published paper, it is the author's obligation to promptly cooperate with the Editor to provide retractions or corrections of mistakes.

Responsibility for the Editor and Reviewers

Editor-in-Chief evaluates manuscripts for their scientific content without regard to ethnic origin, gender, sexual orientation, citizenship, religious belief or political philosophy of the authors. He/ She provides a fair double-blind peer review of the submitted articles for publication and ensures that all the information related to submitted manuscripts is kept as confidential before publishing.

Editor-in-Chief is responsible for the contents and overall quality of the publication. He/She must publish errata pages or make corrections when needed.

Editor-in-Chief does not allow any conflicts of interest between the authors, editors and reviewers. Only he has the full authority to assign a reviewer and is responsible for final decision for publication of the manuscripts in the Journal.

Reviewers must have no conflict of interest with respect to the research, the authors and/or the research funders. Their judgments must be objective.

Reviewers must ensure that all the information related to submitted manuscripts is kept as confidential and must report to the editor if they are aware of copyright infringement and plagiarism on the author's side.

A reviewer who feels unqualified to review the topic of a manuscript or knows that its prompt review will be impossible should notify the editor and excuse himself from the review process.

The editor informs the reviewers that the manuscripts are confidential information and that this is a privileged interaction. The reviewers and editorial board cannot discuss the manuscripts with other persons. The anonymity of the referees must be ensured. In particular situations, the editor may share the review of one reviewer with other reviewers to clarify a particular point.

MANUSCRIPT ORGANIZATION

Language

The language of the journal is English.

Manuscript Organization and Submission

All correspondence will be sent to the first-named author unless otherwise specified. Manuscpript is to be submitted online via http://ibr.istanbul.edu.tr and it must be accompanied by a cover letter indicating that the manuscript is intended for publication, specifying the article category (i.e. research article, review etc.) and including information about the manuscript (see the Submission Checklist). Manuscripts should be prepared in Microsoft Word 2003 and upper versions. In addition, Copyright Agreement Form that has to be signed by all authors must be submitted.

- 1. The manuscripts should be in A4 paper standards: having 2.5 cm margins from right, left, bottom and top, Times New Roman font style in 12 font size and line spacing of 1.5.
- 2. The manuscripts should contain between 4500 and 8500 words and the page numbers must be at the bottom and in the middle.
- 3. The name(s) of author(s) should be given just beneath the title of the study aligned to the left. Also the affiliation, title, e-mail and phone of the author(s) must be indicated on the bottom of the page as a footnote marked with an asterisk (*).
- 4. Submitted manuscripts in English must have before the introduction section, the abstract in English, between 180 and 200 words, summarizing the scope, the purpose, the results of the study and the methodology used. Underneath the abstracts, 3 to 5 keywords that inform the reader about the content of the study should be specified in English.
- The manuscripts should contain mainly these components: title, abstract and keywords; extended abstract, body text with sections, footnotes and references.
- 6. The 6th version of the APA (American Psychological Association) reference style should be used for in-text citation and references list. For information and samples on the APA 6 style please refer to http://ibr.istanbul.edu.tr/en/content/information-for-authors/references and/or REFERENCES section in the Information for Authors.
- 7. Tables, graphs and figures can be given with a number and a defining title if and only if it is necessary to follow the idea of the article. the article. The title of table should be placed above the table; caption of figure should be placed beneath the figure. Features like demographic characteristics that can be given within the text should not be indicated as tables separately.

- 8. Please submit tables as editable text and not as images. When preparing tables, please avoid using vertical rules and shading in tables cells. If you are using a table grid, use only one grid for each individual table and not a grid for each row. If no grid is used, use tabs, not spaces, to align columns.
- 9. It is important that the file be saved in the native format of the word processor used. You are strongly advised to use the "spell-check" and grammer-check" functions of your word processor to avoid unnecessary errors.
- 10. The text should be in single-column format. Keep the layout of the text as simple as possible.
- 11. A title page including author information must be submitted together with the manuscript. The title page is to include fully descriptive title of the manuscript and, affiliation, title, e-mail address, postal address, phone and fax number of the author(s) (see The Submission Checklist).
- 12. Authors are responsible for all statements made in their work submitted to the Journal for publication.
- 13. The author(s) can be asked to make some changes in their articles due to peer reviews.
- 14. A copy of the journal will be sent to each author of the accepted articles upon their request.

Figures and Figures Legends

Figures, graphics, and photographs should be submitted as separate files (in TIFF or JPEG format) through the submission system. The files should not be embedded in a Word document or the main document. When there are figure subunits, the subunits should not be merged to form a single image. Each subunit should be submitted separately through the submission system. Images should not be labeled (a, b, c, etc.) to indicate figure subunits. Thick and thin arrows, arrowheads, stars, asterisks, and similar marks can be used on the images to support figure legends. Like the rest of the submission, the figures too should be blinded. The minimum resolution of each submitted figure should be 300 DPI. To prevent delays in the evaluation process, all submitted figures should be clear in resolution and large in size (minimum dimensions: 100×100 mm). Figure legends should be listed at the end of the main document.

Tables

Tables should be included in the main document, presented after the reference list, and they should be numbered consecutively in the order they are referred to within the main text. A descriptive title must be placed above the tables. Abbreviations used in the tables should be defined below the tables by footnotes (even if they are defined within the main text). Tables should be created using the"insert table" command of the word processing software and they should be arranged clearly to provide easy reading. Data presented in the tables should not be a repetition of the data presented within the main text but should be supporting the main text.

REFERENCES

Although references to review articles can be an efficient way to guide readers to a body of literature, review articles do not always reflect original work accurately. Readers should therefore be provided with direct references to original research sources whenever possible. On the other hand, extensive lists of references to original work on a topic can use excessive space on the printed page. Small numbers of references to key original papers often serve as well as more exhaustive lists, particularly since references can now be added to the electronic version of published papers, and since electronic literature searching allows readers to retrieve published literature efficiently. Papers accepted but not yet included in the issue are published online in the Early View section and they should be cited as "advance online publication". Citing a "personal communication" should be avoided unless it provides essential information not available from a public source, in which case the name of the person and date of communication should be cited in parentheses in the text. For scientific articles, written permission and confirmation of accuracy from the source of a personal communication must be obtained.

Reference Style and Format

Istanbul Business Research complies with APA (American Psychological Association) style 6th Edition for referencing and quoting. For more information:

- American Psychological Association. (2010). Publication manual of the American Psychological Association (6th ed.). Washington, DC: APA.
- http://www.apastyle.org

Citations in the Text

Citations must be indicated with the author surname and publication year within the parenthesis. If more than one citation is made within the same paranthesis, separate them with (;).

Samples:

More than one citation;

(Stein, et al., 2002; Pierce, 1995) *Citation with one author;* (Tang, 2007) *Citation with two authors;* (Tang & Pierce, 2007)

Citation with three, four, five authors;

First citation in the text: (Ailen, Ciambrune, & Welch, 2000) Subsequent citations in the text: (Ailen, et al., 2000)

Citations with more than six authors; (Robbins, et al., 2003)

Citations in the Reference

All the citations done in the text should be listed in the References section in alphabetical order of author surname without numbering. Below given examples should be considered in citing the references.

Basic Reference Types

Book

a) Books

Kamien R., & Kamien A. (2014). *Music: An appreciation*. New York, NY: McGraw-Hill Education. *b) Edited Book*

Ören, T., Üney, T., & Çölkesen, R. (Eds.). (2006). *Türkiye bilişim ansiklopedisi* [Turkish Encyclopedia of Informatics]. İstanbul, Turkey: Papatya Yayıncılık.

c) Chapter in an Edited Book

Bassett, C. (2006). Cultural studies and new media. In G. Hall & C. Birchall (Eds.), New cultural studies: Adventures in theory (pp. 220–237). Edinburgh, UK: Edinburgh University Press.

d) Book with the same organization as author and publisher

American Psychological Association. (2009). *Publication manual of the American psychological association* (6th ed.). Washington, DC: Author.

Article

a) Journal article with DOI

- de Cillia, R., Reisigl, M., & Wodak, R. (1999). The discursive construction of national identity. *Discourse and Society*, 10(2), 149–173. http://dx.doi.org/10.1177/0957926599010002002
- b) Journal Article with DOI and More Than Seven Authors
- Lal, H., Cunningham, A. L., Godeaux, O., Chlibek, R., Diez-Domingo, J., Hwang, S.-J. ... Heineman, T. C. (2015). Efficacy of an adjuvanted herpes zoster subunit vaccine in older adults. *New England Journal of Medicine*, 372, 2087–2096. http://dx.doi.org/10.1056/NEJMoa1501184

c) Journal Article from Web, without DOI

Sidani, S. (2003). Enhancing the evaluation of nursing care effectiveness. Canadian Journal of Nursing Research, 35(3), 26–38. Retrieved from http://cjnr.mcgill.ca

d) Journal Article wih DOI

Turner, S. J. (2010). Website statistics 2.0: Using Google Analytics to measure library website effectiveness. *Technical Services Quarterly*, 27, 261–278. http://dx.doi.org/10.1080/0731713 1003765910

e) Advance Online Publication

Smith, J. A. (2010). Citing advance online publication: A review. *Journal of Psychology*. Advance online publication. http://dx.doi.org/10.1037/a45d7867

f) Article in a Magazine

Henry, W. A., III. (1990, April 9). Making the grade in today's schools. Time, 135, 28-31.

Doctoral Dissertation, Master's Thesis, Presentation, Proceeding

a) Dissertation/Thesis from a Commercial Database

Van Brunt, D. (1997). *Networked consumer health information systems* (Doctoral dissertation). Available from ProQuest Dissertations and Theses database. (UMI No. 9943436)

b) Dissertation/Thesis from an Institutional Database

Yaylalı-Yıldız, B. (2014). University campuses as places of potential publicness: Exploring the politicals, social and cultural practices in Ege University (Doctoral dissertation). Retrieved from Retrieved from: http://library.iyte.edu.tr/tr/hizli-erisim/iyte-tez-portali

c) Dissertation/Thesis from Web

Tonta, Y. A. (1992). An analysis of search failures in online library catalogs (Doctoral dissertation, University of California, Berkeley). Retrieved from http://yunus.hacettepe.edu.tr/~tonta/ yayinlar /phd/ickapak.html

d) Dissertation/Thesis abstracted in Dissertations Abstracts International

Appelbaum, L. G. (2005). Three studies of human information processing: Texture amplification, motion representation, and figure-ground segregation. *Dissertation Abstracts International: Section B. Sciences and Engineering*, 65(10), 5428.

e) Symposium Contribution

Krinsky-McHale, S. J., Zigman, W. B., & Silverman, W. (2012, August). Are neuropsychiatric symptoms markers of prodromal Alzheimer's disease in adults with Down syndrome? In W. B. Zigman (Chair), *Predictors of mild cognitive impairment, dementia, and mortality in adults with Down syndrome*. Symposium conducted at the meeting of the American Psychological Association, Orlando, FL.

f) Conference Paper Abstract Retrieved Online

Liu, S. (2005, May). Defending against business crises with the help of intelligent agent based early warning solutions. Paper presented at the Seventh International Conference on Enterprise Information Systems, Miami, FL. Abstract retrieved from http://www.iceis.org/iceis2005/ abstracts 2005.htm

g) Conference Paper - In Regularly Published Proceedings and Retrieved Online

Herculano-Houzel, S., Collins, C. E., Wong, P., Kaas, J. H., & Lent, R. (2008). The basic nonuniformity of the cerebral cortex. *Proceedings of the National Academy of Sciences*, 105, 12593–12598. http://dx.doi.org/10.1073/pnas.0805417105

h) Proceeding in Book Form

Parsons, O. A., Pryzwansky, W. B., Weinstein, D. J., & Wiens, A. N. (1995). Taxonomy for psychology. In J. N. Reich, H. Sands, & A. N. Wiens (Eds.), Education and training beyond the doctoral degree: Proceedings of the American Psychological Association National Conference on Postdoctoral Education and Training in Psychology (pp. 45–50). Washington, DC: American Psychological Association.

i) Paper Presentation

Nguyen, C. A. (2012, August). *Humor and deception in advertising: When laughter may not be the best medicine.* Paper presented at the meeting of the American Psychological Association, Orlando, FL.

SUBMISSION CHECKLIST

Ensure that the following items are present:

- Cover letter to the editor
 - ✓ The category of the manuscript
 - ✓ Confirming that "the paper is not under consideration for publication in another journal".
 - ✓ Including disclosure of any commercial or financial involvement.
 - \checkmark Confirming that the statistical design of the research article is reviewed.
 - ✓ Confirming that last control for fluent English was done.
 - ✓ Confirming that journal policies detailed in Information for Authors have been reviewed.

- ✓ Confirming that the references cited in the text and listed in the references section are in line with APA 6.
- Copyright Agreement Form
- Permission of previously published copyrighted material if used in the present manuscript
- Author form
- Title page
 - ✓ The category of the manuscript
 - ✓ The title of the manuscript
 - ✓ All authors' names and affiliations (institution, faculty/department, city, country), e-mail addresses
 - \checkmark Corresponding author's email address, full postal address, telephone and fax number
 - ✓ ORCIDs of all authors.
 - ✓ Grant support (if exists)
 - ✓ Conflict of interest (if exists)
 - ✓ Acknowledgement (if exists)
- Main Manuscript Document
 - \checkmark The title of the manuscript
 - ✓ Abstract (180-200 words)
 - ✓ Key words: 3 to 5 words
 - ✓ Main article sections
 - ✓ References
 - ✓ All tables, illustrations (figures) (including title, description, footnotes)

İstanbul University İstanbul Üniversitesi



Journal name: Istanbul Business Research Dergi Adı: Istanbul Business Research **Copyright Agreement Form** *Telif Hakkı Anlaşması Formu*

Responsible/Corresponding Author					
Sorumli Title of	u Yazar Monuscrint				
Makalenin Başlığı					
Acceptance date					
Kabul Tarihi					
Yazarların Listesi					
Sıra	Name - Surname	E-mail	Signatur	re Date	
No	Adı-Soyadı	E-Posta	Imza	Tarih	
2					
3					
4					
5					
Manuscript Type (Research Article, Review, etc.)					
Makalenin türü (Araştırma makalesi, Derleme, v.b.)					
Responsible/Corresponding Author: Sorumlu Yazar:					
University/company/institution		Çalıştığı kurum	lıştığı kurum		
Address		Posta adresi	osta adresi		
E-mail		E-posta	posta		
Phone; mobile phone 7		Telefon no; GSM no	lefon no; GSM no		
all authors participated in the work in a substantive way, and are prepared to take public responsibility for the work, all authors have seen and approved the manuscript as submitted, the manuscript has not been published and is not being submitted or considered for publication elsewhere, the text, illustrations, and any other materials included in the manuscript do not infringe upon any existing copyright or other rights of anyone. ISTANBUL UNIVERSITY will publish the content under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) license that gives permission to copy and redistribute the material in any medium or format other than commercial purposes as well as remix, transform and build upon the material by providing appropriate credit to the original work. The Contributor(s) or, if applicable the Contributor's Employer, retain(s) all proprietary rights in addition to copyright, patent rights. I/We indemnify ISTANBUL UNIVERSITY and the Editors of the Journals, and hold them harmless from any loss, expense or damage occasioned by a claim or suit by a third party for copyright infringement, or any suit arising out of any breach of the foregoing warrantic as a result of publication of my/our article. I/We also warrant that the article contains no libelous or unlawful statements, and does not contain material or instructions that might cause harm or injury. This Copyright Agreement Form must be signed/ratified by all authors. Separate copies of the form (completed in full) may be submitted by authors located at different institutions; however, all signatures must be original and authenticated.					
Tüm yazarların bu çalışmaya aslı ölarak katılmış olduklarını ve bu çalışma için her türlü sorumluluğu aldıklarını, Tüm yazarların sunulan makalenin son halini gördüklerini ve onayladıklarını, Makalenin başka bir yerde basılmadığımı veya basılmak için sunulmadığını, Makalede bulunan metnin, şekillerin ve dokümanların diğer şahıslara ait olan Telif Haklarını ihlal etmediğini kabul ve taahhüt ederler. ISTANBUL UNİVERSITESI'nin bu fikri eseri, Creative Commons AttıF-Gayri Ticari 4.0 Uluslararası (CC BY-NC 4.0) lisansı ile yayınlamasına izin verirler. Creative Commons Atıf-Gayri Ticari 4.0 Uluslararası (CC BY-NC 4.0) lisansı, eserin itcari kullanım dışında her boyut ve formatta paylaşılmasına, kopyalanmasına, çoğultımısına ve orijinal esere uygun şekilde atıfta bulunmak kaydıyla yeniden düzenleme, dönüştürme ve eserin üzerine inşa etme dähil adapte edilmesine izin verir. Yazarı(ları) neya varsa yazar(ları)ın işvereninin telif dähil patent hakları, fikri mülkiyet hakları saklıdır. Ben/Bix, telif hakkı ihlali nedeniyle tiçtineti şahıslarca vuku bulacak hak talebi veya açılacak davalarda İSTANBUL ÜNİVERSİTESİ ve Dergi Editörlerinin hiçbir sorumluluğunun olmadığını, tüm sorumluluğun yazarlara ait olduğunu taahhüt dedirin/cetiri. Ayrıca Ben/Biz nakalede hiçbir suç unsuru veya kanuna aykırı ifade bulunmadığını, araştırma yapılırken kanuna aykırı herhangi bir malzeme ve yöntem kullanılmadığını taahhüt ederim/ederiz. Bu Telif Hakkı Anlaşmas Formu tüm yazarlara tarafından imzalanmalıdır/onaylanmalıdır. Form farklı kurumlarda bulunan yazarlar tarafından ayırı kopyalar halinde doldurularak sunulabilir. Ancak, tüm imzaların orijinal veya kanıtlanabilir şekilde onaylı olması gerekir.					
Respon Sorumli	sible/Corresponding Author u Yazar;	; Signature	/ İmza	Date / Tarih	
				/	